UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2022

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from_____ to ____

Commission file number 001-36548

ATARA BIOTHERAPEUTICS, INC.

(Exact name of Registrant as specified in its Charter)

Delaware

(State or other jurisdiction of incorporation or organization)

46-0920988 (I.R.S. Employer Identification No.)

611 Gateway Blvd., Suite 900 South San Francisco, CA (Address of principal executive offices)

94080 (Zip Code)

Accelerated filer

Smaller reporting company

Registrant's telephone number, including area code: (650) 278-8930

Securities registered pursuant to Section 12(b) of the Act:

X

Title of Each Class	Trading Symbol(s)	Name of Each Exchange on Which Registered							
Common Stock, par value \$0.0001 per share	ATRA	The Nasdaq Stock Market LLC							
Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 🛛 No									
Indicate by check mark whether the registrant has submitted electro	nically every Interactive Data File requir	ed to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of							

this chapter) during the preceding 12 months (or for such shorter period that the Registrant was required to submit such files). Yes 🗵 No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer Non-accelerated filer Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the Registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵

The number of outstanding shares of the Registrant's Common Stock as of April 25, 2022 was 93,405,726 shares.

Х

ATARA BIOTHERAPEUTICS, INC.

INDEX

DADTI	FINANCIAL INFORMATION	Page
PART I.	FINANCIAL INFORMATION	
Item 1.	Financial Statements (Unaudited)	
	Condensed Consolidated Balance Sheets	5
	Condensed Consolidated Statements of Operations and Comprehensive Loss	6
	Condensed Consolidated Statements of Changes in Stockholders' Equity	7
	Condensed Consolidated Statements of Cash Flows	8
	Notes to Condensed Consolidated Financial Statements	9
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	30
Item 4.	Controls and Procedures	30
PART II.	OTHER INFORMATION	31
Item 1.	Legal Proceedings	31
Item 1A.	Risk Factors	31
Item 6.	Exhibits	73
	Signatures	74

NOTE REGARDING FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of the safe harbor provisions of the U.S. Private Securities Litigation Reform Act of 1995. Such forward-looking statements, which represent our intent, belief or current expectations, involve risks and uncertainties and other factors that could cause actual results and the timing of certain events to differ materially from future results expressed or implied by such forward-looking statements. In some cases you can identify these statements by forward-looking words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "could," "would," "project," "predict," "plan," "expect" or the negative or plural of these words or similar expressions. The forward-looking statements include, but are not limited to, statements about:

- our expectations regarding the timing of initiating clinical studies, opening client sites, enrolling clinical studies and reporting results of clinical studies for our programs, including in light of the COVID-19 pandemic;
- the likelihood and timing of regulatory submissions or related approvals for our product candidates, including the initiation, completion and expectations about
 the timing of approvals for our biologics license application (BLA) and marketing authorization application (MAA) for tab-cel[®] for patients with Epstein-Barr
 virus with post-transplant lymphoproliferative disease (EBV+ PTLD);
- the potential indications for our product candidates, if approved for commercial use;
- the potential market opportunities for commercializing our product candidates;
- our Research, Development and License Agreement with Bayer AG, including potential milestone and royalty payments under the agreement;
- our Commercialization Agreement with Pierre Fabre Medicament, including potential milestone and royalty payments under the agreement;
- our Asset Purchase Agreement and related transactions with FUJIFILM Diosynth Biotechnologies California, Inc., including the Master Services and Supply Agreement;
- our expectations regarding the potential market size and the size of the patient populations for our product candidates, if approved for commercial use;
- estimates of our expenses, capital requirements and need for additional financing;
- our expectation regarding the length of time that our existing capital resources will be sufficient to enable us to fund our planned operations;
- our ability to commercialize our product candidates, if approved for commercial use;
- our ability to develop, acquire and advance product candidates into, and successfully complete, clinical studies;
- the initiation, timing, costs, progress and results of future preclinical studies and clinical studies and our research and development programs;
- our ability to enter into and maintain contracts with clinical research organizations, manufacturing organizations and other vendors for clinical and pre-clinical studies, supplies and other services;
- the scope of protection we are able to obtain and maintain for the intellectual property rights covering our product candidates;
- our financial performance;
- developments and projections relating to our competitors and our industry;
- our ability to have our product candidates manufactured for our clinical studies, or if approved, for commercial sale;
- the impact of COVID-19 to our business and operations, as well as the businesses and operations of third parties on which we rely;
- our ability to sell or have manufactured approved products at commercially reasonable values; and
- timing and costs related to qualification of our CMOs' manufacturing facilities for commercial production.

These statements are only current predictions and are subject to known and unknown risks, uncertainties, including, without limitation, risks and uncertainties associated with the costly and time-consuming pharmaceutical product development process and the uncertainty of clinical success; the COVID-19 pandemic, which may significantly impact (i) our business, research, clinical

development plans and operations, including our operations in South San Francisco and Southern California and at our clinical trial sites, as well as the business or operations of our third-party manufacturers, contract research organizations or other third parties with whom we conduct business, (ii) our ability to access capital, and (iii) the value of our common stock; the sufficiency of our cash resources and need for additional capital, and other factors that may cause our or our industry's actual results, levels of activity, performance or achievements to be materially different from those anticipated by the forward-looking statements. We discuss many of these risks in this report in greater detail under the heading "1A. Risk Factors" and elsewhere in this report. You should not rely upon forward-looking statements as predictions of future events. New risk factors and uncertainties may emerge from time to time, and it is not possible for management to predict all risks and uncertainties.

In this Quarterly Report on Form 10-Q, unless the context requires otherwise, "Atara," "Atara Biotherapeutics," "Company," "we," "our," and "us" means Atara Biotherapeutics, Inc. and, where appropriate, its subsidiaries.

Summary Risk Factors

Our business is subject to numerous risks and uncertainties that may have a material adverse effect on our business, financial condition or results of operations. These risks are more fully described below. These risks include, among others:

- we have incurred substantial losses since our inception and anticipate that we will continue to incur substantial and increasing losses for the foreseeable future;
- we currently have no approved products and thus have no revenues from commercialization of any products and may never generate revenues from the commercial sale of products or achieve profitability;
- we are generally early in our development efforts, have only a small number of product candidates in clinical development, and we will need to successfully complete preclinical and clinical testing of our product candidates before we can seek regulatory approval and potentially generate commercial sales;
- we will require substantial additional financing to achieve our goals, which may not be available to us on acceptable terms, or at all;
- our future success depends on our ability to retain our executive officers and to attract, retain and motivate qualified personnel;
- the results of preclinical studies or earlier clinical studies are not necessarily predictive of future results, and our existing product candidates in clinical studies, and any other product candidates we advance into clinical studies may not have favorable results in later clinical studies or receive regulatory approval;
- clinical drug development, both in the U.S. and international jurisdictions, involves a lengthy and expensive process with an uncertain outcome and even if our
 product candidates receive regulatory approval, they may still face future development and regulatory difficulties;
- our T-cell immunotherapy product candidates and our next-generation CAR T programs represent new therapeutic approaches that could result in heightened
 regulatory scrutiny and delays in or our inability to achieve regulatory approval, commercialization or payor coverage of our product candidates;
- there can be no assurance that we will achieve all of the anticipated benefits of the Fujifilm Transaction and we could face unanticipated challenges;
- the market opportunities for our product candidates may be limited to those patients who are ineligible for or have failed prior treatments and may be small;
- we may not be able to obtain or maintain orphan drug exclusivity for our product candidates;
- the COVID-19 pandemic continues to impact our business and operations and could materially and adversely affect our business and operations in the future, as well as the businesses and operations of third parties on which we rely;
- our success depends upon our ability to obtain and maintain sufficient intellectual property protection for our product candidates, and we may not be able to protect our intellectual property rights throughout the world;
- our principal stockholders own a significant percentage of our stock and will be able to exert significant control over matters subject to stockholder approval; and
- we may not be able to obtain and maintain the relationships with third parties that are necessary to develop, commercialize and manufacture some or all of our product candidates.



ATARA BIOTHERAPEUTICS, INC. Condensed Consolidated Balance Sheets (Unaudited) (In thousands, except per share amounts)

		March 31, 2022	December 31, 2021		
Assets					
Current assets:					
Cash and cash equivalents	\$	101,669	\$	106,084	
Short-term investments		200,142		264,984	
Restricted cash		146		194	
Accounts receivable		668		986	
Prepaid expenses and other current assets		11,574		12,373	
Assets held for sale		190			
Total current assets		314,389		384,621	
Property and equipment, net		9,109		53,780	
Operating lease assets		18,933		26,159	
Restricted cash - long-term		1,200		1,200	
Other assets		2,220		2,367	
Long-term assets held for sale		51,429			
Total assets	\$	397,280	\$	468,127	
Liabilities and stockholders' equity					
Current liabilities:					
Accounts payable	\$	11,945	\$	17,368	
Accrued compensation	ψ	16,381	Ψ	25,150	
Accrued research and development expenses		13,990		13,451	
Deferred revenue		50,943		40,760	
Other current liabilities		8,051		9,057	
Liabilities held for sale		252			
Total current liabilities		101,562		105,786	
Deferred revenue - long-term		45,000		55,708	
Operating lease liabilities - long-term		17,453		25,518	
Other long-term liabilities		616		1,501	
Long-term liabilities held for sale		8,333		1,501	
Total liabilities		172,964		188,513	
Commitments and contingencies (Note 8)					
Stockholders' equity:					
Common stock—\$0.0001 par value, 500,000 shares authorized as of March 31, 2022 and December 31, 2021; 93,406 and 91,671 shares issued and outstanding as of March 31, 2022 and December 31, 2021, respectively		9		9	
Additional paid-in capital		1,779,026		1,744,695	
Accumulated other comprehensive (loss) income		(1,892)		(368)	
Accumulated other comprehensive (ross) income		(1,892)		(1,464,722)	
Total stockholders' equity		224,316		279.614	
	e	/	¢		
Total liabilities and stockholders' equity	\$	397,280	\$	468,127	

See accompanying notes.

ATARA BIOTHERAPEUTICS, INC. Condensed Consolidated Statements of Operations and Comprehensive Loss (Unaudited) (In thousands, except per share amounts)

	Three Months E	nded March	31,
	 2022		2021
License and collaboration revenue	\$ 7,314	\$	3,552
Operating expenses:			
Research and development	74,963		64,059
General and administrative	20,571		17,738
Total operating expenses	95,534		81,797
Loss from operations	(88,220)		(78,245)
Interest and other income (expense), net	115		(90)
Loss before provision for income taxes	(88,105)		(78,335)
Provision for income taxes	_		
Net loss	\$ (88,105)	\$	(78,335)
Other comprehensive gain (loss):			
Unrealized gain (loss) on available-for-sale securities	(1,524)		(135)
Comprehensive loss	\$ (89,629)	\$	(78,470)
Net loss per common share:	 		
Basic and diluted net loss per common share	\$ (0.87)	\$	(0.86)
-	<u> </u>	-	
Weighted-average shares outstanding used to calculate basic			
and diluted net loss per common share	100,726		91,456

See accompanying notes.

ATARA BIOTHERAPEUTICS, INC. Condensed Consolidated Statements of Changes in Stockholders' Equity (Unaudited) (In thousands)

		Common Stock					cumulated Other prehensive	А	ccumulated	Sto	Total :kholders'
For the Three Months Ended March 31, 2022	Shares	1	Amount	Capital		Income (Loss)		Deficit		Equity	
Balance as of January 1, 2022	91,671	\$	9	\$	1,744,695	\$	(368)	\$	(1,464,722)	\$	279,614
Issuance of common stock through ATM facilities, net of commissions											
and offering costs of \$419	1,320		_		20,516		_		_		20,516
RSU settlements, net of shares withheld	405		_		(616)		_		_		(616)
Issuance of common stock pursuant to employee stock awards	10		_		96		_		_		96
Stock-based compensation expense	_		_		14,335		_		_		14,335
Net loss	_		_		_		_		(88,105)		(88,105)
Unrealized loss on available-for-sale securities							(1,524)				(1,524)
Balance as of March 31, 2022	93,406	\$	9	\$	1,779,026	\$	(1,892)	\$	(1,552,827)	\$	224,316

						Ac	cumulated				
	Common Stock			Additional Paid-in		Other Comprehensive		Accumulated		Total Stockholders'	
For the Three Months Ended March 31, 2021	Shares		Amount		Capital	Inc	come (Loss)		Deficit		Equity
Balance as of January 1, 2021	83,372	\$	8	\$	1,586,616	\$	296	\$	(1,124,581)	\$	462,339
Issuance of common stock through ATM facilities, net of commissions and offering costs of \$139	146		_		2,382		_		_		2,382
RSU settlements, net of shares withheld	449		_		(1,231)		_		_		(1,231)
Issuance of common stock pursuant to employee stock awards	108		_		1,749		_		_		1,749
Stock-based compensation expense	_		_		12,268		_		_		12,268
Net loss	_		_		_		_		(78,335)		(78,335)
Unrealized loss on available-for-sale securities							(135)				(135)
Balance as of March 31, 2021	84,075	\$	8	\$	1,601,784	\$	161	\$	(1,202,916)	\$	399,037

See accompanying notes.

7

ATARA BIOTHERAPEUTICS, INC. Condensed Consolidated Statements of Cash Flows (Unaudited) (In thousands)

		Three Months En	ded Mar	March 31,		
		2022		2021		
Operating activities						
Net loss	\$	(88,105)	\$	(78,335)		
Adjustments to reconcile net loss to net cash used in operating activities:						
Stock-based compensation expense		14,335		12,268		
Depreciation and amortization expense		1,591		2,222		
Amortization of investment premiums		498		360		
Non-cash operating lease expense		627		373		
Asset retirement obligation accretion expense		7		21		
Changes in operating assets and liabilities:						
Accounts receivable		318		(8,360)		
Prepaid expenses and other current assets		609		1,828		
Other assets		51		38		
Accounts payable		(4,588)		9,555		
Accrued compensation		(8,769)		(7,672)		
Accrued research and development expenses		539		(5,158)		
Other current liabilities		(570)		1,510		
Deferred revenue		(525)		6,058		
Operating lease liabilities		(645)		(406)		
Other long-term liabilities		98				
Net cash used in operating activities		(84,529)		(65,698)		
Investing activities						
Purchases of short-term investments		(26,427)		(94,055)		
Proceeds from maturities and sales of short-term investments		89,247		109,751		
Purchases of property and equipment		(2,605)		(2,197)		
Net cash provided by (used in) investing activities		60,215		13,499		
Financing activities						
Proceeds from issuance of common stock through ATM facilities, net		20,539		2,457		
Proceeds from employee stock awards		96		1,749		
Taxes paid related to net share settlement of restricted stock units		(616)		(1,231)		
Principal payments on finance lease obligations		(64)		(66)		
Other financing activities, net		(104)		(17)		
Net cash provided by financing activities		19,851		2,892		
Decrease in cash, cash equivalents and restricted cash		(4,463)		(49,307)		
Cash, cash equivalents and restricted cash at beginning of period		107,478		201,798		
Cash, cash equivalents and restricted cash at end of period	\$	103,015	\$	152,491		
Non-cash investing and financing activities						
Property and equipment purchases included in accounts payable and other accrued liabilities	\$	1,188	\$	1,245		
Accrued costs related to underwritten public offering	\$		\$	176		
Accrued costs related to ATM facilities	\$	23	\$	75		
Supplemental cash flow disclosure	<u></u>					
Cash paid for interest	\$	5	\$	10		
	\$	14	\$	8		
Cash paid for income taxes	\$	14	Ф	8		

See accompanying notes.

ATARA BIOTHERAPEUTICS, INC. Notes to Condensed Consolidated Financial Statements (Unaudited)

1. Description of Business

Atara Biotherapeutics, Inc. ("Atara", "we", "our" or "the Company") was incorporated in August 2012 in Delaware. Atara is a pioneer in T-cell immunotherapy, leveraging its novel allogeneic Epstein-Barr Virus ("EBV") T-cell platform to develop transformative therapies for patients with serious diseases, including solid tumors, hematologic cancers and autoimmune disease.

We have several T-cell immunotherapies in clinical development and are progressing multiple next-generation allogeneic chimeric antigen receptor T-cell ("CAR T") programs. Our most advanced T-cell immunotherapy program, tab-cel® (tabelecleucel), is currently in Phase 3 development, and in October 2021, we entered into a commercialization agreement ("Pierre Fabre Commercialization Agreement") with Pierre Fabre Medicament ("Pierre Fabre"), pursuant to which we granted to Pierre Fabre an exclusive, field-limited license to commercialize and distribute tab-cel in Europe and select emerging markets in the Middle East, Africa, Eastern Europe and Central Asia, following regulatory approval. Atara will retain full rights to tab-cel in other major markets, including North America, Asia Pacific and Latin America. See Note 6 for further information. In December 2020, we entered into a research, development and license agreement ("Bayer License Agreement") with Bayer AG ("Bayer") pursuant to which we granted to ATA2271 and ATA3271. In March 2021, as contemplated under the Bayer License Agreement; and to further advance our collaboration, we entered into (a Manufacturing and Supply Agreement; (ii) a Quality Agreement; and (iv) a Technology Transfer Agreement;"). See Note 6 for further information.

We have licensed rights to T-cell product candidates from Memorial Sloan Kettering Cancer Center ("MSK"), rights related to our next-generation CAR T programs from MSK and from H. Lee Moffitt Cancer Center ("Moffitt"), and rights to know-how and technology from the Council of the Queensland Institute of Medical Research ("QIMR Berghofer"). See Note 8 for further information.

In January 2022, we entered into an asset purchase agreement with FUJIFILM Diosynth Biotechnologies California, Inc. ("FDB") and, for certain limited purposes, FUJIFILM Holdings America Corporation, to sell all of the Company's right, title and interest in and to certain assets related to the Atara T-Cell Operations and Manufacturing facility ("ATOM Facility") located in Thousand Oaks, California for \$100 million in cash, subject to potential post-closing adjustments pursuant to the asset purchase agreement (the "Fujifilm Transaction"). The closing of the Fujifilm Transaction occurred on April 4, 2022, at which time 136 of our ATOM Facility employees transitioned to FDB as part of the transaction. We also entered into a Master Services and Supply Agreement with FDB (the "Fujifilm MSA") which became effective upon the closing and could extend for up to ten years. Pursuant to the Fujifilm MSA, FDB will supply us with specified quantities of our cell therapy product candidates and any products approved by regulatory authorities, manufactured in accordance with cGMP standards. See to Note 7 for further information.

2. Summary of Significant Accounting Policies

Basis of Presentation

The accompanying unaudited condensed consolidated financial statements include the accounts of Atara and its wholly owned subsidiaries and have been prepared in accordance with U.S. generally accepted accounting principles ("U.S. GAAP") and the requirements of the Securities and Exchange Commission ("SEC") for interim reporting. As permitted under those rules, certain footnotes or other financial information that are normally required by U.S. GAAP can be condensed or omitted. These condensed consolidated financial statements have been prepared on the same basis as the Company's annual consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2021. In the opinion of management, the condensed consolidated financial statements reflect all adjustments, consisting only of normal recurring adjustments, which are necessary for a fair presentation of the Company's consolidated financial statements. The results of operations for any interim period are not necessarily indicative of the results to be expected for the full year or any other future period. The condensed consolidated balance sheet as of December 31, 2021 has been derived from audited consolidated financial statements at that date but does not include all of the information required by U.S. GAAP for complete consolidated financial statements.



Liquidity

We have incurred significant operating losses since inception and have relied primarily on public and private equity financings and receipts from license and collaboration agreements to fund our operations. As we continue to incur losses, our transition to profitability will depend on the successful development, approval and commercialization of product candidates and on the achievement of sufficient revenues to support our cost structure. We may never achieve profitability, and unless and until we do, we will need to continue to raise additional capital. We expect that existing cash, cash equivalents and short-term investments as of March 31, 2022 will be sufficient to fund our planned operations for at least the next twelve months from the date of issuance of these financial statements.

Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates, assumptions, and judgments that affect the amounts reported in the financial statements and accompanying notes. Significant estimates relied upon in preparing these financial statements include estimates related to revenue recognition, clinical study and other accruals, stock-based compensation expense and income taxes. Actual results could differ materially from those estimates.

Recent Accounting Pronouncements

We consider the applicability and impact of any recent Accounting Standards Update ("ASU") issued by the Financial Accounting Standards Board ("FASB"). Based on our assessment, the ASUs were determined to be either not applicable or are expected to have minimal impact on our condensed consolidated financial statements.

3. Net Loss per Common Share

Basic net loss per common share is calculated by dividing net loss by the weighted-average number of shares of common stock and pre-funded warrants outstanding during the period, without consideration of common share equivalents. Diluted net loss per common share is computed by dividing net loss by the weighted-average number of shares of common stock, pre-funded warrants and common share equivalents outstanding for the period. The pre-funded warrants are included in the computation of basic and diluted net loss per common share as the exercise price is negligible and the pre-funded warrants are fully vested and exercisable. Common share equivalents are only included in the calculation of diluted net loss per common share when their effect is dilutive.

Potential dilutive securities, which include unvested restricted stock units ("RSUs"), unvested performance-based RSUs and performance-based options to purchase common stock for which established performance criteria have been achieved as of the end of the respective periods, vested and unvested options to purchase common stock and shares to be issued under our employee stock purchase plan ("ESPP"), have been excluded from the computation of diluted net loss per share as the effect is antidilutive. Therefore, the denominator used to calculate both basic and diluted net loss per common share is the same in all periods presented.

The following table represents the potential common shares issuable pursuant to outstanding securities as of the related period end dates that were excluded from the computation of diluted net loss per common share, as their inclusion would have an antidilutive effect:

	As of Mar	ch 31,
	2022	2021
Unvested RSUs	7,932,532	4,900,248
Vested and unvested options	10,857,620	9,030,928
ESPP share purchase rights	220,029	164,954
Total	19,010,181	14,096,130

4. Financial Instruments

Our financial assets are measured at fair valueon a recurring basis using the following hierarchy to prioritize valuation inputs, in accordance with applicable U.S. GAAP:

Level 1: Quoted prices in active markets for identical assets or liabilities that we have the ability to access

Level 2:Observable market-based inputs or unobservable inputs that are corroborated by market data such as quoted prices, interest rates and yield curves

Level 3:Inputs that are unobservable data points that are not corroborated by market data

We review the fair value hierarchy classification on a quarterly basis. Changes in the ability to observe valuation inputs may result in a reclassification of levels of certain securities within the fair value hierarchy. We recognize transfers into and out of levels within the fair value hierarchy in the period in which the actual event or change in circumstances that caused the transfer occurs. There have been no transfers between Level 1, Level 2 and Level 3 in any periods presented.

Financial assets and liabilities are considered Level 2 when their fair values are determined using inputs that are observable in the market or can be derived principally from or corroborated by observable market data such as pricing for similar securities, recently executed transactions, cash flow models with yield curves, and benchmark securities. In addition, Level 2 financial instruments are valued using comparisons to like-kind financial instruments and models that use readily observable market data as their basis. U.S. Treasury, government agency and corporate debt obligations, commercial paper and asset-backed securities are valued primarily using market prices of comparable securities, bid/ask quotes, interest rate yields and prepayment spreads and are included in Level 2.

Financial assets and liabilities are considered Level 3 when their fair values are determined using pricing models, discounted cash flow methodologies, or similar techniques, and at least one significant model assumption or input is unobservable. We have no Level 3 financial assets or liabilities.

The following tables summarize the estimated fair value and related valuation input hierarchy of our available-for-sale securities as of each period end:

As of March 31, 2022:	Input Level	Amortized Unreali		Total Unrealized Gain		lized Unrealized		Unrealized		Total Estimated Fair Value
				(in thous	ands)					
Money market funds	Level 1	\$ 74,767	\$		\$		\$	74,767		
U.S. Treasury obligations	Level 2	93,764		_		(557)		93,207		
Government agency obligations	Level 2	16,082		—		(107)		15,975		
Corporate debt obligations	Level 2	97,559		_		(1,151)		96,408		
Commercial paper	Level 2	10,881		_		_		10,881		
Asset-backed securities	Level 2	8,795				(77)		8,718		
Total available-for-sale securities		 301,848		_		(1,892)		299,956		
Less: amounts classified as cash equivalents		(99,814)		_		_		(99,814)		
Amounts classified as short-term investments		\$ 202,034	\$		\$	(1,892)	\$	200,142		

As of December 31, 2021:	Input Level	1	Amortized Unrealized Unre Cost Gain L		Total realized Loss	Total Estimated Fair Value		
					(in thous	ands)		
Money market funds	Level 1	\$	89,738	\$	_	\$	_	\$ 89,738
U.S. Treasury obligations	Level 2		111,832		1		(138)	111,695
Government agency obligations	Level 2		21,346		_		(23)	21,323
Corporate debt obligations	Level 2		99,757		6		(190)	99,573
Commercial paper	Level 2		36,993		_		_	36,993
Asset-backed securities	Level 2		10,174		1		(25)	10,150
Total available-for-sale securities			369,840		8		(376)	369,472
Less: amounts classified as cash equivalents			(104,488)		_		_	(104,488)
Amounts classified as short-term investments		\$	265,352	\$	8	\$	(376)	\$ 264,984

The amortized cost and fair value of our available-for-sale securities by contractual maturity were as follows:

		As of Marc	022		As of Decem	ber 31	31, 2021	
	Α	Amortized Cost		Estimated Tair Value	1	Amortized Cost		Estimated Fair Value
		(in thou					usands)	
Maturing within one year	\$	243,102	\$	242,264	\$	278,457	\$	278,354
Maturing in one to five years		58,746		57,692		91,383		91,118
Total available-for-sale securities	\$	301,848	\$	299,956	\$	369,840	\$	369,472

We considered the current and expected future global economic and market conditions, including the COVID-19 pandemic and the current events involving Russia and Ukraine, and determined that our investments were not significantly impacted. As of March 31, 2022, no significant facts or circumstances were present to indicate a deterioration in the creditworthiness of the issuers of the available-for-sale securities we hold, and the Company has no requirement or intention to sell these securities before maturity or recovery of their amortized cost basis. For all securities with a fair value less than its amortized cost basis, we determined the decline in fair value below amortized cost basis to be non-credit related and no allowance for losses has been recorded. During the three months ended March 31, 2022 and 2021, we did not recognize any impairment losses on our investments.

We have elected the practical expedient to exclude the applicable accrued interest from both the fair value and the amortized cost basis of our available-for-sale securities for purposes of identifying and measuring an impairment. We present accrued interest receivable related to our available-for-sale securities in prepaid expenses and other current assets, separate from short-term investments on our condensed consolidated balance sheet. As of March 31, 2022 and December 31, 2021, accrued interest receivable was \$0.6 million and \$0.8 million, respectively. We have not written off any accrued interest receivables during the three months ended March 31, 2022 and 2021.

In addition, restricted cash collateralized by money market funds is a financial asset measured at fair value and is a Level 1 financial instrument under the fair value hierarchy. As of March 31, 2022 and December 31, 2021, restricted cash was \$1.3 million and \$1.4 million, respectively.

The following table provides a reconciliation of cash, cash equivalents and restricted cash within the condensed consolidated balance sheets that sum to the total of the same such amounts in the condensed consolidated statement of cash flows:

	Ν	March 31, 2022	Dec	ember 31, 2021		
		(in thousands)				
Cash and cash equivalents	\$	101,669	\$	106,084		
Restricted cash - short term		146		194		
Restricted cash - long term		1,200		1,200		
Total cash, cash equivalents and restricted cash	\$	103,015	\$	107,478		

5. Property and Equipment

Property and equipment consisted of the following as of each period end:

	Μ	larch 31, 2022	Dee	cember 31, 2021		
		(in thousands)				
Leasehold improvements	\$	1,355	\$	50,142		
Lab equipment		14,340		14,060		
Machinery and equipment		604		5,228		
Computer equipment and software		1,173		4,245		
Furniture and fixtures		1,226		2,518		
Construction in progress		662		6,325		
Property and equipment, gross		19,360		82,518		
Less: accumulated depreciation and amortization		(10,251)		(28,738)		
Property and equipment, net	\$	9,109	\$	53,780		

Property and equipment in the table above as of March 31, 2022 exclude amounts classified as held for sale. See Note 7 for additional information. Depreciation and amortization expense was \$1.6 million and \$2.2 million for the three months ended March 31, 2022 and 2021, respectively.

6. Out-license Agreements

Bayer Agreements

Research, Development and License Agreement

In December 2020, we entered into the Bayer License Agreement to develop mesothelin-directed CAR T-cell therapies for the treatment of solid tumors, pursuant to which we granted to Bayer an exclusive, field-limited license under the applicable patents and know-how owned or controlled by us and our affiliates covering or related to ATA2271 and ATA3271 (the "Licensed Products").

Under the terms of the Bayer License Agreement, we will be responsible at our cost for all mutually agreed preclinical and clinical activities for ATA2271 through the first in human Phase 1 clinical study in collaboration with MSK, following which Bayer will be responsible for the further development of ATA2271 at its cost. Bayer will be responsible for the development of ATA3271 at Bayer's cost, except for certain mutually agreed preclinical, translational, manufacturing and supply chain activities to be performed by us relating to ATA3271. Bayer will also be solely responsible for commercializing the Licensed Products at its cost.

In December 2020, we received an upfront cash payment of \$45.0 million from Bayer for the exclusive license grant, net of applicable withholding taxes, which were fully recovered in August 2021, and an additional \$15.0 million reimbursement payment for certain research and process development activities to be performed by us. We are also entitled to receive (i) up to an additional \$5.0 million for additional, specified translational activities under the Bayer License Agreement, of which we have invoiced and received \$1.3 million, and (ii) an aggregate of up to \$610.0 million in milestone payments upon achieving certain development, regulatory and commercial milestones relating to the Licensed Products. In addition, we are eligible to receive from Bayer tiered royalties at percentages up to low double digits on worldwide net product sales of the Licensed Products on a country-by-country and product-by-product basis until the later of 12 years after the first commercial sale in such country or the expiration of specified patent rights in such country, subject to certain reductions and aggregate minimum floors.

Bayer and we have formed a joint steering committee ("JSC") that will provide oversight, decision making and implementation guidance regarding the collaboration activities covered under the agreement.

The transaction price at inception consisted of a \$45.0 million upfront payment for the license, \$15.0 million for certain research and process development activities and the \$5.0 million for additional specified translational activities, and this amount was allocated to the single performance obligation. The potential development and commercial milestone payments that we are eligible to receive were excluded from the transaction price, as all milestone amounts were fully constrained based on the probability of achievement. None of the future royalty and sales-based milestone payments were included in the transaction price, as the potential payments represent sales-based consideration. We will reevaluate the transaction price at the end of each reporting period and as uncertain events are resolved or other changes in circumstances occur, and, if necessary, adjust our estimate of the transaction price.

Technology Transfer Agreement

In March 2021, we entered into a Technology Transfer Agreement with Bayer (the "Bayer Tech Transfer Agreement"), which was contemplated as part of the Bayer License Agreement, to transfer to Bayer the ATA3271 manufacturing process being developed as part of the CMC services in the Bayer License Agreement. Upon entering into the agreement, we invoiced Bayer 20 percent of the total fee of \$15.3 million under the Bayer Tech Transfer Agreement, or \$3.1 million, which we received in the second quarter of 2021 and invoiced and received 40 percent of the total fee, or \$6.1 million, in the first quarter of 2022. The remainder of the fee will be billed as follows: (i) 20 percent on January 1, 2023 and (ii) 20 percent upon the technology transfer completion.

Under the Bayer Tech Transfer Agreement, in order to evaluate the appropriate transaction price, we determined that the \$5.3 million fee constituted the entire consideration to be included in the transaction price, and this amount was allocated to the single performance obligation as identified under the Bayer License Agreement.

We utilize a cost-based input method to recognize revenue based on the amount of actual costs incurred relative to the total budgeted costs expected to be incurred for the combined performance obligation.

Manufacturing and Supply Agreement

In March 2021, we entered into a Manufacturing and Supply Agreement with Bayer (the "Bayer Manufacturing Agreement"), which was contemplated as part of the Bayer License Agreement, to manufacture Phase 1 and 2 allogeneic mesothelin-directed CAR T-cell therapies for Bayer to use in clinical trials at a price based on our costs plus a margin, which is consistent with our standalone



selling price. Under the Bayer Manufacturing Agreement, we will also provide storage and distribution services to Bayer at a price that is consistent with our standalone selling price for these services.

Upon entering into the Bayer Manufacturing Agreement, Bayer submitted, and we approved, a binding purchase order for manufacturing services and storage services. Any fees for the manufacturing services will be invoiced as follows: (i) 50 percent upon written acceptance by us of the binding purchase order, and (ii) the remainder upon delivery of the certification of analysis of such lots to Bayer. Storage and distribution services are billed monthly as those services are provided to Bayer.

In March 2021, we invoiced Bayer 50 percent of the total estimated supply price of \$13.1 million for manufacturing services under the initial purchase order for the supply of six lots, or \$6.6 million, which we received in the second quarter of 2021. The remainder of the supply price will be billed upon the release of the lots ordered by Bayer.

Under the Bayer Manufacturing Agreement, in order to evaluate the appropriate transaction price, we determined that the \$3.1 million fee constituted the entire consideration to be included in the transaction price, and this amount was allocated to the single performance obligation as identified under the Bayer License Agreement. Revenue for the manufacturing services for the initial six lots will be recognized based on the amount of actual costs incurred relative to the total budgeted costs expected to be incurred for the combined performance obligation. Revenue for the storage services will be recognized over time as those services are provided. Revenue for the distribution services will be recognized at a point in time when the product is delivered to a clinical site designated by Bayer.

Bayer Revenue Recognition

We utilize a cost-based input method to recognize revenue based on the amount of actual costs incurred relative to the total budgeted costs expected to be incurred for the combined performance obligation. For the three months ended March 31, 2022 and 2021, we recognized license and collaboration revenue of \$6.6 million and \$3.6 million, respectively, under the Bayer Agreements. Deferred revenue related to the Bayer Agreements aggregated to \$50.9 million and \$51.5 million as of March 31, 2022 and December 31, 2021, respectively. The \$50.9 million of deferred revenue as of March 31, 2022 is included in current liabilities and is expected to be recognized over the next year. No development or sales-based milestone payments have been earned or received through March 31, 2022.

Pierre Fabre Commercialization Agreement

In October 2021, we entered into the Pierre Fabre Commercialization Agreement, pursuant to which, we granted to Pierre Fabre an exclusive, field-limited license to commercialize and distribute tab-cel in Europe and select emerging markets in the Middle East, Africa, Eastern Europe and Central Asia (the "Territory") following regulatory approval. Atara will retain full rights to tab-cel in other major markets, including North America, Asia Pacific and Latin America.

We are responsible at our cost for the conclusion of the ongoing Phase 3 ALLELE clinical study and the Phase 2 multi-cohort clinical study. We will also be responsible at our cost for certain other activities directed to obtaining regulatory approval for tab-cel for EBV-positive lymphoproliferative disease pursuant to the terms of the Pierre Fabre Commercialization Agreement in Europe and the UK. Pierre Fabre will be responsible at its cost for obtaining and maintaining all other regulatory approvals and for commercialization and distribution of tab-cel in the Territory. We will own any intellectual property rights developed solely by us under the Agreement.

Pierre Fabre paid us an upfront cash payment of \$45.0 million for the exclusive license grant in the fourth quarter of 2021. We are also entitled to receive an aggregate of up to \$318.0 million in milestone payments upon achieving certain regulatory and commercial milestones. In addition, we are eligible to receive double-digit tiered royalties as a percentage of net sales of tab-cel until the later of 12 years after the first commercial sale in such country, the expiration of specified patent rights, or the expiration of all regulatory exclusivity for such product on a country-by-country basis.

We are negotiating a separate manufacturing and supply agreement with Pierre Fabre for us to manufacture tab-cel for Pierre Fabre to use in the Territory based on a fixed price until January 1, 2024 and cost plus a margin post January 1, 2024. We are responsible for manufacturing and supplying Pierre Fabre with tab-cel for commercialization in the Territory at Pierre Fabre's cost for a minimum of seven years. Following this period, we have the option to transfer the manufacturing responsibility and receive the related manufacturing technology to a third party CMO, and Pierre Fabre may elect to directly assume the manufacturing responsibility and receive the related manufacturing technology.

Pierre Fabre and we have formed a joint steering committee ("JSC") that will provide oversight, decision making and implementation guidance regarding the commercialization activities covered under the agreement.

Under the Pierre Fabre Commercialization Agreement, in order to evaluate the appropriate transaction price, we determined that the \$5 million upfront payment, constituted the entire consideration to be included in the transaction price at the outset of the arrangement. Revenue associated with the upfront fee for the single performance obligation will be deferred until the initial delivery of services related to the manufacture and supply and cell selection and then recognized over the contract performance period . We have not recognized to date any of the \$45.0 million upfront payment as revenue. All of the \$45.0 million of deferred revenue was included in long-term liabilities as of March 31, 2022.



The potential development and commercial milestone payments that we are eligible to receive were excluded from the transaction price, as all milestone amounts were fully constrained based on the probability of achievement. None of the future royalty and sales-based milestone payments were included in the transaction price, as the potential payments represent sales-based consideration. We will reevaluate the transaction price at the end of each reporting period and as uncertain events are resolved or other changes in circumstances occur, and, if necessary, adjust our estimate of the transaction price. No development or commercial milestone payments have been earned or received through March 31, 2022.

7. Assets and Liabilities Held for Sale

In January 2022, we entered into an asset purchase agreement with FDB and, for certain limited purposes, FUJIFILM Holdings America Corporation, to sell all of the Company's right, title and interest in and to certain assets and liabilities related to the ATOM facility for \$100 million in cash, subject to potential post-closing adjustments pursuant to the asset purchase agreement. As a result, certain assets and liabilities related to the ATOM facility were classified as held for sale. Upon classification as held for sale, we assessed the disposal group's fair value less costs to sell and determined that no impairment existed. The operations of the ATOM facility did not meet the criteria to be presented as discontinued operations. The closing of the Fujifilm Transaction occurred on April 4, 2022, at which time, we received the \$100 million of cash and assigned the lease for the ATOM Facility to FDB in connection with the closing of the Fujifilm Transaction. The carrying amounts of the assets and liabilities classified as held for sale as of March 31, 2022 included the following;

		rch 31, 2022
	(in the	ousands)
Prepaid expenses and other current assets	\$	190
Total current assets held for sale	<u>\$</u>	190
Property and equipment, net	\$	44,791
Operating lease assets		6,599
Other assets		39
Total long-term assets held for sale	\$	51,429
Other current liabilities	\$	252
Total current liabilities held for sale	\$	252
Operating lagge lightlitigg long term	S	7 272
Operating lease liabilities - long-term Other long-term liabilities	\$	7,373 960
Total long-term liabilities held for sale	\$	8,333

8. Commitments and Contingencies

MSK In-license Agreements

In June 2015, we entered into an exclusive license agreement with MSK for three clinical stage T-cell therapies. We are required to make payments to MSK based on achievement of specified regulatory and sales-related milestones, as well as mid-single-digit percentage tiered royalty payments based on future sales of products resulting from the development of the licensed product candidates, if any. In addition, under certain circumstances, we are required to make certain minimum annual royalty payments to MSK, which are creditable against earned royalties owed for the same annual period. We are also required to pay a low double-digit percentage of any consideration we receive for sublicensing the licensed rights. The license agreement expires on a product-by-product and country-by-country basis on the latest of: (i) expiration of the last licensed pattent rights related to each licensed product, (ii) expiration of any market exclusivity period granted by law with respect to each licensed product, and (iii) a specified number of years after the first commercial sale of the licensed product in each country. Upon expiration of the license agreement, Atara will retain non-exclusive rights to the licensed products.

In May and December 2018, we licensed additional technology from MSK. We are obligated to make additional milestone payments based on achievement of specified development, regulatory and sales-related milestones, as well as mid-single-digit percentage tiered royalty payments based on future sales of products resulting from the development of the licensed product candidates, if any.

In March 2021, we amended and restated our license agreement with MSK tt terminate our license to certain rights and license additional know-how rights not otherwise covered by our existing agreements.

OIMR Berghofer In-license Agreements

In October 2015, we entered into an exclusive license agreement and a research and development collaboration agreement with QIMR Berghofer. Under the terms of the license agreement, we obtained an exclusive, worldwide license to develop and commercialize allogeneic T-cell therapy programs utilizing technology and know-how developed by QIMR Berghofer. In September 2016, the exclusive license agreement and research and development collaboration agreement were amended and restated. Under the amended and restated agreements, we obtained an exclusive, worldwide license to develop and commercialize additional T-cell programs, as well as the option to license additional technology that we exercised in June 2018. We further amended and restated our license agreement and research and development collaboration agreement collaboration agreements with QIMR Berghofer in August 2019 to terminate our license to certain rights. Our current license agreement also provides for various milestone and rought to programs developed under the collaboration. These payments are expensed on a straight-line basis over the related development periods. The agreement also provides for various milestones.

Other In-license and Collaboration Agreements

From time to time, we have entered into other license and collaboration agreements with other parties. For example, we licensed additional rights related to our MSKpartnered next-generation CAR T programs from MSK in May 2018 and we licensed rights related to our next-generation CAR T programs from Moffitt Cancer Center in August 2018, and we agreed to collaborate through sponsored research in connection with each of these licenses. We also licensed rights related to our MSK-partnered nextgeneration CAR T programs from the National Institutes of Health in December 2018.

Milestones and royalties under each of the above agreements are contingent upon future events and will be recorded as expense when it is probable that the milestones will be achieved or royalties are due. As of March 31, 2022 and December 31, 2021, there were no material outstanding obligations for milestones and royalties under our license and collaboration agreements.

CRL Manufacturing Agreement

In December 2019, we entered into a Commercial Manufacturing Services Agreement (the "Manufacturing Agreement") with Cognate BioServices, Inc., which was acquired by Charles River Laboratories Inc. ("CRL") in March 2021. Pursuant to the Manufacturing Agreement, CRL provides manufacturing services for certain of our product candidates. The initial term of the Manufacturing Agreement, as amended, runs until May 31, 2022, and we are currently in negotiations with CRL to extend the term of the Manufacturing Agreement. We may terminate the Manufacturing Agreement for convenience on six months' written notice to CRL, or immediately if CRL is unable to perform the services under the Manufacturing Agreement or fails to obtain or maintain certain necessary approvals.

Fujifilm Master Services and Supply Agreement

In January 2022, we entered into a Master Services and Supply Agreement with FDB (Fujifilm MSA) which became effective upon the closing of the sale of the ATOM facility on April 4, 2022 and could extend for up to 10 years. Pursuant to the Fujifilm MSA, FDB will supply us with specified quantities of our cell therapy products (if approved) and product candidates, manufactured in accordance with cGMP standards. The Fujifilm MSA does not obligate us to purchase products (if approved) and product candidates exclusively from FDB. We are required to contract services with FDB under the Fujifilm MSA the minimum amounts set forth below:

Calendar Year	Closin	ig Commitment
2022	\$	25,273
2023		33,275
2024		28,179
2025		27,645
2026		24,189
2027 (First quarter)		6,229
Total commitment	\$	144,790

Other Research, Development and Manufacturing Agreements

We may enter into other contracts in the normal course of business with clinical research organizations for clinical trials, with contract manufacturing organizations for clinical supplies, and with other vendors for pre-clinical studies, supplies and other services for our operating purposes. These contracts generally provide for termination on notice.

As of March 31, 2022 and December 31, 2021, there wereno material amounts accrued related to contract termination charges or minimum purchase volumes not being met.



Indemnification Agreements

In the normal course of business, we enter into contracts and agreements that contain a variety of representations and warranties and provide for indemnification for certain liabilities. The exposure under these agreements is unknown because it involves claims that may be made against us in the future but have not yet been made. To date, we have not paid any claims or been required to defend any action related to our indemnification obligations. However, we may record charges in the future as a result of these indemnification obligations. We also have indemnification obligations to our directors and executive officers for specified events or occurrences, subject to some limits, while they are serving at our request in such capacities. There have been no claims to date, and we believe the fair value of these indemnification agreements to be minimal. Accordingly, we did not record liabilities for these agreements as of March 31, 2022 and December 31, 2021.

Contingencies

From time to time, we may be involved in legal proceedings, as well as demands, claims and threatened litigation, which arise in the normal course of our business or otherwise. The ultimate outcome of any litigation is uncertain and unfavorable outcomes could have a negative impact on our results of operations and financial condition. Regardless of outcome, litigation can have an adverse impact on us because of the defense costs, diversion of management resources and other factors. We are not currently involved in any material legal proceedings.

9. Stockholders' Equity

Our authorized capital stock consists of 520,000,000 shares, all with a par value of \$0.0001 per share, of which 500,000,000 shares are designated as common stock and 20,000,000 shares are designated as preferred stock. There were no shares of preferred stock outstanding as of March 31, 2022 and December 31, 2021.

Equity Offerings

As part of our July 2019 underwritten public offering, we issued and sold pre-funded warrants to purchase2,945,026 shares of common stock in an underwritten public offering pursuant to a shelf registration on Form S-3. Each pre-funded warrant entitles the holder to purchase one share of common stock at an exercise price of \$0.0001 per share and expires seven years from the date of issuance. These warrants were recorded as a component of stockholders' equity within additional paid-in capital. Per the terms of the warrant agreement, a holder of the outstanding warrants is not entitled to exercise any portion of any pre-funded warrant if, upon exercise of the warrant, the holder's ownership (together with its affiliates) of our common stock or combined voting power of our securities beneficially owned by such holder (together with its affiliates) would exceed 9.99% after giving effect to the exercise ("Maximum Ownership Percentage"). Upon at least61 days' prior notice to us by the holder, any holder may increase or decrease the Maximum Ownership Percentage not to exceed 19.99%. As of March 31, 2022, pre-funded warrants to purchase2,888,526 shares of our common stock from the July 2019 underwritten public offering were outstanding.

In the second and fourth quarters of 2020, we issued and sold pre-funded warrants to purchase2,866,961 and 2,040,816 shares of common stock, respectively, in underwritten public offerings pursuant to a shelf registration on Form S-3, with terms similar to those described above. As of March 31, 2022, all of the pre-funded warrants issued and sold as part of the 2020 underwritten public offerings were outstanding.

ATM Facility

In November 2021, we entered into a sales agreement (the "2021 ATM Facility") with Cowen and Company, LLC ("Cowen"), which provides for the sale, in our sole discretion, of shares of our common stock having an aggregate offering price of up to \$100.0 million through Cowen, as our sales agent. The issuance and sale of these shares by us pursuant to the 2021 ATM Facility are deemed "at the market" offerings as defined in Rule 415 under the Securities Act of 1933, as amended (the "Securities Act"), and are registered under the Securities Act. We pay a commission of up to 3.0% of gross sales proceeds of any common stock sold under the 2021 ATM Facility.

During the three months ended March 31, 2022, we sold an aggregate of 1,319,878 shares of common stock under the 2021 ATM Facility, at an average price of \$5.88 per share, for gross proceeds of \$21.0 million and net proceeds of \$20.5 million, after deducting commissions and other offering expenses payable by us. As of March 31, 2022, \$57.4 million of common stock remained available to be sold under the 2021 ATM Facility, subject to certain conditions as specified in the sales agreements.

Equity Incentive Plans

Under the terms of the 2014 Equity Incentive Plan, as amended ("2014 EIP"), we may grant stock options, restricted stock awards ("RSAs") and RSUs to employees, directors, consultants and other service providers. RSUs generally vest over four years.

We have granted performance-based RSUs to certain of our employees that provide for the issuance of common stock if specified Company performance criteria related to our clinical programs are achieved. The number of performance-based RSUs that ultimately vests depends upon if, when and which performance criteria are achieved, as well as the employee's continuous service, as defined in the 2014 EIP, through the date of vesting. The fair value of performance-based RSUs is determined as the closing stock price on the date of grant.

Stock options are granted at prices no less than100% of the estimated fair value of the shares on the date of grant as determined by the board of directors, provided, however, that the exercise price of an option granted to a 10% shareholder cannot be less than 110% of the estimated fair value of the shares on the date of grant. Options granted generally vest over four years and expire in seven to ten years. We have granted performance-based stock options to certain of our employees that provide for the issuance of a right to purchase a share of common stock if specified Company performance criteria related to product candidate partnerships are achieved. The vesting of performance-based stock options depends upon if and when the performance criteria are achieved, as well as the employee's continuous service as defined in the 2014 EIP, through the date of vesting.

As of March 31, 2022, a total of 20,275,111 shares of common stock were reserved for issuance under the 2014 EIP, of which3,873,561 shares were available for future grant and 16,401,550 shares were subject to outstanding options and RSUs, including performance-based awards.

In February 2018, we adopted the 2018 Inducement Plan ("Inducement Plan"), under which we may grant options, stock appreciation rights, RSAs and RSUs to new employees. In November 2020, we amended the Inducement Plan to reserve an additional 1,500,000 shares of the Company's common stock for issuance under the Inducement Plan. In September 2021, we further amended the Inducement Plan to reserve an additional 1,500,000 shares of the Company's common stock for issuance under the Inducement Plan.

As of March 31, 2022, 4,002,970 shares of common stock were reserved for issuance under the Inducement Plan, of whichl, 101,975 shares were available for future grant and 2,900,995 shares were subject to outstanding options and RSUs.

Restricted Stock Units

The following is a summary of RSU activity under our 2014 EIP and Inducement Plan:

	RSUs				
	Shares		Weighted Average Grant Date Fair Value		
Balance as of December 31, 2021	5,592,358	\$	16.22		
Granted	3,368,090	\$	10.37		
Forfeited	(367,370)	\$	15.09		
Vested	(446,696)	\$	22.22		
Balance as of March 31, 2022	8,146,382	\$	13.53		

As of March 31, 2022, there was \$98.9 million of unrecognized stock-based compensation expense related to RSUs that is expected to be recognized over a weighted average period of 3.1 years. This excludes unrecognized stock-based compensation expense for performance-based RSUs that were deemed not probable of vesting in accordance with U.S GAAP.

Stock Options

The following is a summary of stock option activity under our 2014 EIP and Inducement Plan:

		Weighted Average Remaining Weighted Average Contractual Term				Aggregate Intrinsic Value			
	Shares		Exercise Price	(Years)		(in thousands)			
Balance as of December 31, 2021	9,219,837	\$	20.81	6.4	\$	12,810			
Granted	2,352,082		10.46						
Exercised	(10,000)		9.58						
Forfeited or expired	(405,756)		24.99						
Balance as of March 31, 2022	11,156,163	\$	18.49	6.8	\$	328			

Aggregate intrinsic value represents the difference between the closing stock price of our common stock on March 31, 2022 and the exercise price of outstanding, inthe-money options. As of March 31, 2022, there was \$47.4 million of unrecognized stock-based compensation expense related to stock options that is expected to be recognized over a weighted average period of 2.9 years. This excludes unrecognized stock-based compensation expense for performance-based stock options that were deemed not probable of vesting in accordance with U.S GAAP.

Reserved Shares

The following shares of common stock were reserved for future issuance under our equity incentive plans as of March 31, 2022:

	Total Shares Reserved
2014 Equity Incentive Plan	20,275,111
2018 Inducement Plan	4,002,970
2014 Employee Stock Purchase Plan	1,093,151
Total reserved shares of common stock	25,371,232

Stock-based Compensation Expense

Total stock-based compensation expense related to all employee and non-employee stock awards was as follows:

	Three Months Ended March 31,						
	 2022						
	 (in thousands)						
Research and development	\$ 8,506	\$		7,530			
General and administrative	5,829			4,738			
Total stock-based compensation expense	\$ 14,335	\$		12,268			

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion and analysis of our financial condition and results of operations together with our condensed consolidated financial statements and related notes included elsewhere in our Quarterly Report on Form 10-Q for the quarter ended March 31, 2022. This discussion and other parts of this Quarterly Report contain forward-looking statements that involve risk and uncertainties, such as statements of our plans, objectives, expectations and intentions. As a result of many factors, including those factors set forth in the "Risk Factors" section of this Quarterly Report, our actual results could differ materially from the results described in or implied by the forward-looking statements contained in the following discussion and analysis.

Overview

Atara Biotherapeutics is a pioneer in T-cell immunotherapy, leveraging its novel allogeneic Epstein-Barr virus (EBV) T-cell platform to develop transformative therapies for patients with serious diseases, including solid tumors, hematologiccancers and autoimmune disease. With our lead program in Phase 3 clinical development, we are the most advanced allogeneic T-cell immunotherapy company and intend to rapidly deliver off-the-shelf treatments to patients with high unmet medical need. Our platform leverages the unique biology of EBV T cells and has the capability to treat a wide range of EBV-driven diseases or other serious diseases through incorporation of engineered chimeric antigen receptors (CARs) or T-cell receptors (TCRs). Atara is applying this one platform, that does not require TCR or HLA gene editing, to create a robust pipeline. Our strategic priorities are:

- Tab-cel[®]: Atara's most advanced T-cell immunotherapy program, tab-cel (tabelecleucel), is partnered with Pierre Fabre Medicant (Pierre Fabre) for commercialization in Europe and select emerging markets and is currently in Phase 3 development for patients with EBV-driven post-transplant lymphoproliferative disease (EBV+ PTLD) who have failed rituximab or rituximab plus chemotherapy, as well as other EBV-driven diseases;
- ATA188: T-cell immunotherapy targeting EBV antigens believed to be important for the potential treatment of primary and secondary progressive multiple sclerosis;
- CAR T Programs:
 - ATA2271: Autologous CAR T immunotherapy targeting solid tumors expressing the tumor antigen mesothelin, which is partnered with Bayer AG (Bayer);
 - o ATA3271: Allogeneic CAR T immunotherapy currently in preclinical development targeting mesothelin, which is partnered with Bayer; and
 - ATA3219: Allogeneic CAR T targeting CD19, currently in preclinical development, and being developed as a potential best-in-class product, based on a next generation 1XX co-stimulatory domain and the innate advantages of EBV T cells as the foundation for an allogeneic CAR T platform.

Our T-cell immunotherapy platform includes the capability to progress both allogeneic and autologous programs and is potentially applicable to a broad array of targets and diseases. Our off-the-shelf, allogeneic T-cell platform allows for rapid delivery of a T-cell immunotherapy product manufactured in advance of patient need and stored in inventory, with each manufactured lot of cells providing therapy for numerous potential patients. This differs from autologous treatments, in which each patient's own cells must be extracted, genetically modified outside the body and then delivered back to the patient, requiring a complex logistics network. For our allogeneic programs, we select the appropriate set of cells for use based on a patient's unique immune profile. Our CMOs are currently in the process of completing commercial production qualification activities for tab-cel while we build inventory according to our commercial product supply strategy.

In October 2021, we entered into the Commercialization Agreement with Pierre Fabre (Pierre Fabre Commercialization Agreement), pursuant to which we granted to Pierre Fabre an exclusive, field-limited license to commercialize and distribute tab-cel in Europe and select emerging markets in the Middle East, Africa, Eastern Europe and Central Asia following regulatory approval. We will retain full rights to tab-cel in other major markets, including North America, Asia Pacific and Latin America. Under the terms of the Pierre Fabre Commercialization Agreement, we are currently negotiating a Manufacturing and Supply Agreement as well as a number of ancillary agreements to further advance our collaboration with Pierre Fabre.

In December 2020, we entered into a Research, Development and License Agreement with Bayer (the Bayer License Agreement) pursuant to which we granted to Bayer an exclusive, field-limited license under the applicable patents and know-how owned or controlled by us and our affiliates covering or related to ATA2271 and ATA3271. In March 2021, as contemplated by the Bayer License Agreement, we entered into (i) a Manufacturing and Supply Agreement (Bayer Manufacturing Agreement); (ii) a Pharmacovigilance Agreement; (iii) a Quality Agreement; and (iv) a Technology Transfer Agreement, in each case, with Bayer, to further advance our collaboration with Bayer.



We have also entered into research collaborations with leading academic institutions such as Memorial Sloan Kettering Cancer Center (MSK), the Council of the Queensland Institute of Medical Research (QIMR Berghofer) and H. Lee Moffitt Cancer Center and Research Institute (Moffitt) pursuant to which we acquired rights to novel and proprietary technologies and programs.

Our research facilities in Thousand Oaks, California (ARC) and Denver, Colorado contain our translational and pre-clinical sciences, analytical development and process science functions. These facilities support our product pipeline, process development and leverage our allogeneic cell therapy platform to drive innovation.

In January, 2022, we entered into an asset purchase agreement with FUJIFILM Diosynth Biotechnologies California, Inc. (FDB) and, for certain limited purposes, FUJIFILM Holdings America Corporation, to sell all of the Company's right, title and interest in and to certain assets related to the Atara T-Cell Operations and Manufacturing facility (ATOM Facility) located in Thousand Oaks, California for \$100 million in cash, subject to potential post-closing adjustments pursuant to the asset purchase agreement (the Fujifilm Transaction). The closing of the Fujifilm Transaction occurred on April 4, 2022, at which time we assigned the lease for the ATOM Facility to FDB in connection with the closing of the Fujifilm Transaction. We also entered into a Master Services and Supply Agreement with FDB (Fujifilm MSA) which became effective upon the closing and could extend for up to 10 years. Pursuant to the Fujifilm MSA, FDB will supply us with specified quantities of our cell therapy products (if approved) and product candidates, manufactured in accordance with cGMP standards. The Fujifilm MSA does not obligate us to purchase products (if approved) and product candidates exclusively from FDB. We currently have on hand sufficient tab-cel drug product inventory to supply commercial demand, if approved and subject to the specifications set forth in such approval, for at least 12 months.

In addition to the Fujifilm MSA, we also work with Charles River Laboratories (CRL) pursuant to the Manufacturing Agreement that we entered into with Cognate in December 2019. In March 2021, CRL acquired Cognate. Pursuant to the Manufacturing Agreement, CRL provides manufacturing services for certain of our product candidates. The initial term of the Manufacturing Agreement, as amended, runs until May 31, 2022, and we are in negotiations with CRL to extend the term of the Manufacturing Agreement.

Pipeline

Tab-cel®

We continue to advance development of tab-cel in a Phase 3 clinical trial for patients with EBV+ PTLD (ALLELE), which has received Breakthrough Therapy Designation (BTD) in the U.S. for treatment of rituximab-refractory EBV-associated lymphoproliferative disorders, PRIority MEdicines (PRIME) designation in the European Union (EU), and orphan drug designation in the U.S. for treatment of EBV+ PTLD.

- In the first quarter of 2022, the EMA transitioned our EU marketing authorization application (MAA) fortab-cel in patients with EBV+ PTLD from accelerated assessment to standard assessment. The EMA informed us additional time was needed to adequately review our responses to the EMA's questions. We anticipate a decision with respect to potential approval of the MAA in the fourth quarter of 2022.
- In connection with a potential BLA submission for tab-cel in the United States, we have been in discussions with the FDA on the content of chemistry, manufacturing and controls (CMC) module 3, including the assessment of comparability between the product used in the pivotal ALLELE study and that intended for commercialization.
- In late February 2022, we held a Type B CMC meeting with the FDA to discuss and potentially align on comparability between commercial and pivotal clinical trial process versions. We did not reach alignment with the FDA on comparability and the FDA has recommended we conduct an additional clinical study with commercial product as FDA does not agree that comparability has been demonstrated between product used in the pivotal ALLELE study and the intended commercial product.
- We remain in active dialogue with the FDA and have made further progress on discussing proposals to enable a potential BLA filing that does not require a new Phase 3 clinical study.
- We continue to adapt our investment in pre-commercial activities and are continuing our commercial readiness activities based on expected timing of potential approval and commercialization of tab-cel in the U.S.
- We continue to pursue development of tab-cel in additional patient populations, with a primary focus on immunodeficiency-associated lymphoproliferative diseases (IA-LPDs), given the commonality of their EBV-driven mechanism of disease in immunocompromised patients, high unmet medical need and positive clinical data to date with tab-cel.

- We continue to open sites and enroll patients in our Phase 2 multi-cohort study in six patient populations, including four within IA-LPDs and two in other EBV-driven diseases, in both the U.S. and EU. Data from this study is expected in 2023.
- Due to the evolving treatment landscape of EBV-driven nasopharyngeal carcinoma (NPC), we are not actively conducting any development activities while we
 reassess our approach and the development and regulatory pathways for patients with platinum resistant or recurrent EBV-driven NPC.

ATA188

We continue to progress our development of ATA188, which has received fast track designation from the FDA for treatment of primary progressive MS (PPMS) and secondary progressive MS (SPMS), an allogeneic T-cell immunotherapy targeting EBV antigens believed to be important for the potential treatment of multiple sclerosis (MS).

- We are currently progressing an open-label extension (OLE) of our open label, single arm, multi-center, multi-national Phase 1 study of allogeneic ATA188 for patients with PPMS and SPMS.
- We continue to advance enrollment in our Phase 2, randomized, double-blind, placebo-controlled dose-expansion trial (EMBOLD) to evaluate the efficacy and safety of ATA188 in patients with PPMS and SPMS.
- We plan to conduct a formal interim analysis of the EMBOLD study (IA) in June 2022 to assess efficacy, safety and biomarker data to further inform our current development strategy and whether adjustments to sample size are needed. We expect to complete target enrollment in the EMBOLD study soon after conducting the IA.
- After the IA is conducted, we plan to communicate next steps for the program in July 2022, including rationale, while maintaining the integrity of the EMBOLD study.
- We also plan to meet with the FDA following the IA to share the data and to discuss next steps for the development pathway.
- In March 2022, we conducted a EBV and MS Day, where we presented updated Phase 1 and OLE data that demonstrated 20 out of 24 patients have had either EDSS improvement or EDSS stability throughout their observation in the study with up to 42 months follow-up; 33% of patients in the high-dose cohorts achieved confirmed expanded disability status scale (EDSS) improvement at the 12-month timepoint.

ATA2271/ATA3271

Our next-generation CAR T immunotherapy programs include autologous ATA2271 and allogeneic ATA3271 targeting mesothelin, which is a tumor antigen expressed on a number of solid tumors including mesothelioma, ovarian cancer, pancreatic cancer, non-small cell lung cancer and other tumors over-expressing mesothelin. Both programs were licensed to Bayer in December 2020, pursuant to an exclusive, field-limited license (the Bayer License Agreement).

- MSK is currently conducting an open-label, single-arm Phase 1 clinical study of ATA2271 for patients with advanced mesothelioma.In February 2022, MSK
 notified the FDA of a fatal serious adverse event (SAE) in a patient treated in the third, higher dose cohort in this study. MSK has voluntarily paused enrollment
 of new patients in this study on a temporary basis while additional information regarding this case is gathered and reviewed. The FDA notified MSK of its
 agreement with MSK's decision. MSK continues to analyze the data and we expect MSK to share the results with the FDA when appropriate, in addition to any
 intended informed consent and/or protocol amendments. Subject to the outcome of this review and resumption of enrollment of new patients in this study, we
 expect to provide a data update from this Phase 1 study in the second half of 2022.
- We continue to advance IND-enabling studies for ATA3271, an off-the-shelf, allogeneic CAR T therapy targeting mesothelin using a PD-1 DNR and 1XX CAR co-stimulatory signaling domain through our EBV T-cell platform. We expect our partner Bayer to file the IND in the fourth quarter of 2022. We do not believe the temporary pause in the ATA2271 study enrollment impacts the IND-enabling work to advance ATA3271, a separate, off-the-shelf, allogeneic CAR T therapy. In preclinical data for ATA3271, we observed anti-tumor activity that we believe indicates functional persistence and significant survival benefit, and we found no evidence of allocytotoxicity in vivo, suggesting that allogeneic MSLN-CAR-engineered EBV T cells are a promising approach for the treatment of MSLN-positive cancers.



ATA3219

We are also developing ATA3219, a potential best-in-class, allogeneic CD19 CAR T immunotherapy targeting B-cell malignancies, leveraging our next-generation 1XX CAR co-stimulatory signaling domain and EBV T-cell platform and does not require TCR or human leukocyte antigen (HLA) gene editing.

We continue to make progress on IND-enabling studies and plan to submit an IND for ATA3219 for patients with B-cell malignancies in the fourth quarter of 2022.

Additional Programs and Platform Expansion Activities

In addition to the prioritized programs described above, we have a number of preclinical programs. We are collaborating with Moffitt to develop ATA2431, a multitargeted CAR T immunotherapy targeting B-cell malignancies. We are also collaborating with QIMR Berghofer to develop a potential next generation EBV vaccine which is differentiated from earlier EBV vaccine efforts that solely focused on B cell responses to EBV.

We believe our platform will have utility beyond the current set of targets to which it has been directed. We continue to evaluate additional product candidates, including those derived from collaborations with our partners. We also continue to evaluate opportunities to license or acquire additional product candidates or technologies to enhance our existing platform.

Manufacturing

On January 26, 2022, we entered into an asset purchase agreement with FDB, and, for certain limited purposes, FUJIFILM Holdings America Corporation, to sell all of the Company's right, title and interest in and to certain assets related to the ATOM Facility for \$100 million in cash, pursuant to the Fujifilm Transaction. The closing of the Fujifilm Transaction occurred on April 4, 2022. We also entered into the Fujifilm MSA which became effective upon the closing and could extend for up to 10 years. Pursuant to the Fujifilm MSA, FDB will supply us with specified quantities of our cell therapy products (if approved) and product candidates, manufactured in accordance with cGMP standards. The Fujifilm MSA does not obligate us to purchase products (if approved) and product candidates exclusively from FDB. We currently have on hand sufficient tabcel drug product inventory to supply commercial demand, if approved and subject to the specifications set forth in such approval, for at least 12 months.

In addition to the Fujifilm MSA, we also work with CRL pursuant to the Manufacturing Agreement that we entered into with Cognate in December 2019. In March 2021, CRL acquired Cognate. Pursuant to the Manufacturing Agreement, CRL provides manufacturing services for certain of our product candidates. The initial term of the Manufacturing Agreement, as amended, runs until May 31, 2022, and we are in negotiations with CRL to extend the term of the Manufacturing Agreement.

COVID-19 Business Update

We continue to closely monitor the impact of the ongoing and evolving COVID-19 pandemic on our business and operations and have taken steps designed to ensure the health and safety of our employees, staff, clinical site staff and patients and to maintain business continuity. Based on guidance issued by federal, state and local authorities, we have temporarily transitioned most of our workforce to a remote, work-from-home model, while maintaining essential in-person laboratory functions in order to advance key research, development and manufacturing priorities. We implemented safety protocols and procedures to support our onsite workforce.

In addition to implementing measures designed to protect the health and safety of our workforce, our clinical study and operational teams are working closely with clinical sites to support the safety of site staff and patients as well as preserve data integrity and access to treatment as appropriate. Where needed, remote study visits, tele-medicine, home health care, and other methods have been leveraged to ensure continuity of care for patients while preserving key endpoint data.

To date, the COVID-19 pandemic has not materially impacted our or our partners' clinical, research and development, regulatory and manufacturing operations or timelines. However, at the onset of the pandemic, we experienced, and depending on the evolving impacts of the ongoing COVID-19 pandemic, we may again experience, some transient delays in clinical study operations, as a result COVID-19.

The full extent to which the COVID-19 pandemic may impact our business and operations is subject to future developments which are uncertain and difficult to predict. We continue to monitor the impact of the COVID-19 pandemic on our business and operations and will adjust our activities as appropriate.



For additional information about risks and uncertainties related to the COVID-19 pandemic that may impact our business, financial condition and operations, see the section titled "Risk Factors" under Part II, Item 1A in this Quarterly Report on Form 10-Q.

Financial Overview

We have a limited operating history. Since our inception in 2012, we have devoted substantially all of our resources to identify, acquire and develop our product candidates, including conducting preclinical and clinical studies, acquiring or manufacturing materials for clinical studies and providing general and administrative support for these operations.

Our net losses were \$88.1 million and \$78.3 million for the three months ended March 31, 2022 and 2021, respectively. As of March 31, 2022, we had an accumulated deficit of \$1.6 billion. Substantially all of our net losses have resulted from costs incurred in connection with our research and development programs and from general and administrative expenses associated with our operations. As of March 31, 2022, our cash, cash equivalents and short-term investments totaled \$301.8 million, which we intend to use to fund our operations.

Revenues

We have never generated revenues from the commercial sale of products and have incurred losses since inception. We do not expect to receive any revenue from product sales unless and until we obtain regulatory approval for and commercialize one of our current or future product candidates. Our revenues to date are derived solely from agreements with Bayer and Pierre Fabre and include upfront license fees, fees for research, process development and translational activities and technology transfer fees.

We expect that any revenue we generate from our Bayer Agreements, the Pierre Fabre Commercialization Agreement and any future collaboration and research and license partners will fluctuate from year to year as a result of the timing and number of milestones and other payments.

Research and Development Expenses

The largest component of our total operating expenses since inception has been our investment in research and development activities, including the preclinical and clinical development of our product candidates. Research and development expenses consist primarily of compensation and benefits for research and development employees, including stock-based compensation; expenses incurred under agreements with contract research organizations and investigative sites that conduct preclinical and clinical studies; the costs of acquiring and manufacturing clinical study materials and other supplies; payments under licensing and research and development agreements; other outside services and consulting costs; and facilities, information technology and overhead expenses. Research and development costs are expensed as incurred.

We plan to continue investment in the development of our product candidates. Our current planned research and development activities include the following:

- continuing to initiate sites and enroll patients in our Phase 3 clinical study oftab-cel for the treatment of patients with EBV+ PTLD after HCT and SOT who have failed rituximab;
- process development, testing and manufacturing of drug supply to support clinical and IND-enabling studies;
- continuing development of ATA188 in PMS;
- continuing to develop product candidates based on our next-generation CAR T programs;
- continuing to develop our product candidates in additional indications, including tab-cel for EBV+ cancers;
- · continuing to develop other pre-clinical product candidates; and
- leveraging our relationships and experience to in-license or acquire additional product candidates or technologies.

In addition, we believe it is important to invest in the development of new product candidates to continue to build the value of our product candidate pipeline and our business. We plan to continue to advance our most promising early product candidates into preclinical development with the objective to advance these early-stage programs to human clinical studies over the next several years.

Our expenditures on current and future preclinical and clinical development programs are subject to numerous uncertainties in timing and cost to completion. The duration, costs, and timing of clinical studies and development of our product candidates will depend on a variety of factors, including:

- the availability of qualified drug supply for use in our ongoing Phase 3 or other clinical studies;
- the scope, rate of progress, and expenses of our ongoing clinical studies, potential additional clinical studies and other research and development activities;
- the potential review or reanalysis of our clinical study results;
- future clinical study results;
- uncertainties in clinical study enrollment rates or discontinuation rates of patients, including any potential impact of the COVID-19 pandemic;
- potential additional safety monitoring or other studies requested by regulatory agencies;
- changing medical practice patterns related to the indications we are investigating;
- significant and changing government regulation;
- disruptions caused by man-made or natural disasters or public health pandemics or epidemics, including, for example, the COVID-19 pandemic; and
- the timing and receipt of any regulatory approvals, as well as potential post-market requirements.

The process of conducting the necessary clinical research to obtain approval from the FDA and other regulators is costly and time consuming and the successful development of our product candidates is highly uncertain. The risks and uncertainties associated with our research and development projects are discussed more fully in the section of this report titled "1A. Risk Factors." As a result of these risks and uncertainties, we are unable to determine with any degree of certainty the duration and completion costs of our research and development projects, or if, when, or to what extent we will generate revenues from the commercialization and sale of any of our product candidates that obtain regulatory approval. We may never succeed in achieving regulatory approval for any of our product candidates.

General and Administrative Expenses

General and administrative expenses consist primarily of compensation and benefits for legal, human resources, finance, commercial and other general and administrative employees, including stock-based compensation; professional services costs, including legal, patent, human resources, audit and accounting services; other outside services and consulting costs, including those related to pre-commercial activities; and information technology and facilities costs.

Interest and Other Income (Expense), net

Interest and other income (expense), net consists primarily of interest earned on our cash, cash equivalents and short-term investments and translation gains and losses on transactions denominated in foreign currencies.

Critical Accounting Policies and Significant Judgments and Estimates

There have been no significant changes to our critical accounting policies and significant judgments and estimates during the three months ended March 31, 2022 from those disclosed in our management's discussion and analysis of financial condition and results of operations included in our Annual Report on Form 10-K for the year ended December 31, 2021 filed with the SEC on February 28, 2022.

Results of Operations

Comparison of the Three Months Ended March 31, 2022 and 2021

Revenues

Revenue consisted of the following in the periods presented:

		Three Months Ended						
		March 31,			Increase			
	2022	2022 2021		2021	(Decrease)			
		(in thousands)						
License and collaboration revenue	\$	7,314	\$	3,552	\$	3,762		

License and collaboration revenue increased to \$7.3 million for the three months ended March 31, 2022 compared to \$3.6 million for the three months ended March 31, 2021. The increase primarily related to additional revenue recognized under the Bayer Agreements.

Research and development expenses

Research and development expenses consisted of the following costs, by program, in the periods presented:

	Three Months Ended March 31,				Increase	
		2022 2021		(Decrease)		
	(in thousands)					
Tab-cel [®] expenses	\$	10,626	\$	11,953	\$	(1,327)
ATA188, CAR T and other program expenses		8,379		9,060		(681)
Employee and overhead expenses		55,958		43,046		12,912
Total research and development expenses	\$	74,963	\$	64,059	\$	10,904

Tab-cel expenses were \$10.6 million in the three months ended March 31, 2022, as compared to \$12.0 million in the comparative 2021 period. The decrease in 2022 was primarily due to lower clinical trial and manufacturing spend.

ATA188, CAR T and other program expenses were \$8.4 million in the three months ended March 31, 2022, as compared to \$9.1 million in the comparative 2021 period. The decrease in the three months ended March 31, 2022 was primarily due to milestone and license fees for MSK recorded in the 2021 period related to the ATA2271 CAR T program, partially offset by higher ATA188 clinical trial and development related costs in the 2022 period.

Employee and overhead expenses were \$56.0 million in the three months ended March 31, 2022, as compared to \$43.0 million in the comparative 2021 period. The increase was primarily due to higher compensation-related costs from increased headcount, higher outside services, and higher facility-related costs in support of our continuing expansion of research and development activities. For the three months ended March 31, 2022, payroll and related costs increased by \$8.9 million, facility-related costs increased by \$2.9 million and outside services costs increased by \$1.1 million.

Total research and development expenses for the three months ended March 31, 2022 and 2021 were not significantly impacted as a result of the COVID-19 pandemic.

General and administrative expenses

		Three Mon Marc		1	Increase		
		2022 2021		2021	(Decrease)		
	-	(in thousands)					
neral and administrative expenses	\$	20,571	\$	17,738	\$	2,833	

General and administrative expenses increased to \$20.6 million in the three months ended March 31, 2022, as compared to \$17.7 million in the comparative 2021 period. The increase was primarily driven by higher compensation-related costs from increased headcount.

Total general and administrative expenses for the three months ended March 31, 2022 and 2021 were not significantly impacted as a result of the COVID-19 pandemic.

Liquidity and Capital Resources

Sources of Liquidity

Since our inception in 2012, we have funded our operations primarily through the issuance of common and preferred stock, issuance of pre-funded warrants to purchase common stock and upfront fees from the Bayer License Agreement and the Pierre Fabre Commercialization Agreement.

In November 2021, we entered into a sales agreement (the 2021 ATM Facility) with Cowen, which provides for the sale, in our sole discretion, of shares of our common stock having an aggregate offering price of up to \$100.0 million through Cowen, as our sales agent. We pay a commission of up to 3.0% of gross sales proceeds of any common stock sold under the 2021 ATM Facility.

During the three months ended March 31, 2022, we sold an aggregate of 1,319,878 shares of common stock under the 2021 ATM Facility, at an average price of \$15.88 per share for gross proceeds of \$21.0 million and net proceeds of \$20.5 million, after deducting commissions and other offering expenses payable by us.

As of March 31, 2022, we have \$57.4 million of common stock remained available to be sold under the 2021 ATM Facility, subject to certain conditions as specified in the agreement.

We have incurred losses and negative cash flows from operations in each year since inception. We do not expect to receive any revenues from the sale of products unless and until we obtain regulatory approval for and commercialize any of our product candidates. As such, we anticipate that we will continue to incur losses in the foreseeable future. We expect that our operating expenses will continue to increase. As a result, we will need additional capital to fund our operations, which we may raise through a combination of equity offerings, debt financings, other third-party funding, marketing and distribution arrangements and other collaborations, strategic alliances and licensing arrangements. We may borrow funds on terms that may include restrictive covenants, including covenants that restrict the operation of our business, liens on assets, high effective interest rates and repayment provisions that reduce cash resources and limit future access to capital markets. In addition, we expect to continue to opportunistically seek access to additional funds through additional public or private equity offerings or debt financings including by utilizing the 2021 ATM Facility, through potential collaborations, partnering or other strategic arrangements, or a combination of the foregoing. To the extent that we raise additional capital by issuing equity securities, our stockholders may experience substantial dilution. To the extent that we raise additional funds through some of our rights to our technologies or rights to market and sell our products in certain geographies, grant licenses or other rights on terms that are not favorable to us, or issue equity that may be substantially dilutive to our stockholders.

Cash in excess of immediate requirements is invested in accordance with our investment policy, primarily with a view to liquidity and capital preservation. Currently, our cash, cash equivalents and short-term investments are held in bank and custodial accounts and consist of money market funds, U.S. Treasury, government agency and corporate debt obligations, commercial paper and asset-backed securities.

Our cash, cash equivalents and short-term investments balances as of the dates indicated were as follows:

	March 31,		cember 31,	
	2022		2021	
	 (in thousands)			
Cash and cash equivalents	\$ 101,669	\$	106,084	
Short-term investments	200,142		264,984	
Total cash, cash equivalents and short-term investments	\$ 301,811	\$	371,068	

Comparison of the Three Months Ended March 31, 2022 and 2021

The following table details the primary sources and uses of cash for each of the periods set forth below:

	Three Months Ended March 31,		
	 2022	2021	
	 (in thousands)		
Net cash (used in) provided by:			
Operating activities	\$ (84,529) \$	(65,698)	
Investing activities	60,215	13,499	
Financing activities	 19,851	2,892	
Net (decrease) increase in cash, cash equivalents and restricted cash	\$ (4,463) \$	(49,307)	

Operating activities

Net cash used in operating activities was \$84.5 million in the three months ended March 31, 2022 as compared to \$65.7 million in the comparative 2021 period. The increase of \$18.8 million was primarily due to a \$9.8 million increase in net loss and a \$9.0 million increase in net working capital usage.

Investing activities

Net cash provided by investing activities in the three months ended March 31, 2022 consisted of \$89.2 million received from maturities and sales of available-for-sale securities, offset by \$26.4 million used to purchase available-for-sale securities and \$2.6 million in purchases of property and equipment. Net cash provided by investing activities in the comparative 2021 period consisted of \$109.8 million used to purchase available-for-sale, partially offset by \$94.1 million used to purchase available-for-sale securities and \$2.2 million in purchases of property and equipment.

Financing activities

Net cash provided by financing activities in the three months ended March 31, 2022 consisted primarily of \$20.5 million of net proceeds received from ATM facilities, partially offset by \$0.6 million of taxes paid related to the net share settlement of RSUs. Net cash provided by financing activities in the comparative 2021 period consisted primarily of \$2.5 million of net proceeds received from the ATM facilities and \$1.7 million of net proceeds from employee stock award transactions, partially offset by \$1.2 million of taxes paid related to the net share settlement of RSUs.

Operating Capital Requirements and Plan of Operations

To date, we have not generated any revenue from commercial product sales. We do not know when, or if, we will generate any revenue from commercial product sales. We do not expect to generate any revenue from commercial product sales unless and until we obtain regulatory approval for and commercialize one of our current or future product candidates. We anticipate that we will continue to generate losses for the foreseeable future, and we expect the accumulated losses to increase as we continue the development of and seek regulatory approvals for our product candidates and begin to commercialize any approved products. We are subject to all of the risks inherent in the development of new products, and we may encounter unforeseen expenses, difficulties, complications, delays and other unknown factors that may adversely affect our business. We anticipate that we will need to raise substantial additional funding in connection with our continuing and expected expansion of our operations.

We expect that our existing cash, cash equivalents and short-term investments as of March 31, 2022, together with the \$100.0 million received from FDB on April 4, 2022, will be sufficient to fund our planned operations into the fourth quarter of 2023. In order to complete the process of obtaining regulatory approval for any of our product candidates and to build the sales, marketing and distribution infrastructure that we believe will be necessary to commercialize our product candidates, if approved, we will require substantial additional funding. In addition, we expect to continue to opportunistically seek access to additional funds through additional public or private equity offerings or debt financings, through potential collaborations, partnering or other strategic arrangements, or a combination of the foregoing.

We have based our projections of operating capital requirements on assumptions that may prove to be incorrect and we may use all of our available capital resources sooner than we expect. Because of the numerous risks and uncertainties associated with research, development and commercialization of pharmaceutical products, we are unable to estimate the exact amount of our operating capital requirements. Our future funding requirements will depend on many factors, including, but not limited to:

- the timing, costs and results of our ongoing and planned clinical and preclinical studies for our product candidates;
- our success in establishing and scaling commercial manufacturing capabilities;
- the number and characteristics of product candidates that we pursue;
- the outcome, timing and costs of seeking regulatory approvals;
- subject to receipt of regulatory approval, costs associated with the commercialization of our product candidates and the amount of revenues received from commercial sales of our product candidates;
- the timing of proceeds from the Bayer License Agreement and the Pierre Fabre Commercialization Agreement, as well as the terms and timing of any future collaborations, licensing, consulting or other arrangements that we may establish;
- the amount and timing of any payments we may be required to make in connection with the licensing, filing, prosecution, maintenance, defense and enforcement of any patents or patent applications or other intellectual property rights;
- the extent to which we in-license or acquire other products and technologies; and
- the timing of capital expenditures, including the qualification of our CMOs' manufacturing facilities.

Until we are able to generate a sufficient amount of product revenue and generate positive net cash flows from operations, which we may never do, meeting our longterm capital requirements is in large part reliant on access to public and private equity and debt capital markets, augmented by cash generated from operations and interest income earned on the investment of our cash balances. We expect to continue to seek access to the equity and debt capital markets to support our development efforts and operations. To the extent that we raise additional capital by issuing equity securities, our stockholders may experience substantial dilution. To the extent that we raise additional funds through collaboration or partnering arrangements, we may be required to relinquish some of our rights to our technologies or rights to market and sell our products in certain geographies, grant licenses or other rights on terms that are not favorable to us, or issue equity that may be substantially dilutive to our stockholders.

As a result of economic conditions, general global economic uncertainty, political change and other factors, including the ongoing COVID-19 pandemic, we do not know whether additional capital will be available when needed, or that, if available, we will be able to obtain additional capital on reasonable terms. If we are unable to raise additional capital due to the volatile global financial markets, general economic uncertainty or other factors, we will be forced to delay, limit, reduce or terminate preclinical studies, clinical studies or other development activities for one or more of our product candidates.

Contractual Obligations and Commitments

We lease our corporate headquarters in South San Francisco, California under a non-cancellable lease agreement for approximately 13,670 square feet of office space. In October 2020, we entered into an amendment of this lease to extend the lease term by one year and in December 2021, we entered into a further amendment to extend the lease for an additional three years. The amended lease expires in May 2025. The contractual obligations during the extended lease term are not material. In connection with this lease, we are required to issue a letter of credit in the amount of \$0.1 million to the landlord, which is recorded as restricted cash in our condensed consolidated balance sheet.

In February 2017, we entered into a lease agreement (ATOM Lease) for approximately 90,580 square feet of office, lab and cellular therapy manufacturing space in Thousand Oaks, California. The initial 15-year term of this lease commenced in February 2018, and the contractual obligations during the initial term are \$16.4 million in aggregate. We have the option to extend this lease for two additional periods of ten and nine years, respectively, after the initial term. In connection with this lease, we were required to issue a letter of credit in the amount of \$1.2 million to the landlord, which is recorded as long-term restricted cash in our condensed consolidated balance sheet. In April 2022, we assigned the ATOM Lease to FDB in connection with the closing of the Fujifilm Transaction.

In November 2018, we entered into a lease agreement for approximately 51,160 square feet of office space in Thousand Oaks, California. The initial term of this lease expires in February 2026. The contractual obligations during the initial term are \$8.5 million in aggregate. We have the option to extend the lease for an additional period of five years after the initial term.

In May 2019, we entered into a new lease agreement for approximately 8,800 square feet of office and lab space in Aurora, Colorado. The initial term of this lease expires in April 2024. In February 2021, we further amended this lease to add an additional 2,861 square feet of lab space. The contractual obligations during the lease term are not material. We have the option to extend this lease for two additional five-year periods after the initial term.



In March 2021, we entered into a new lease agreement for approximately 33,659 square feet of office, lab and warehouse space in Thousand Oaks, California. The initial 10.5-year term of this lease commenced in August 2021 and the contractual obligations during the initial term are \$21.0 million in aggregate. We have the option to extend this lease for two additional five-year periods after the initial term.

Our contractual obligations primarily consist of our obligations under non-cancellable operating and finance leases and contracts we enter into in the normal course of business with clinical research organizations for clinical studies, with contract manufacturing organizations for clinical supplies, and with other vendors for preclinical studies and supplies and other services and products for operating purposes. These contracts generally provide for termination on notice, with the exception of one of our contract manufacturing agreements which we may terminate for convenience upon six months' written notice. Other than the assignment of the ATOM Lease described above, there have been no material changes to our contractual obligations and commitments reported in our Annual Report on Form 10-K for the year ended December 31, 2021 filed with the SEC on February 28, 2022.

Off-Balance Sheet Arrangements

We did not have, and we do not currently have, any off-balance sheet arrangements, as defined in the rules and regulations of the SEC, during the periods presented.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

During the three months ended March 31, 2022, there were no material changes to our interest rate risk disclosures, market risk disclosures and foreign currency exchange rate risk disclosures reported in our Annual Report on Form 10-K for the year ended December 31, 2021 filed with the SEC on February 28, 2022.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Under the supervision of our Chief Executive Officer and Chief Financial Officer, we evaluated the effectiveness of our disclosure controls and procedures, as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act as of March 31, 2022. Based on that evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that our disclosure controls and procedures were effective as of March 31, 2022 to ensure that information required to be disclosed by us in the reports we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely discussion regarding required disclosures. In designing and evaluating our disclosure controls and procedures, management recognizes that any disclosure controls and procedures must reflect the fact that there are resource constraints and that management is required to apply its judgment in evaluating the benefits of possible controls and procedures relative to their costs.

Changes in Internal Control over Financial Reporting

There was no change in our internal control over financial reporting identified in connection with the evaluation required by Rules 13a-15(d) and 15d-15(d) of the Exchange Act that occurred during the three months ended March 31, 2022 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.



Item 1. Legal Proceedings

None.

Item 1A. Risk Factors

Investing in our common stock involves a high degree of risk. Investors should carefully consider all of the risk factors and uncertainties described below, in addition to the other information contained in this Quarterly Report on Form 10-Q, including the section of this report titled "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our consolidated and combined financial statements and related notes, before investing in our common stock.

The risks described below may not be the only ones relating to our company and additional risks that we currently believe are immaterial may also affect us. If any of these risks, including those described below, materialize, our business, competitive position, reputation, financial condition, results of operations, cash flows and future prospects could be seriously harmed. In these circumstances, the market price of our common stock could decline, and investors may lose all or a part of their investment.

Risks Related to Our Financial Results and Capital Needs

We have incurred substantial losses since our inception and anticipate that we will continue to incur substantial and increasing losses for the foreseeable future.

We are a clinical-stage biopharmaceutical company. Investment in biopharmaceutical product development is highly speculative because it entails substantial upfront capital expenditures and significant risk that product candidates will fail to prove effective, gain regulatory approval or become commercially viable. We do not have any products approved by regulatory authorities and have not generated any revenues from product sales, and have incurred significant research, development and other expenses related to our ongoing operations and expect to continue to incur such expenses. As a result, we have not been profitable and have incurred significant operating losses in every reporting period since our inception. For the three months ended March 31, 2022, we reported a net loss of \$88.1 million.

We do not expect to generate product revenues in the near future, if at all. We expect to continue to incur significant expenses and operating losses for the foreseeable future. We anticipate these losses to increase as we continue to research, develop and seek regulatory approvals for our product candidates and any additional product candidates we may acquire, in-license or develop, and potentially begin to commercialize product candidates that may achieve regulatory approval. We may encounter unforeseen expenses, difficulties, complications, delays and other unknown factors that may adversely affect our business. The size of our future net losses will depend, in part, on the rate of future growth of our expenses and our ability to generate revenues. If any of our product candidates fails in clinical studies or does not gain regulatory approval, or if approved, fails to achieve market acceptance, we may never become profitable. Even if we achieve profitability in the future, we may not be able to sustain profitability in subsequent periods. We anticipate that our expenses will increase in the future as we continue to invest in research and development of our existing product candidates, investigate and potentially acquire new product candidates and expand our manufacturing and commercialization activities.

We have a limited operating history, which may make it difficult to evaluate the success of our business to date and to assess our future viability.

Our company was formed in August 2012. Our operations to date have been limited to organizing and staffing our company, acquiring product and technology rights and conducting product development activities for our product candidates. We have not yet demonstrated our ability to successfully complete any Phase 2 or Phase 3 clinical studies, obtain regulatory approval, consistently manufacture a commercial scale product or arrange for a third party to do so on our behalf, or conduct sales and marketing activities necessary for successful commercialization for any of our product candidates. In addition, the adoptive immunotherapy technology underlying our T-cell product candidates, including our next-generation CAR T programs, is new and largely unproven. Any predictions about our future success, performance or viability, particularly in view of the rapidly evolving immunotherapy field, may not be as accurate as they could be if we had a longer operating history or approved products on the market.

In addition, as a young business, we may encounter unforeseen expenses, difficulties, complications, delays and other known and unknown factors. If we commercialize tab-cel[®] in the US, subject to submission and approval of a BLA filing for tab-cel by the FDA, we would need to transition from a company with a research and development focus to a company capable of supporting commercial activities. We may not be successful in such a transition. We expect our financial condition and operating results to



continue to fluctuate significantly from quarter to quarter and year to year due to a variety of factors, many of which are beyond our control. Accordingly, any of our quarterly or annual periods' results are not indicative of future operating performance.

We currently have no approved products and thus have no product revenues. We may never generate revenues from the sale of products or achieve profitability.

To date, we have not generated any revenues from commercial product sales. Even if we are able to successfully achieve regulatory approval for our product candidates, we do not know when we will generate revenues from product sales or become profitable, if at all. Our ability to generate revenues from product sales and achieve profitability will depend on our ability to successfully commercialize products, including any of our current product candidates, and other product candidates that we may develop, inlicense or acquire in the future. Our ability to generate revenues from the sale of products and achieve profitability also depends on a number of additional factors, including our ability to:

- successfully complete development activities, including the necessary clinical studies with positive results;
- complete and submit regulatory submissions to the FDA, EMA or other agencies and obtain regulatory approval for indications for which there is a commercial market;
- obtain coverage and adequate reimbursement from third parties, including government and private payors;
- set commercially viable prices for our products, if any;
- develop manufacturing and distribution processes for our novel T-cell immunotherapy product candidates;
- develop commercial quantities of our products at acceptable cost levels;
- establish and maintain adequate supply of our products, including cell lines with sufficient breadth to treat patients;
- establish and maintain manufacturing relationships with reliable third parties or qualify our CMOs' manufacturing facilities such that we can maintain the supply of our products by ensuring adequate, manufacturing of bulk drug substances and drug products in a manner that is compliant with global legal and regulatory requirements;
- achieve market acceptance of our products, if any;
- attract, hire and retain qualified personnel;
- protect our rights in our intellectual property portfolio;
- develop a commercial organization capable of sales, marketing and distribution for any products we intend to sell ourselves in the markets in which we choose
 to commercialize on our own; and
- find suitable distribution partners to help us market, sell and distribute our approved products in other markets.

Our revenues for any product candidate for which regulatory approval is obtained will be dependent, in part, upon the size of the markets in the territories for which we gain regulatory approval, the accepted price for the product, the ability to get reimbursement at any price, and whether we own the commercial rights for that territory. If the number of our addressable disease patients is not as significant as we estimate, the indication approved by regulatory authorities is narrower than we expect, or the reasonably accepted population for treatment is narrowed by competition, physician choice, treatment guidelines or a reduction in the incidence of the addressable disease, we may not generate significant revenues from sales of our products, even if approved. In addition, we anticipate incurring significant costs associated with commercializing any approved product candidate. As a result, even if we generate product revenues, we may not become profitable and may need to obtain additional funding to continue operations. If we fail to become profitable to sustain profitability on a continuing basis, then we may be unable to continue our operations at planned levels and may be forced to reduce our operations.

We will require substantial additional financing to achieve our goals, and a failure to obtain this necessary capital when needed could force us to delay, limit, reduce or terminate our product development or commercialization efforts.

We expect to expend substantial resources for the foreseeable future to continue the clinical development and manufacturing of our T-cell immunotherapy product candidates, and the advancement and expansion of our preclinical research pipeline. We also expect to continue to expend resources for the development and manufacturing of product candidates and the technology we have licensed or have an exclusive right to license from our partners. These expenditures will include costs associated with research and development, potentially acquiring or licensing new product candidates or technologies, conducting preclinical and clinical studies and potentially obtaining regulatory approvals and manufacturing products, as well as marketing and selling products approved for sale, if any. Under the terms of our license agreements with each of our in-license partners, we are obligated to make payments upon the achievement of certain development, regulatory and commercial milestones. We will also need to make significant expenditures to develop a commercial organization capable of sales, marketing and distribution for any products, if any, that we intend to sell



ourselves in the markets in which we choose to commercialize on our own. In addition, other unanticipated costs may arise. Because the design and outcome of our ongoing, planned and anticipated clinical studies is highly uncertain, we cannot reasonably estimate the actual amounts necessary to successfully complete the development and commercialization of our product candidates.

Our future capital requirements depend on many factors, including:

- the scope, progress, results and costs of researching and developing our product candidates, and conducting preclinical and clinical studies;
- the timing of, and the costs involved in, obtaining regulatory approvals for our product candidates, if clinical studies are successful, including any costs from post-market requirements;
- the cost of commercialization activities for our product candidates, if any of these product candidates is approved for sale, including marketing, sales and distribution costs;
- the cost of contracting for the manufacture of our product candidates for clinical studies in preparation for regulatory approval and in preparation for commercialization;
- our ability to establish and maintain strategic licensing or other arrangements and the financial terms of such agreements;
- the costs to develop, acquire or in-license future product candidates or technologies;
- the costs involved in preparing, filing, prosecuting, maintaining, expanding, defending and enforcing patent claims, including litigation costs and the outcome
 of such litigation;
- · the timing, receipt and amount of sales of, or royalties on, our future products, if any; and
- the emergence of competing technologies or other adverse market developments.

We expect that existing cash, cash equivalents and short-term investments as of March 31, 2022, together with the \$100.0 million received from FDB on April 4, 2022, will be sufficient to fund our planned operations into the fourth quarter of 2023. As of March 31, 2022, we had total cash, cash equivalents and short-term investments of \$301.8 million. However, our operating plan may change as a result of many factors currently unknown to us, and we may need additional funds sooner than planned.

We do not have any committed external source of funds other than reimbursements, milestone and royalty payments that we may receive under the Bayer Agreements and milestone and royalty payments that we may receive under Pierre Fabre Commercialization Agreement. While we expect to continue to opportunistically seek access to additional funds through additional public or private equity offerings or debt financings, through potential collaborations, partnering or other strategic arrangements, or a combination of the foregoing, additional funds may not be available when we need them on terms that are acceptable to us, or at all. If adequate funds are not available to us on a timely basis, we may be required to delay, limit, reduce or terminate preclinical studies, clinical studies or other development activities for one or more of our product candidates or delay, limit, reduce or terminate our establishment of sales, marketing and distribution capabilities or other activities that may be necessary to commercialize our product candidates.

Raising additional capital may cause dilution to our existing stockholders, restrict our operations or require us to relinquish rights to our product candidates on terms that are unfavorable to us.

We may seek required additional capital through a variety of means, including through private and public equity offerings and debt financings. To the extent that we raise additional capital through the sale of equity or convertible debt securities, or if existing holders of warrants exercise their rights to purchase common stock, the ownership interest of existing stockholders will be diluted, and the terms may include liquidation or other preferences that adversely affect the rights of stockholders. To the extent equity valuations, including the trading price of our common stock, are depressed as a result of economic disruptions or other uncertainties, for example due to the COVID-19 pandemic or other factors, the potential magnitude of this dilution will increase. Debt financing, if available, may involve agreements that include covenants limiting or restricting our ability to take certain actions, including incurring additional debt, making capital expenditures, entering into licensing arrangements, or declaring dividends. If we raise additional funds from third parties, we may have to relinquish valuable rights to our technologies or product candidates or grant licenses or other rights on terms that are not favorable to us. If we are unable to raise additional funds through equity or debt financing when needed, we may be required to delay, limit, reduce or terminate our product adversely or develop and market product candidates that we would otherwise prefer to develop and market ourselves or take other actions that are adverse to our business.



Risks Related to the Development of Our Product Candidates

We are generally early in our development efforts and have only a small number of product candidates in clinical development. All of our other product candidates are still in preclinical development. If we or our collaborators are unable to successfully develop and commercialize product candidates or experience significant delays in doing so, our business may be materially harmed.

We are generally early in our development efforts, and only a small number of our product candidates are in clinical development. The majority of our product candidates are currently in preclinical development. We have invested substantial resources in identifying and developing potential product candidates, conducting preclinical and clinical studies, manufacturing activities and preparing for the potential commercial launch of our product candidates. Our ability to generate revenues from the sale of products, if approved, will depend heavily on the successful development and eventual commercialization of our product candidates. The success of our product candidates will depend on many factors, including the following:

- completion of preclinical and clinical studies with positive results, demonstrating the safety, purity, and potency of our product candidates to the satisfaction of the FDA or other regulatory agencies;
- receipt of regulatory approvals from applicable authorities, including required authorizations for clinical trials and marketing authorizations;
- protecting our rights in our intellectual property portfolio, including by obtaining and maintaining patent and trade secret protection and regulatory exclusivity for our product candidates;
- establishing or making arrangements with third-party manufacturers or qualifying our CMOs' manufacturing facilities for clinical and commercial manufacturing purposes;
- developing manufacturing and distribution processes for our novel T-cell product candidates and next-generation CAR T programs;
- contracting with third parties for the manufacture of our product candidates at an acceptable cost;
- launching commercial sales of our products, if approved by applicable regulatory authorities, whether alone or in collaboration with others;
- acceptance of our products, if approved by applicable regulatory authorities, by patients and the medical community;
- obtaining and maintaining coverage and adequate reimbursement by third-party payors, including government payors, for our products, if approved by applicable regulatory authorities;
- effectively competing with other therapies;
- maintaining a continued acceptable benefit/risk profile of the products following approval; and
- maintaining and growing an organization of scientists and functional experts who can develop and commercialize our products and technology.

If we do not achieve one or more of these factors in a timely manner or at all, we could experience significant delays or an inability to successfully develop and commercialize our product candidates, which could materially harm our business.

Our business and operations have been affected by and could be materially and adversely affected in the future by the effects of health epidemics and pandemics, including the evolving and ongoing effects of the COVID-19 pandemic, in particular with respect to any new variants or resurgences of the pandemic. The COVID-19 pandemic continues to impact our business and operations and could materially and adversely affect our business and operations in the future, as well as the businesses and operations of third parties on which we rely.

Our business could be adversely affected by health epidemics and pandemics, including the ongoing COVID-19 pandemic, which has presented a substantial public health and economic challenge around the world and has affected, and continues to affect, our employees, patients, communities and business operations, as well as the U.S. economy and financial markets. As a result of the ongoing COVID-19 pandemic, the work-from-home model we implemented in March 2020 for most of our employees remains in place. We continue to maintain essential in-person manufacturing and laboratory functions in order to advance key research, development and manufacturing priorities. In connection with these measures, we may be subject to claims based upon, arising out of, or related to COVID-19 and our actions and responses thereto, including any determinations that we may make to continue to operate or to re-open our offices and facilities where permitted by applicable law. The effects of current and potential future state executive orders, local shelter-in-place orders, government-imposed quarantines, our work-from-home policies and potential return-to-office strategy and other similar actions may negatively impact productivity, disrupt our business and delay our clinical programs and timelines, the magnitude of which will depend, in part, on the length and severity of the restrictions and other limitations on our ability to conduct our business in the ordinary course.

Further quarantines, shelter-in-place or similar restrictions and other actions taken by foreign, federal, state and local governments, or the perception that such orders, shutdowns or other restrictions on the conduct of business operations could occur or could be reinstated, related to the ongoing COVID-19 pandemic or other infectious diseases, could impact our manufacturing capabilities and third-party manufacturing facilities in the United States and other countries, or the availability or cost of materials, which would disrupt our supply chain. In particular, standard transportation channels have been impacted and we and other manufacturing, testing, product disposition, contract manufacturing organizations (CMOs) and external testing laboratories are subject to enhanced risk assessment and mitigation measures. In addition, there have been and may continue to be interruptions in the supply of leukapheresis collections, which supply raw materials used in our products.

Our clinical trials may also be affected by health epidemics and have been affected by the ongoing and evolving COVID-19 pandemic. Clinical site initiation and patient enrollment have experienced delays as a result of the COVID-19 pandemic, including due to the prioritization of hospital resources toward COVID-19 and away from clinical trials or as a result of changing practice patterns that impact the diseases our trials address. Some patients may not be able to comply with clinical trial protocols if quarantines impede patient movement or interrupt healthcare services or if patients contract COVID-19 or are forced to quarantine. For example, while most clinical trial sites for our studies, including our Phase 3 clinical trial of tab-cel in patients with EBV+ PTLD, remain open to enrollment for patients, some sites have limited the screening and enrollment of new patients due to governmental orders related to COVID-19, or fear of infection of COVID-19, have limited, and may continue to limit, patients' abilities to access clinical sites. COVID-19-related travel restrictions may also interrupt key clinical trial activities, such as clinical trial site data monitoring and efficacy, safety and translational data collection, processing and analyses. At the outset of the COVID-19 pandemic, we observed a temporary slow-down in stem cell and solid organ transplant volumes, which may have decreased the eligible patient population for the tab-cel Phase 3 study. In April 2020, we initiated a temporary pause in the screening and enrollment of patients in our EMBOLD study of ATA188 in patients with PMS. Although we were able to resume the screening and enrollment of patients in our EMBOLD study. Similarly, our ability to recruit and retain principal investigators and site staff who, as healthcare providers, may have heightened exposure to COVID-19, may be adversely impacted.

While we expect the ongoing and evolving COVID-19 pandemic to continue to adversely affect our business operations, the extent of the impact on our clinical development and regulatory efforts and the value of and market for our common stock will depend on future developments that are highly uncertain and cannot be predicted with confidence at this time, such as the ultimate duration of the pandemic, travel restrictions, quarantines, social distancing and business closure requirements in the U.S. and in other countries, and the effectiveness of actions taken globally to contain and treat COVID-19. In addition, to the extent the evolving effects of the ongoing and evolving COVID-19 pandemic adversely affect our business and results of operations, it may also have the effect of heightening many of the other risks and uncertainties described elsewhere in this "Risk Factors" section.

Our future success is dependent on the regulatory approval of our product candidates.

We do not have any products that have gained regulatory approval. Currently, our prioritized clinical-stage product candidates include ATA188 and tab-cel. Our business is substantially dependent on our ability to obtain regulatory approval for, and, if approved, to successfully commercialize our product candidates in a timely manner.

We cannot commercialize product candidates in the U.S. without first obtaining regulatory approval for the product from the FDA; similarly, we cannot commercialize product candidates outside of the U.S. without obtaining regulatory approval from comparable foreign regulatory authorities. Before obtaining regulatory approvals for the commercial sale of any product candidate for a target indication, we must demonstrate with substantial evidence gathered in preclinical and clinical studies that the product candidate is safe and effective for use for that target indication and that the manufacturing facilities, processes and controls are adequate with respect to such product candidate to assure safety, purity and potency.

The time required to obtain approval by the FDA and comparable foreign regulatory authorities is unpredictable but typically takes many years following the commencement of preclinical and clinical studies and depends upon numerous factors, including the substantial discretion of the regulatory authorities. The novel nature of our product candidates may create further challenges in obtaining regulatory approval. For example, the FDA and comparable foreign regulatory authorities have limited experience with regulating the development and commercialization of T-cell immunotherapies, particularly allogeneic T-cell product candidates, and CAR T therapies, including assessing the comparability of different versions of such product candidates. In addition, approval policies, regulatory positions or the type and amount of clinical and other data necessary to gain approval may change during the course of a product candidate's clinical development and throughout regulatory interactions, and may vary among jurisdictions, particularly approval. We have not obtained regulatory approval for any product candidate, and it is possible that none of our existing product candidates or any future product candidates will ever obtain regulatory approval.

Our product candidates could fail to receive regulatory approval from the FDA or a comparable foreign regulatory authority for many reasons, including:

- disagreement with the design or conduct of our clinical studies;
- failure to demonstrate positive benefit/risk profile of the product candidate for its proposed indication;
- failure of clinical sites to conduct the study in accordance with applicable regulatory requirements;
- failure of clinical studies to meet the level of statistical significance required for approval;
- disagreement with our interpretation of data from preclinical studies or clinical studies;
- the insufficiency of data collected from clinical studies of our product candidates to support the submission and filing of a BLA or other submission or to
 obtain regulatory approval;
- inability to reach agreement with the FDA or comparable foreign regulatory authorities on the methodologies for, and assessment of, comparability of different versions of product candidates used in non-pivotal studies, pivotal studies and for intended commercial use;
- failure to obtain approval of our manufacturing processes or facilities of third-party manufacturers with whom we contract for clinical and commercial supplies or our own manufacturing facility; or
- changes or inconsistencies in the requested or required methodologies, statistical analyses, specification criteria or regulatory submission requirements for a
 product candidate, including changes to, or inconsistencies with, applicable industry practice or precedent; or
- changes in the: (i) approval policies or regulations that render our preclinical and clinical data insufficient for approval; or (ii) positions, guidance or feedback communicated by the FDA or comparable foreign regulatory authorities.

The FDA or a comparable foreign regulatory authority may require more information, including additional CMC information, preclinical or clinical data to support approval, which may delay or prevent approval and our commercialization plans, or we may decide to abandon the development program. For example, at a Type B meeting in February 2022, we were not able to align with the FDA on comparability between tab-cel product used for in the pivotal ALLELE study and the intended commercial product. The FDA recommended we conduct a new clinical trial with the commercial product to address the lack of alignment on comparability and to gain additional clinical experience with the intended commercial product. While we continue to explore alternative pathways with the FDA to enable filing of a BLA for tab-cel based on data from the pivotal ALLELE study, we may be unable to establish comparability to the FDA's satisfaction or otherwise reach agreement with the FDA on a pathway to BLA submission with the current clinical dataset. In this case, the conduct of an additional clinical trial or trials in the lead indication may be necessary to support a BLA for tab-cel, which would result in considerable delay to a BLA submission or could lead us not to pursue a BLA submission. Conducting a clinical trial may prove too difficult or too expensive, and the process of designing a clinical trial, enrolling enough patients, and completing treatment and data collection under the protocol could take a significant amount of time, effort, and resources. Even if we complete the clinical trial, the study may not meet its prespecified endpoints, and even if it does, the FDA may still disagree with our determination that the clinical trial is sufficient to support submission and approval of a BLA for tab-cel.

Our development and commercialization activities could be harmed or delayed by governmental or regulatory delays due to limitations on the availability of governmental and regulatory agency personnel to review regulatory filings or engage with us, caused by global health concerns, including the ongoing and evolving COVID-19 pandemic, changes to governmental regulatory requirements, policies, guidelines or priorities, reallocation, or availability of government resources, or for other reasons, which may significantly delay the FDA's, or other regulatory agency, ability to review and process any submissions we have filed or may file or cause other regulatory delays. If global health concerns continue to prevent the FDA or other regulatory authorities from conducting their regular inspections, or impact reviews or other regulatory activities, it could significantly impact the ability of the FDA or other regulatory authorities to timely review and process our regulatory submissions, which could have a material adverse effect on our business. For example, in response to the COVID-19 pandemic, on March 10, 2020, the FDA announced its intention to postpone most inspections of foreign manufacturing facilities and products inspections of domestic manufacturing facilities through April 2020. On March 18, 2020, the FDA announced its intention to temporarily postpone routine surveillance inspections of domestic manufacturing facilities and provided guidance regarding the conduct of clinical trials. On July 10, 2020, the FDA announced that it is working toward the goal of restarting on-site inspections it deems to be "mission critical." In May 2021, the FDA updated its guidance, first published in August 2020, clarifying how it intends to conduct inspections during the COVID-19 pandemic, including how it plans to determine which inspections are "mission critical." Additionally, on April 14, 2021, the FDA issued a guidance document in which the FDA described its plans to conduct voluntary remote interactive evaluations of certain drug manufacturing facilities and clinical research sites. According to the guidance, the FDA intends to request such remote interactive evaluations in situations where an in-person inspection would not be prioritized, deemed mission-critical, or where direct inspection is otherwise limited by travel restrictions, but where the FDA determines that remote evaluation would still be appropriate. The FDA intends to use this risk-based assessment system to identify the categories of regulatory activity that can occur within a given geographic area, ranging from mission critical inspections to resumption of all regulatory activities. The FDA has since adjusted its inspection activities in response to the ongoing COVID-19 pandemic. On December 29, 2021, the FDA implemented temporary changes to its inspectional activities to ensure the safety of its employees and regulated firms. On February 2, 2022, FDA announced that it would resume domestic surveillance inspections across all product areas on February 7, 2022. We cannot predict whether, and when, FDA will

decide to pause or resume inspections due to the COVID-19 pandemic. Regulatory authorities outside the United States may adopt similar restrictions or other policy measures in response to the COVID-19 pandemic. It is unclear how FDA's and other health agencies' policies and guidance will impact any inspections of our facilities, including our clinical trial sites.

If we do obtain approval, regulatory authorities may approve any of our product candidates for fewer or more limited indications than we request (including failing to approve the most commercially promising indications), may grant approval contingent on the performance of costly post-marketing clinical studies, or may approve a product candidate with a label that does not include the labeling claims necessary or desirable for the successful commercialization of that product candidate. In addition, the clinical study requirements of the FDA, EMA and other regulatory agencies and the criteria these regulators use to determine the safety and efficacy of a product candidate are determined according to the type, complexity, novelty and intended use and market of the potential products. The regulatory approval process for novel product candidates, such as our novel T-cell product candidates and next-generation CAR T programs, can be more complex and consequently more expensive and take longer than for other, better known or extensively studied pharmaceutical or other product candidates. Approvals by the EMA and FDA for existing autologous CAR T therapies, such as Novartis' Kymriah® and Gilead's Yescarta®, may not be indicative of what these regulators may require for approval of our therapies. We have multiple clinical trials of our product candidates currently ongoing. If an adverse safety issue, clinical hold or other adverse finding occurs in one or more of our clinical trials, such event could adversely affect our other clinical trials of the same or related product candidates. Moreover, our product candidates may not perform successfully in clinical studies or may be associated with adverse events that distinguish them from those that have previously been approved, such as existing autologous CAR T therapies. For instance, allogeneic product candidates may result in adverse events not experienced with autologous products. Even if a product candidate were to successfully obtain approval from the FDA and comparable foreign regulatory authorities, any approval might contain significant limitations related to use restrictions for specified age groups, warnings, precautions or contraindications, or may be subject to burdensome post-approval study or risk management requirements. If we are unable to obtain regulatory approval for one of our product candidates in one or more jurisdictions, or any approval contains significant limitations, we may not be able to obtain sufficient funding to continue the development of that product or generate revenues attributable to that product candidate. Also, any regulatory approval of our current or future product candidates, once obtained, may be withdrawn.

Our T-cell immunotherapy product candidates and our next-generation CAR T programs represent new therapeutic approaches that could result in heightened regulatory scrutiny, delays in clinical development or delays in or our inability to achieve regulatory approval, commercialization or payor coverage of our product candidates.

Our future success is dependent on the successful development of T-cell immunotherapies and our next-generation CAR T programs in general and our development product candidates in particular. Because these programs, particularly our pipeline of allogeneic T-cell product candidates that are bioengineered from donors, represent a new approach to immunotherapy for the treatment of cancer and other diseases, developing and commercializing our product candidates subject us to a number of challenges, including:

- obtaining regulatory approval from the FDA and other regulatory authorities, which have limited experience with regulating the development and commercialization of T-cell immunotherapies, particularly allogeneic T-cell product candidates;
- developing and deploying consistent and reliable processes for procuring blood from consenting third-party donors, isolating T cells from the blood of such donors, activating the isolated T cells against a specific antigen, characterizing and storing the resulting activated T cells for future therapeutic use, selecting and delivering a sufficient supply and breadth of appropriate partially HLA-matched cell line from among the available T-cell lines, and finally infusing these activated T cells into patients;
- utilizing these product candidates in combination with other therapies (e.g., immunomodulatory approaches such as checkpoint inhibitors), which may
 increase the risk of adverse side effects;
- educating medical personnel regarding the potential side effect profile of each of our product candidates, particularly those that may be unique to our allogeneic T-cell product candidates and to our next-generation CAR T programs;
- understanding and addressing variability in the quality of a donor's T cells, which could ultimately affect our ability to manufacture product in a reliable and consistent manner;
- developing processes for the safe administration of these products, including long-term follow-up and registries, for all patients who receive these product candidates;
- establishing or making arrangements with third-party manufacturers to manufacture, or manufacturing on our own, product candidates to our specifications and in a timely manner to support our clinical studies and, if approved, commercialization;
- sourcing clinical and, if approved by applicable regulatory authorities, commercial supplies for the materials used to manufacture and process these product candidates that are free from viruses and other pathogens that may increase the risk of adverse side effects;

- developing a manufacturing process and distribution network that can provide a stable supply with a cost of goods that allows for an attractive return on investment;
- establishing sales and marketing capabilities ahead of and after obtaining any regulatory approval to gain market acceptance, and obtaining adequate coverage, reimbursement and pricing by third-party payors and government authorities; and
- developing therapies for types of diseases beyond those initially addressed by our current product candidates.

We cannot be sure that the manufacturing processes used in connection with our T-cell immunotherapy product candidates will yield a sufficient supply of satisfactory products that are safe, pure, potent, and comparable to those T cells historically produced by our partners, be scalable or profitable.

Moreover, actual or perceived safety issues, including adoption of new therapeutics or novel approaches to treatment, may adversely influence the willingness of subjects to participate in clinical studies, or, if one of our product candidates is approved by applicable regulatory authorities, of physicians to subscribe to the novel treatment mechanics or of patients to provide consent to receive a novel treatment despite its regulatory approval. The FDA or other applicable regulatory authorities may require specific post-market studies or additional information that communicates the benefits or risks of our products. New data may reveal new risks of our product candidates at any time prior to or after regulatory approval.

Physicians, hospitals and third-party payors often are slow to utilize new products, technologies and treatment practices that require additional upfront costs and training. Physicians may not be willing to undergo training on this novel therapy, may decide the therapy is too complex to adopt without appropriate training or not cost-efficient, and may choose not to administer the therapy. Based on these and other factors, hospitals and payors may decide that the benefits of this new therapy do not or will not outweigh its costs.

The results of preclinical studies or earlier clinical studies are not necessarily predictive of future results. Our existing product candidates in clinical studies, and any other product candidate we advance into clinical studies, may not have favorable results in later clinical studies or receive regulatory approval.

Success in preclinical studies and early clinical studies does not ensure that later clinical studies will generate adequate data to demonstrate the efficacy and safety of an investigational drug. Likewise, a number of companies in the pharmaceutical and biotechnology industries, including those with greater resources and experience than us, have suffered significant setbacks in clinical studies, even after seeing promising results in earlier preclinical studies or clinical studies. Despite the results reported in earlier preclinical studies for our product candidates, there can be significant variability in safety and efficacy results between different clinical trials of the same product candidate due to numerous factors, we do not know whether the clinical studies we may conduct, or clinical studies in progress, will demonstrate adequate efficacy and safety to result in regulatory approval to market tab-cel, ATA188, any product candidates resulting from our next-generation CAR T programs or any of our other product candidates in any particular jurisdiction.

Tab-cel has been predominantly evaluated in single-center studies under investigator-sponsored investigational new drug (INDs) held by MSK and in our EAP, utilizing different response criteria and endpoints from those we may utilize in later clinical studies. The findings may not be reproducible in late phase studies we conduct. For instance, the current protocol for our ALLELE study is designed to rule out a 20% ORR as the null hypothesis. This means that if the lower bound of the 95% confidence interval on ORR among patients receiving at least one dose of tab-cel exceeds 20% at the end of the study, then the study would be expected to meet the primary endpoint for the treatment of PTLD. For example, assuming enrollment of 33 patients in a cohort of ALLELE, an observed ORR above approximately 37% would be expected to meet the primary endpoint for that cohort. In addition, our amended ALLELE study protocol includes an interim analysis as well as a final study analysis. Depending on discussions with regulators, we may, for example, file a marketing application on the basis of interim data from a subset of the required patients or file a marketing application based on interim data would impact the required ORR, may impact the approved indication, and may also result in post-marketing requirements that must be fulfilled. Similarly, if conditional marketing authorization is granted from the European Commission, we may be subject to ongoing obligations, including the need to provide additional clinical data at a later stage to confirm a positive benefit/risk balance.

For regulatory approvals of tab-cel, we plan on using independent radiologist and/or oncologist assessment of responses which may not correlate with the investigatorreported assessments. In addition, the Phase 2 clinical studies with tab-cel enrolled a heterogeneous group of patients with a variety of EBV-driven malignancies, including EBV+ PTLD after HCT and EBV+ PTLD after SOT. These Phase 2 studies were not prospectively designed to evaluate the efficacy of tab-cel in the treatment of a single disease state for which we may later seek approval.

Moreover, final study results may not be consistent with interim study results. Efficacy data from prospectively designed studies may differ significantly from those obtained from retrospective subgroup analyses. In addition, clinical data obtained from a clinical study with an allogeneic product candidate may not yield the same or better results as compared to an autologous product candidate. If later-stage clinical studies do not produce favorable results, our ability to achieve regulatory approval for any of our product candidates may be adversely impacted. Even if we believe that we have adequate data to support an application for regulatory approval to market any of our product candidates, the FDA or other regulatory authorities may not agree and may require that we conduct additional clinical studies.

Interim "top line" and preliminary data from clinical studies that we or our partners may announce or share with regulatory authorities from time to time may change as more patient data become available and are subject to audit and verification procedures that could result in material changes in the final data.

From time to time, we or our partners may announce or share with regulatory authorities interim "top line" or preliminary data from clinical studies. Interim data from completed clinical studies are subject to the risk that one or more of the clinical outcomes may materially change as patient enrollment continues and more patient data become available. Preliminary or "top line" data also remain subject to audit and verification procedures that may result in the final data being materially different from the preliminary data previously announced. As a result, interim and preliminary data should be viewed with caution until the final data are available. Adverse differences between preliminary or interim data and final data could impact the regulatory approval of, and significantly harm the prospects of any product candidate that is impacted by the applicable data.

Clinical drug development involves a lengthy and expensive process with an uncertain outcome.

Clinical testing is expensive and can take many years to complete, and its outcome is inherently uncertain. Failure can occur at any time during the clinical study process. Product candidates in later stages of clinical studies may fail to show the desired safety and efficacy traits despite having progressed through preclinical and clinical studies.

We may experience delays in our ongoing or future clinical studies and we do not know whether clinical studies will begin or enroll subjects on time, will need to be redesigned or will be completed on schedule, if at all. There can be no assurance that the FDA or comparable foreign regulatory authorities will not put clinical studies of any of our product candidates on clinical hold in the future. Clinical studies may be delayed, suspended or prematurely terminated for a variety of reasons, such as:

- delays in enrollment due to travel, shelter-in-place or quarantine policies, or other factors, related to the ongoing COVID-19 pandemic or other epidemics or pandemics;
- delay or failure in reaching agreement with the FDA or a comparable foreign regulatory authority on a study design that we are able to execute;
- delay or failure in obtaining authorization to commence a study or inability to comply with conditions imposed by a regulatory authority regarding the scope or design of a study;
- delay or failure in reaching agreement on acceptable terms with prospective contract research organizations (CROs) and clinical study sites, the terms of which
 can be subject to extensive negotiation and may vary significantly among different CROs and study sites;
- delay or failure in obtaining institutional review board (IRB) approval or the approval of other reviewing entities, including comparable foreign regulatory authorities, to conduct a clinical study at each site;
- withdrawal of clinical study sites from our clinical studies or the ineligibility of a site to participate in our clinical studies;
- delay or failure in recruiting and enrolling eligible subjects to participate in a study;
- delay or failure in subjects completing a study or returning for post-treatment follow-up;
- clinical sites and investigators deviating from study protocol, failing to conduct the study in accordance with regulatory requirements, or dropping out of a study;
- an FDA or other regulatory authority clinical site inspection reveals serious violations of regulations applicable to clinical investigations, which may result in requests for additional data analyses and/or rejection of data deemed unreliable;
- inability to identify and maintain a sufficient number of study sites, including because potential study sites may already be engaged in competing clinical study programs enrolling the same population;
- failure of our third-party clinical study managers to satisfy their contractual duties, meet expected deadlines or return trustworthy data;

- delay or failure in adding new study sites;
- interim results or data that are ambiguous or negative or are inconsistent with earlier results or data;
- feedback from the FDA, the IRB, data safety monitoring boards or comparable foreign authorities, or results from earlier stage or concurrent preclinical and clinical studies, that might require modification to the protocol for a study;
- a decision by the FDA, the IRB, comparable foreign authorities, or us, or a recommendation by a data safety monitoring board or comparable foreign authority, to suspend or terminate clinical studies at any time for non-compliance with regulatory requirements, safety issues, including a finding that our product candidates have undesirable side effects or other unexpected characteristics, or a finding that the participants are being exposed to unacceptable health risk, or for any other reason;
- unacceptable benefit/risk profile, unforeseen safety issues or adverse side effects;
- failure to demonstrate a benefit from using a product candidate;
- difficulties in manufacturing or obtaining from third parties sufficient quantities and breadth of appropriate partially HLA matched cell lines from among the available T-cell lines to start or to use in clinical studies;
- lack of adequate funding to continue a study, including the incurrence of unforeseen costs due to enrollment delays, requirements to conduct additional studies
 or increased expenses associated with the services of our CROs and other third parties; or
- changes in governmental regulations or administrative actions or lack of adequate funding to continue a clinical study.
- Patient enrollment, a significant factor in the timing of clinical studies, is affected by many factors including:
- the size and nature of the patient population;
- the possibility that the rare diseases that many of our product candidates address are under-diagnosed;
- changing medical practice patterns or guidelines related to the diseases or conditions we are investigating;
- the severity of the disease under investigation, our ability to open clinical study sites;
- the proximity of subjects to clinical sites;
- the patient referral practices of physicians;
- the design and eligibility criteria of the clinical study;
- ability to obtain and maintain patient consents;
- risk that enrolled subjects will drop out or die before completion;
- competition for patients from other clinical studies;
- our or our partner's ability to manufacture the requisite materials for a study;
- risk that we do not have appropriately matched HLA cell lines;
- clinicians' and patients' perceptions as to the potential advantages and risks of the drug being studied in relation to other available therapies, including any new
 drugs that may be approved for the diseases or conditions we are investigating; and
- disruptions caused by man-made or natural disasters or public health pandemics or epidemics, including, for example, the ongoing COVID-19 pandemic.

As an example, we activated additional clinical sites for the ALLELE study of tab-cel over the course of 2018 and increased HLA coverage during this period. As a result, enrollment in our studies was limited in the early part of 2018 and increased through the course of the year as we increased clinical sites and HLA coverage. However, in May 2019, we announced that enrollment in our Phase 3 studies of tab-cel for patients with EBV+ PTLD was proceeding slower than anticipated. Many of our product candidates are designed to treat rare diseases, and as a result, the pool of potential patients with respect to a given disease is small. We may not be able to initiate or continue to support clinical studies of tab-cel, ATA188 or any other product candidates if we are unable to locate and enroll a sufficient number of eligible participants in these studies as required by the FDA or other regulatory authorities. We have experienced some transient delays in clinical trial site initiation and patient enrollment in certain of our clinical trials, including our Phase 3 clinical trial of tab-cel, as a result of the evolving impact of the ongoing COVID-19 pandemic, and if the COVID-19 pandemic continues and persists for an extended period of time, we could experience significant disruptions to our clinical development timelines. Even if we are able to enroll a sufficient number of patients in our clinical studies, if the pace of enrollment is

slower than we expect, the development costs for our product candidates may increase and the completion of our studies may be be available or our studies could become too expensive to complete.

We rely on CROs, other vendors and clinical study sites to ensure the proper and timely conduct of our clinical studies, and while we have agreements governing their committed activities, we have limited influence over their actual performance. Furthermore, these third parties may also have relationships with other entities, some of which may be our competitors. Reliance on CROs entails risks to which we would not be subject if we conducted our clinical studies ourselves, including reliance on the CRO for clinical site initiation and monitoring, the possibility that the CRO does not maintain the financial resources to meet its obligations under our agreements, the possibility of breach of these agreements by the CRO because of factors beyond our control, including a failure to properly perform their obligations under these agreements, and the possibility of termination or nonrenewal of the agreements by the CROs, based on their own business priorities, at a time that is costly or damaging to us. Our reliance on these third parties for research and development activities will reduce our control over these activities but will not relieve us of our regulatory responsibilities. For example, we will remain responsible for ensuring that each of our clinical trials is conducted in accordance with the general investigational plan, study protocols for the trial, statistical analysis plan and other study-specific documents (for example, monitoring and blinding plans). Moreover, the FDA requires us to comply with standards, commonly referred to as Good Clinical Practice (GCP). International Council for Harmonization of Technical Requirements for Pharmaceuticals for Human Use, or ICH, guidelines, and regulations regarding the informed consent process, safety reporting requirements, data collection guidelines, and other regulations for conducting, recording and reporting the results of clinical trials to assure that data and reported results are credible and accurate and that the rights, integrity and confidentiality of trial participants are protected. The EMA also requires us to comply with similar standards. Regulatory authorities enforce these GCP requirements through periodic inspections of trial sponsors, principal investigators and trial sites. If we or any of our CROs fail to comply with applicable GCP requirements, the clinical data generated in our clinical trials may be deemed unreliable and the FDA, EMA or comparable foreign regulatory authorities may require us to perform additional clinical trials before approving our marketing applications. We cannot assure you that upon inspection by a given regulatory authority, such regulatory authority will determine that any of our clinical trials comply with GCP and other applicable regulations. In addition, our clinical trials must be conducted with product produced under applicable current Good Manufacturing Practices (cGMP) and current Good Tissue Practices (cGTP) regulations. Our failure to comply with these regulations may require us to conduct new clinical trials, which would delay the marketing approval process. We also are required to register certain ongoing clinical trials and post the results of certain completed clinical trials on a government-sponsored database, ClinicalTrials.gov, within certain timeframes. Failure to do so can result in fines, adverse publicity and civil and criminal sanctions.

If we experience delays or quality issues in the conduct, completion or termination of any clinical study of our product candidates, the approval and commercial prospects of such product candidate will be harmed, and our ability to generate product revenues from such product candidate will be delayed. In addition, any delays in completing our clinical studies will increase our costs, slow down our product candidate development and approval process and jeopardize our ability to commence product sales and generate revenues. Any delays in completing our clinical studies for our product candidates may also decrease the period of commercial exclusivity. In addition, many of the factors that could cause a delay in the commencement or completion of clinical studies may also ultimately lead to the denial of regulatory approval of our product candidates.

Our product candidates, the methods used to deliver them or their dosage levels may cause undesirable side effects or have other properties that could delay or prevent their regulatory approval, limit the commercial profile of an approved label or result in significant negative consequences following any regulatory approval.

Undesirable side effects caused by our product candidates, their delivery methods or dosage levels could cause us or regulatory authorities to interrupt, delay or halt clinical studies and could result in a more restrictive label or the delay or denial of regulatory approval by the FDA or other comparable foreign regulatory authority. As a result of safety or toxicity issues that we or our partners may experience in our clinical studies, we or our partners may not receive approval to market any product candidates, which could prevent us from ever generating product or royalty revenues or achieving profitability. Results of our studies could reveal an unacceptably high severity and incidence of side effects, or risks that outweigh the benefits of our product candidates. In such an event, our studies could be suspended or terminated, and the FDA or comparable foreign regulatory authorities could order us to cease further development of or deny approval of our product candidates for any or all targeted indications. The drug-related side effects could affect patient recruitment or the ability of enrolled subjects to complete the study or result in potential product liability claims.

Additionally, if any of our product candidates receives regulatory approval, and we or others later identify undesirable side effects caused by such product, a number of potentially significant negative consequences could result, including that:

- we may be forced to suspend marketing of that product;
- regulatory authorities, IRBs, or other clinical trial oversight bodies may place a hold on any ongoing clinical trials;
- regulatory authorities may withdraw or change their approvals of that product;

- regulatory authorities may require additional warnings on the label or limit access of that product to selective specialized centers with additional safety
 reporting and with requirements that patients be geographically close to these centers for all or part of their treatment;
- we may be required to conduct post-marketing studies;
- we may be required to change the way the product is administered;
- we could be sued and held liable for harm caused to subjects or patients;
- our products may be seized, or we may be required to recall our products;
- our products may become less competitive in the marketplace; and
- our reputation may suffer.

Any of these events could diminish the usage or otherwise limit the commercial success of our product candidates and prevent us from achieving or maintaining market acceptance of the affected product candidate, if approved by applicable regulatory authorities.

The market opportunities for our product candidates may be limited to those patients who are ineligible for or have failed prior treatments and may be small.

The FDA often approves new cancer therapies initially only for use in patients with relapsed or refractory metastatic disease. We expect to initially seek approval of tabcel and our other oncology product candidates in this setting. Subsequently, for those products that prove to be sufficiently beneficial, if any, we may seek approval for earlier lines of treatment and potentially as a first line therapy, but there is no guarantee that our product candidates, even if approved, would be approved for earlier lines of therapy, and, prior to any such approvals, we will have to conduct additional clinical trials.

Our projections of both the number of people who have the diseases we are targeting, as well as the subset of people with these diseases in a position to receive second or later lines of therapy, and who have the potential to benefit from treatment with our product candidates, are based on our current beliefs and estimates. These estimates have been derived from a variety of sources, including scientific literature, surveys of clinics, patient foundations, or our own market research, and may prove to be incorrect, including if the COVID-19 pandemic and associated responses impact our ability to engage with key stakeholders within the transplant center in person. Further, new studies, product approvals or market research may change the estimated incidence or prevalence of these diseases, and the number of patients may turn out to be lower than expected. Additionally, the potentially addressable patient population for our product candidates may be limited or may not be amenable to treatment with our product candidates. For instance, we expect our lead product candidate, tab-cel, to initially target a patient population that suffers from aggressive EBV+PTLD who have failed rituximab or rituximab plus chemotherapy. At the outset of the COVID-19 pandemic, we initially observed a temporary slow-down in stem cell and solid organ transplant volumes. These reductions were transient, but if a reduction in such volumes resumes, it could result in lower PTLD incidence and thus reduce the demand for tab-cel. Even if we obtain significant market share for our product candidates, because the potential target populations are small, we may never achieve profitability without obtaining regulatory approval for additional indications.

We may not be able to obtain or maintain orphan drug exclusivity for our product candidates.

Regulatory authorities in some jurisdictions, including the U.S., EU and the United Kingdom (U.K.), may designate drugs for relatively small patient populations as orphan drugs. Under the Orphan Drug Act, the FDA may designate a product as an orphan drug if it is a drug intended to treat a rare disease or condition, which is generally defined as a patient population of fewer than 200,000 individuals annually in the U.S. Both the FDA and the EMA have granted us orphan designation for tab-cel for EBV+ PTLD after HCT or SOT.

Generally, if a product with an orphan drug designation subsequently receives the first regulatory approval for the indication for which it has such designation, the product is entitled to a period of marketing exclusivity, which precludes the EMA or the FDA from approving another marketing application for the same biologic for the same indication for that time period. The applicable period is seven years in the U.S. and ten years in Europe. The European exclusivity period can be reduced to six years if a drug no longer meets the criteria for orphan drug designation or if the drug is sufficiently profitable so that market exclusivity is no longer justified. Orphan drug exclusivity may be lost if the FDA or EMA determines that the request for designation was materially defective or if the manufacturer is unable to assure sufficient quantity of the drug to meet the needs of patients with the rare disease or condition. In the U.S., the FDA may still approve a later marketing application blocked by an ongoing period of orphan drug exclusivity in limited circumstances such as a showing of clinical superiority to the product with orphan drug exclusivity or if the FDA finds that the holder of the orphan drug exclusivity has not shown that it can assure the availability of sufficient quantities of the orphan drug to meet the



needs of patients with the disease or condition for which the biological product was designated. As a result, even if one of our product candidates receives orphan exclusivity, the FDA can still approve or license other drugs or biological products that have a different active ingredient for use in treating the same indication or disease. Further, the FDA can waive orphan exclusivity if we are unable to manufacture sufficient supply of our product.

Even if we obtain orphan drug exclusivity for a product, that exclusivity may not be maintained or effectively protect the product from competition because different drugs can be approved for the same condition. Even after an orphan drug is approved, the FDA can subsequently approve a new drug for the same condition if the FDA concludes that the later drug is clinically superior in that it is shown to be safer, more effective or makes a major contribution to patient care.

BTD by the FDA and PRIME designation by the EMA may not lead to a faster development or regulatory review or approval process and it does not increase the likelihood that our product candidates will receive marketing approval.

Although we have obtained BTD and PRIME designation for tab-cel in the U.S. and EU, respectively, for treatment of patients with EBV+ PTLD who have failed rituximab, these designations may not lead to faster development or regulatory review and do not increase our likelihood of success. A breakthrough therapy is defined as a drug or biologic that is intended, alone or in combination with one or more other drugs or biologics, to treat a serious or life-threatening disease or condition and preliminary clinical evidence indicates that the drug, or biologic in our case, may demonstrate substantial improvement over existing therapies on one or more clinically significant endpoints, such as substantial treatment effects observed early in clinical development. For product candidates that have been designated as breakthrough therapies, interaction and communication between the FDA and the sponsor of the study can help to identify the most efficient path for clinical development while minimizing the number of patients placed in ineffective control regimens. Biologics designated as breakthrough therapies by the FDA may also be eligible for other expedited review programs, such as priority review. Based on our BTD, we may pursue a rolling submission strategy for our BLA for tab-cel for EBV+ PTLD in the U.S. While rolling review process may provide the opportunity for ongoing communications with and feedback from the FDA, it may not result in a faster timeline to marketing approval or result in approval at all. The FDA may raise issues and pose questions to us that may delay the initiation and completion of our BLA submission, acceptance of the complete BLA for filing, and approval of the BLA. We may not be able to provide a satisfactory or a timely response to FDA questions or we may not be able to timely gather the required data to prepare our BLA submissions as we plan. If we are unable to address all questions or concerns the FDA may raise or if we do not have timely access to the data required for the preparation of t

PRIME designation supports the development and accelerated review by the EMA of new therapies to treat patients with unmet medical need. Despite this designation and the associated opportunity for accelerated assessment, the EMA may decide that additional time is needed for the MAA review and convert the MAA to a standard review timeline. For example, the EMA recently informed us that it has transitioned our MAA for tab-cel to a standard assessment as they informed us an additional month was needed to adequately review our responses to the EMA's questions.

Designation as a breakthrough therapy is within the discretion of the FDA, and access to PRIME is at the discretion of the EMA. Receipt of a BTD or PRIME designation for a product candidate may not result in a faster development process, review or approval compared to drugs considered for approval under non-expedited FDA or EMA review procedures and does not assure ultimate approval by either the FDA or EMA. In addition, the FDA or EMA, respectively, may later decide that the product no longer meets the conditions for qualification and rescind the BTD or PRIME designation or decide that the time period for FDA or EMA, respectively, review or approval will not be shortened.

A Fast Track designation by the FDA, even if granted for other current or future product candidates, may not lead to a faster development or regulatory review, licensure process and does not increase the likelihood that our product candidates will receive marketing licensure.

We may seek Fast Track designation for one or more of our future product candidates. In December 2021, ATA188 received Fast Track designation for treatment of patients with PPMS and SPMS. If a drug or biological product is intended for the treatment of a serious or life-threatening disease or condition and it demonstrates the potential to address unmet medical needs for such a disease or condition, the drug sponsor may apply for FDA Fast Track designation for a particular indication. We may seek Fast Track designation for our product candidates, but there is no assurance that the FDA will grant this designation to any of our proposed product candidates. Marketing applications submitted by sponsors of products in Fast Track development may qualify for priority review under the policies and procedures offered by the FDA, but the Fast Track designation does not assure any such qualification or ultimate marketing licensure by the FDA. The FDA has broad discretion whether or not to grant Fast Track designation, so even if we believe a particular product candidate is eligible for this designation, there can be no assurance that the FDA would decide to grant it.

Even if we do receive Fast Track designation, we may not experience a faster development process, review or licensure compared to conventional FDA procedures or pathways and receiving a Fast Track designation does not provide assurance of ultimate FDA licensure. In addition, the FDA may withdraw Fast Track designation if it believes that the designation is no longer supported by data from our clinical development program. In addition, the FDA may withdraw any Fast Track designation at any time.

Failure to obtain regulatory approval in international jurisdictions would prevent our product candidates from being marketed abroad.

In addition to regulations in the U.S., to market and sell our products in the EU, the U.K., many Asian countries and other jurisdictions, we, or our current or future commercialization partners, must obtain separate regulatory approvals and comply with numerous and varying regulatory requirements, both from a clinical and manufacturing perspective. The approval procedure varies among countries and can involve additional testing. The time required to obtain approval may differ substantially from that required to obtain FDA approval. The regulatory approval process outside the U.S. generally includes all of the risks associated with obtaining FDA approval and may include additional risks. Clinical studies accepted in one country may not be accepted by regulatory authorities in other countries. In addition, many countries outside the U.S. require that a product be approved for reimbursement before it can be approved for sale in that country. A product candidate that has been approved for sale in a particular country may not receive reimbursement approval in that country. We may not be able to obtain approvals from regulatory authorities or payor authorities outside the U.S. on a timely basis, if at all. Approval by and regulatory agency or payor does not ensure approval by any other regulatory approvals and may not receive necessary approvals to commercialize our products in any market. If we are unable to obtain approval of any of our product candidates by regulatory approvals of that product candidate may be significantly diminished.

Even if our product candidates receive regulatory approval, they may still face future development and regulatory difficulties.

Even if we, or our partners obtain regulatory approval for a product candidate, it would be subject to ongoing requirements by the FDA and comparable foreign regulatory authorities governing the manufacture, quality control, further development, labeling, packaging, storage, distribution, adverse event reporting, safety surveillance, import, export, advertising, promotion, recordkeeping and reporting of safety and other post-marketing information. These requirements include submissions of safety and other post-marketing information and reports, registration, as well as continued compliance by us and/or our CMOs and CROs for any post-approval clinical studies that we conduct. They also include any post-approval requirements or commitments imposed by FDA as a condition of approval, or any risk evaluation or mitigation strategies (REMS), if applicable. The safety profile of any product will continue to be closely monitored by the FDA and comparable foreign regulatory authorities after approval. If the FDA or comparable foreign regulatory authorities become aware of new safety information after approval of any of our product candidates, they may require labeling changes or establishment of a risk evaluation and mitigation strategy, impose significant restrictions on a product's indicated uses or marketing or impose ongoing requirements for potentially costly post-approval studies or post-market surveillance.

In addition, manufacturers of drug products and their facilities are subject to initial and continual review and periodic inspections by the FDA and other regulatory authorities for compliance with current good manufacturing practices (cGMP), current Good Clinical Practices (GCP), current good tissue practices (cGTP) and other regulations. If we or a regulatory agency discover previously unknown problems with a product, such as adverse events of unanticipated severity or frequency, or problems with the facility where the product is manufactured, a regulatory agency may impose restrictions on that product, the manufacturing facility or us, including requiring recall or withdrawal of the product from the market or suspension of manufacturing. If we, our product candidates, or the manufacturing facilities for our product candidates fail to comply with applicable regulatory requirements, a regulatory agency may:

- issue warning letters or untitled letters;
- mandate modifications to promotional materials or require us to provide corrective information to healthcare practitioners;
- require us or our partners to enter into a consent decree, which can include imposition of various fines, reimbursements for inspection costs, required due dates for specific actions and penalties for noncompliance;
- seek an injunction or impose civil or criminal penalties or monetary fines;
- suspend, withdraw or modify regulatory approval;
- suspend or modify any ongoing clinical studies;
- refuse to approve pending applications or supplements to applications filed by us;

- suspend or impose restrictions on operations, including costly new manufacturing requirements; or
- seize or detain products, refuse to permit the import or export of products, or require us to initiate a product recall.

The occurrence of any event or penalty described above may also generate negative publicity or inhibit our ability to successfully commercialize our products.

Advertising and promotion of any product candidate that obtains approval in the U.S. will be heavily scrutinized by the FDA, the Department of Justice (DOJ), the Office of Inspector General of the Department of Health and Human Services (HHS), state attorneys general, members of the U.S. Congress and the public. Additionally, advertising and promotion of any product candidate that obtains approval outside of the U.S. will be heavily scrutinized by comparable foreign entities and stakeholders. For example, a company may not promote "off-label" uses for its drug products. An off-label use is the use of a product for an indication that is not described in the product's FDA-approved label in the U.S. or for uses in other jurisdictions that differ from those approved by the applicable regulatory agencies. Physicians, on the other hand, may prescribe products for off-label uses. Although the FDA and other regulatory agencies do not regulate a physician's choice of drug treatment made in the physician's independent medical judgment, they do restrict promotional communications from companies or their sales force with respect to off-label uses of product's FDA approved labeling. Violations, including actual or alleged promotion of our products for unapproved or off-label uses, are subject to enforcement letters, inquiries and investigations, and civil and criminal sanctions by the FDA or comparable foreign bodies. Any actual or alleged failure to comply with labeling and promotion requirements may result in fines, warning letters, mandates to corrective information to healthcare practitioners, injunctions, or civil or criminal penalties.

Regulations, guidelines and recommendations published by various government agencies and organizations may affect the use of our product candidates.

Changes to regulations, recommendations or other guidelines advocating alternative therapies for the indications we treat could result in decreased use of our products. For example, although treatment with EBV-specific T cells is recognized as a recommended treatment for persistent or progressive EBV+ PTLD as set forth in the National Comprehensive Cancer Network Guidelines, future guidelines from governmental agencies, professional societies, practice management groups, private health/science foundations and other organizations could lead to decreased ability of developing our product candidates, or decreased use of our products once approved by applicable regulatory authorities.

We may not successfully identify, acquire, develop or commercialize new potential product candidates.

Part of our business strategy is to expand our product candidate pipeline by identifying and validating new product candidates, which we may develop ourselves, inlicense or otherwise acquire from others. In addition, in the event that our existing product candidates do not receive regulatory approval or are not successfully commercialized, then the success of our business will depend on our ability to expand our product pipeline through internal development, in-licensing or other acquisitions. We may be unable to identify relevant product candidates. If we do identify such product candidates, we may be unable to reach acceptable terms with any third party from which we desire to inlicense or acquire them. Any product candidates we identify, acquire, in-license, or develop may not be safe or effective for their targeted diseases, and may not receive marketing authorization in a timely manner, or at all.

Risks Related to Manufacturing

We are subject to a multitude of manufacturing risks, any of which could substantially increase our costs and limit supply of our product candidates.

Concurrently with the in-license of our existing product candidates, we acquired manufacturing process know-how and, in some cases, inventory of process intermediates and clinical materials from our partners. Transferring manufacturing processes, testing and associated know-how from our partners, and to our CMOs, is complex and involves review and incorporation of both documented and undocumented processes that may have evolved over time. In addition, transferring production to different facilities may require utilization of new or different processes to meet the specific requirements of a given facility. Each stage is retroactively and concurrently verified to be compliant with appropriate regulations. As a result, there is a risk that all relevant know-how was not adequately transferred to us from our partners or that previous execution was not compliant with applicable regulations.

In addition, we need to conduct significant development and scale-up work to transfer these processes and manufacture each of our product candidates for various studies, clinical studies and commercial launch readiness. To the extent we elect to transfer manufacturing within our network of third-party CMOs, we are required to demonstrate that the product manufactured in the new or "receiving" facility is comparable to the product manufactured in the original or "sending" facility. The inability to demonstrate to

each of the applicable regulatory authorities that comparable drug product was manufactured could delay the development of our product candidates

The processes by which our product candidates are manufactured were initially developed by our partners for clinical purposes. We intend to evolve the existing processes with our partners to support advanced clinical studies and commercialization requirements. Developing commercially viable manufacturing processes is a difficult and uncertain task, and there are risks associated with scaling to the level required for advanced clinical studies or commercialization, including cost overruns, potential problems with process scale-up, process reproducibility, comparability issues, stability issues, consistency and timely availability of reagents or raw materials. The manufacturing facilities in which our product candidates will be made could be adversely affected by the ongoing COVID-19 pandemic, earthquakes and other natural or man-made disasters, equipment failures, labor shortages, power failures, and numerous other factors. In addition, there have been, and there may continue to be, transient interruptions in the supply of raw materials and consumables used in our product candidates. If we are unable to obtain such raw materials or other necessary raw materials in a timely manner, our business operations and manufacturing capabilities could be adversely affected.

The process of manufacturing cellular therapies is susceptible to product loss due to contamination, equipment failure or improper installation or operation of equipment, or vendor or operator error. Even minor deviations from normal manufacturing and distribution processes for any of our product candidates could result in reduced production yields, impact to key product quality attributes, and other supply disruptions. Product defects can also occur unexpectedly. If microbial, viral or other contaminations are discovered in our product candidates or in the manufacturing facilities in which our product candidates are made, these manufacturing facilities may need to be closed for an extended period of time to allow us to investigate and remedy the contamination. Because our T-cell immunotherapy product candidates are manufactured from cells collected from the blood of third-party donors, the process of manufacturing is susceptible to the availability of the third-party donor material. The process of developing products that can be commercialized may be particularly challenging, even if they otherwise prove to be safe and effective. The manufacturing these product candidates will be susceptible to additional risks, given the need to maintain aseptic conditions throughout the manufacturing process. Contamination with viruses or other pathogens in either the donor material or materials utilized in the manufacturing process or ingress of microbiological material at any point in the process may result in contaminated or unusable product. This type of contaminations could result in delays in the manufacture of products which could result in delays in the development of our product candidates. These contaminations could also increase the risk of adverse side effects. Furthermore, our allogeneic products ultimately consist of many individual cell intermediate or cell product lots, each with a different HLA profile. As a result, the selection and distribution of the appropriate cell product lot for therapeutic use in a p

Any adverse developments affecting our, or our CMOs' manufacturing operations for our product candidates may result in lot failures, inventory shortages, shipment delays, product withdrawals or recalls or other interruptions in the supply of our drug product which could delay the development of our product candidates. We may also have to write off inventory, incur other charges and expenses for supply of drug product that fails to meet specifications, undertake costly remediation efforts, or seek more costly manufacturing alternatives. Inability to meet the demand for our product candidates could damage our reputation and the reputation of our products among physicians, healthcare payors, patients or the medical community that supports our product development efforts, including hospitals and outpatient clinics.

Delays in receiving regulatory approvals for product candidates produced at our CMOs' facilities could delay our development plans and thereby limit our ability to generate revenues.

The research and development and process and analytical development labs within ARC and our facility in Denver, Colorado, currently support our preclinical and mid/late development activities. Product-specific qualification to support clinical development is complete and commercial production qualification activities are ongoing at our CMOs' facilities. If the appropriate regulatory approvals for manufacturing product candidates at our CMOs' facilities are delayed, we may not be able to manufacture sufficient quantities of our product candidates, which would limit our development activities and our opportunities for growth.

In addition to the similar manufacturing risks described in "Risks Related to Our Dependence on Third Parties," our facilities, and our CMOs' facilities, will be subject to ongoing, periodic inspection by the FDA, EMA or other comparable regulatory agencies to ensure compliance with cGMP and GTP. Our, or our partner's, failure to follow and document our adherence to these regulations or other regulatory requirements may lead to significant delays in the availability of products for clinical or, in the future, commercial

use, may result in the termination of or a hold on a clinical study, or may delay or prevent filing or approval of commercial marketing applications for our product candidates. We also may encounter problems with the following:

- achieving adequate inventory of clinical-grade materials that meet regulatory agency standards or specifications with consistent and acceptable production yield and costs;
- shortages of qualified personnel, raw materials or key contractors; and
- achieving and maintaining ongoing compliance with cGMP regulations and other requirements of the FDA, EMA or other comparable regulatory agencies.

Failure to comply with applicable regulations could also result in sanctions being imposed on us, including fines, injunctions, civil penalties, a requirement to suspend or put on hold one or more of our clinical studies, failure of regulatory authorities to grant marketing approval of our product candidates, delays, suspension or withdrawal of approvals, license revocation, seizures or recalls of product candidates, operating restrictions and criminal prosecutions, any of which could harm our business.

Developing advanced manufacturing techniques and process controls is required to fully utilize our or our CMOs' facilities. Without further investment, advances in manufacturing techniques may render the facilities and equipment that manufacture our products and product candidates inadequate or obsolete.

A number of the product candidates in our portfolio, if approved by applicable regulatory authorities, may require significant commercial supply to meet market demand. In these cases, we may need to increase, or "scale up," the production process by a significant factor over the initial level of production. If we are unable to do so, are delayed, or if the cost of this scale up is not economically feasible for us or we cannot find a third-party supplier, we may not be able to produce our product candidates in a sufficient quantities to meet future demand.

If our CMO is damaged or destroyed or production at these facilities is otherwise interrupted, our business would be negatively affected.

If any of our CMOs' manufacturing facilities, or the equipment in any such facilities, is either damaged or destroyed, we may not be able to quickly or inexpensively replace such manufacturing capacity or replace it at all. In the event of a temporary or protracted disruption in operations or loss of a facility or its equipment, we may not be able to transfer manufacturing to another third party in the time required to maintain supply. Even if we could transfer manufacturing to another third party, the shift would likely be expensive and time-consuming, particularly since the new facility would need to comply with the necessary regulatory requirements or may require regulatory approval before selling any products manufactured at that facility. Such an event could delay our clinical studies or reduce our commercial product sales.

Currently, we maintain insurance coverage against damage to our property and to cover business interruption and research and development restoration expenses. However, our insurance coverage may not reimburse us, or may not be sufficient to reimburse us, for any expenses or losses we may suffer. We may be unable to meet our requirements for our product candidates if there were a catastrophic event or failure of our current manufacturing facility or processes.

Risks Related to Our Dependence on Third Parties

Maintaining clinical and commercial timelines is dependent on our end-to-end supply chain network to support manufacturing; if we experience problems with our third party suppliers or CMOs we may delay development and/or commercialization of our product candidates.

We rely on our CMOs or our partners for the current production of our product candidates and the acquisition of materials incorporated in or used in the manufacturing or testing of our product candidates. Our CMOs or partners are not our employees, and except for remedies available to us under our agreements with our CMOs or partners, we cannot directly control whether or not they devote sufficient time and resources, including experienced staff, to the manufacturing of supply for our ongoing clinical, nonclinical and preclinical programs. Our CMOs for tab-cel will need to be prepared to undergo pre-approval inspection in connection with our MAA filing and our anticipated BLA submission, and we cannot be certain that we will be able to adequately support them through such inspection nor that they will successfully pass any such inspection.

To meet our projected supply needs for clinical and commercial materials to support our activities through regulatory approval and commercial manufacturing of tabcel, ATA188, any product candidates resulting from our next-generation CAR T programs or any other product candidates, we will need to transition the manufacturing of these materials to a CMO. Regardless of where production occurs, we will need to develop relationships with suppliers of critical starting materials or reagents, increase the scale of

production and demonstrate comparability of the material produced at these facilities to the material that was previously produced. Transferring manufacturing processes and know-how is complex and involves review and incorporation of both documented and undocumented processes that may have evolved over time.

In addition, transferring production to different facilities may require utilization of new or different processes to meet the specific requirements of a given facility. We would expect additional comparability work will also need to be conducted to support the transfer of certain manufacturing processes and process improvements. We cannot be certain that all relevant know-how and data has been adequately incorporated into the manufacturing process until the completion of studies (and the related evaluations) intended to demonstrate the comparability of material previously produced with that generated by us or our CMO.

If we or our CMOs are not able to successfully transfer and produce comparable product candidates, our ability to further develop and manufacture our product candidates may be negatively impacted.

We still may need to identify additional CMOs for continued production of supply for some of our product candidates. Given the nature of our manufacturing processes, the number of CMOs who possess the requisite skill and capability to manufacture our T-cell immunotherapy product candidates, and the critical intermediates or reagents used to manufacture such products, are limited. We have not yet identified alternate suppliers in the event the current CMOs that we utilize are unable to scale production, or if we otherwise experience any problems with them.

Manufacturing cellular therapies is complicated and tightly regulated by the FDA and comparable regulatory authorities around the world, and although alternative third-party suppliers with the necessary manufacturing and regulatory expertise and facilities exist, it could be expensive and take a significant amount of time to arrange for alternative suppliers, transfer manufacturing procedures to these alternative suppliers, and demonstrate comparability of material produced by such new suppliers. New manufactures of any product candidate or intermediate would be required to qualify under applicable regulatory requirements. These manufacturers may not be able to manufacture our product candidates at costs, or in sufficient quantities, or in a timely manner necessary to complete development of our product candidates or make commercially successful products. If we are unable to arrange for alternative third-party manufacturing sources, or to do so on commercially reasonable terms or in a timely manner, we may not be able to complete development of our product candidates, or market or distribute them. In addition, should the FDA or comparable regulatory authorities may require that we conduct additional studies, including bridging comparability testing, and further clinical development or commercial launch of our product candidates could be substantially delayed.

Reliance on third-party manufacturers entails risks to which we would not be subject if we manufactured product candidates ourselves, including reliance on the third party for regulatory compliance and quality assurance, the possibility that the third-party manufacturer does not maintain the financial resources to meet its obligations under the manufacturing agreement, the possibility of breach of the manufacturing agreement by the third party because of factors beyond our control, including a failure to manufacture our product candidates or any products we may eventually commercialize in accordance with our specifications, misappropriation of our proprietary information, including our trade secrets and know-how, the possibility that the third-party does not devote sufficient time or resources to our product candidates or any products we may eventually commercialize based on its own business priorities, the possibility that the third-party is acquired by another party and changes its business priorities, and the possibility of termination or nonrenewal of the agreement by the third party, based on its own business priorities, at a time that is costly or damaging to us. For example, we entered into the Fujifilm MSA and we rely on Fujifilm for the manufacture and supply of a number of our products (if approved) and product candidates. If Fujifilm does not perform its obligations under the Fujifilm MSA adequately or does not devote sufficient time or resources to our products (if approved) or product candidates, our operations may be adversely impacted. In addition, the FDA and other regulatory authorities require that our product candidates and any products that we may eventually commercialize be manufactured according to cGMP, cGTP and similar regulatory jurisdictional standards. These requirements include, among other things, quality control, quality assurance and the maintenance of records and documentation. The FDA or similar foreign regulatory agencies may also implement new standards at any time or change their interpretations and enforcement of existing standards for manufacture, packaging or testing of products. We have limited control over our manufacturers' compliance with these regulations and standards and although we monitor our manufacturers, we depend on them to provide honest and accurate information. Any failure by our third-party manufacturers to comply with cGMP or cGTP or failure to scale up manufacturing processes, including any failure to deliver sufficient quantities of product candidates in a timely manner, could lead to a delay in, or failure to obtain, regulatory approval of any of our product candidates. In addition, such failure could be the basis for the FDA to issue a warning letter, withdraw approvals for product candidates previously granted to us, or take other regulatory or legal action, including recall or seizure of outside supplies of the product candidate, total or partial suspension of production, suspension of ongoing clinical studies, refusal to approve pending applications or supplemental applications, detention or product, refusal to permit the import or export of products, injunction or imposing civil and criminal penalties.

We depend on third party suppliers and testing laboratories for key materials used to produce or test our product candidates. Any significant disruption in our supplier relationships could harm our business. Any significant delay in the supply of a product candidate for an ongoing clinical study could considerably delay initiation or completion of our clinical studies, product testing and potential regulatory approval of our product candidates. If raw materials or components cannot be purchased or fail to meet approved specifications, the commercial launch of our product candidates could be delayed, or there could be a shortage in supply, which could impair our ability to generate revenues from the sale of our product candidates.

Bayer is generally responsible for the conduct and funding of the development and commercialization of ATA2271 and ATA3271.

Pursuant to the Bayer License Agreement, Bayer holds an exclusive, field-limited license to ATA2271 and ATA3271. As a result, other than the development of ATA2271 through Phase 1, Bayer is generally responsible for the development and obtaining and maintaining regulatory approval and commercialization of ATA2271 and ATA3271.

Although we have joint governance and certain decision making rights, we do not control the development activities being conducted or that may be conducted in the future by Bayer, including, but not limited to, the timing of initiation, termination or completion of clinical trials, the analysis of data arising out of those clinical trials or the timing of release of data concerning those clinical trials, which may impact our ability to report on Bayer's results. Bayer may conduct these activities more slowly or in a different manner than we would if we controlled the development of ATA2271 after Phase 1 and ATA3271. Bayer is responsible for submitting future applications to the FDA and other regulatory authorities for approval of ATA2271 and ATA3271 and will be the owner of marketing approvals issued by the FDA and other regulatory authorities for ATA2271 and ATA3271, if approved. If the FDA or other regulatory authorities approve ATA2271 and/or ATA3271, Bayer will also be responsible for the marketing and sale of the resulting product. However, we cannot control whether Bayer will devote sufficient attention and resources to the development of ATA2271 and/or ATA3271, Bayer may elect not to proceed with the commercialization of the resulting product in one or more countries, unless otherwise specified in the Bayer License Agreement.

In March 2021, we entered into the Bayer Manufacturing Agreement for the supply of allogeneic mesothelin-directed CAR T-cell therapies for clinical trials. Delays to the activities contemplated by the Bayer Manufacturing Agreement may result in a delay in the ATA2271 and/or ATA3271 programs and would delay and could prevent us from obtaining revenues for this product candidate.

Disputes may arise between us and Bayer, which may delay or cause the termination of any clinical trials of ATA2271 and/or ATA3271, result in significant litigation or cause Bayer to act in a manner that is not in our best interest. The costs associated with the continuing development of ATA2271 and/or ATA3271 may cause Bayer to reconsider the terms of its investment and seek to amend or terminate our agreement or to suspend the development of ATA2271 and/or ATA3271. If development of ATA2271 and/or ATA3271 does not progress for these or any other reasons, we would not receive milestone payments or royalties on product sales from Bayer with respect to ATA2271 and/or ATA3271. If the results of one or more clinical trials with ATA2271 and/or ATA3271 and/or ATA327

If Bayer abandons ATA2271 and/or ATA3271, it would result in a delay in or could prevent us from commercializing ATA2271 and/or ATA3271 and would delay and could prevent us from obtaining revenues for this product candidate. If Bayer abandons development of ATA2271 and/or ATA3271 prior to regulatory approval or if it elects not to proceed with commercialization of the resulting product following regulatory approval, we would have to seek a new partner for development or commercialization, curtail or abandon that development or commercialization, or undertake and fund the development of ATA2271 and/or ATA3271 or commercialization of the resulting product ourselves. If we seek a new partner but are unable to do so on acceptable terms, or at all, or do not have sufficient funds to conduct the development or commercialization of ATA2271 and/or ATA3271 ourselves, we would have to curtail or abandon that development or commercialization, which could harm our business.

We are dependent on Pierre Fabre for the potential commercialization of tab-cel in the European Union and several countries outside the United States. The failure of Pierre Fabre to meet its contractual, regulatory or other obligations could adversely affect our business.

We have entered into the Pierre Fabre Commercialization Agreement for tab-cel in Europe and select emerging markets in the Territory for EBV-positive cancers. As a result, we are entirely dependent on Pierre Fabre for marketing and commercialization, including negotiation of pricing and reimbursement, of tab-cel, if approved, in the Territory. The timing and amount of any milestone and royalty payments we may receive under these agreements, as well as the commercialization of tab-cel in the Territory, will depend on, among other things, the efforts, allocation of resources, negotiation of pricing and reimbursement and successful commercialization of tab-cel by Pierre Fabre in the Territory.

We are in the process of negotiating various ancillary agreements as contemplated by the Pierre Fabre Commercialization Agreement, including an agreement for the manufacture and supply of tab-cel to Pierre Fabre for commercialization in the Territory. Delays to the negotiation and execution of the ancillary agreements, or the activities contemplated by the ancillary agreements, including supply of tab-cel from our CMO for Pierre Fabre, may result in a delay to the commercialization of tab-cel in the Territory and would delay and could prevent us from obtaining revenues for tab-cel in the Territory.

Under the terms of the Pierre Fabre Commercialization Agreement, if we receive the EU marketing authorization for tab-cel in patients with EBV+ PTLD, we are required to transfer the marketing authorization to Pierre Fabre. Pierre Fabre will be responsible for obtaining all other regulatory approvals in the Territory and maintaining all regulatory approvals in the Territory. We will depend on Pierre Fabre to comply with numerous and varying regulatory requirements governing, if and when applicable, the manufacture, quality control, further development, labeling, packaging, storage, distribution, adverse event reporting, safety surveillance, import, export, advertising, promotion, recordkeeping and reporting of safety and other post-marketing information. We do not control the individual efforts of Pierre Fabre and have limited ability to terminate the Pierre Fabre Commercialization Agreement if Pierre Fabre does not perform as expected. The failure of Pierre Fabre to devote sufficient time and effort to comply with regulatory requirements and maintain the EU marketing authorization and other regulatory approvals in the Territory and/or to meet their obligations to us, could have an adverse impact on our financial results and operations.

We also depend on Pierre Fabre to comply with all applicable laws relative to the commercialization of tab-cel in the Territory. We do not control the individual efforts of Pierre Fabre and have limited ability to terminate the Pierre Fabre Commercialization Agreement if Pierre Fabre does not perform as expected. The failure of Pierre Fabre to devote sufficient time and effort to the commercialization of tab-cel; to meet their obligations to us, including for future royalty and milestone payments; to adequately deploy business continuity plans in the event of a crisis; and/or to satisfactorily resolve significant disagreements with us or address other factors could have an adverse impact on our financial results and operations. In addition, if Pierre Fabre violates, or are alleged to have violated, any laws or regulations during the performance of their obligations for us, it is possible that we could suffer financial and reputational harm or other negative outcomes, including possible legal consequences.

Any termination, breach or expiration of the Pierre Fabre Commercialization Agreement or ancillary agreements, once entered into, could have a material adverse effect on our financial position by reducing or eliminating the potential for us to receive fees, milestones and royalties. In such an event, we may be required to devote additional efforts and to incur additional costs associated with the transfer of regulatory approvals and commercialization of tab-cel in the Territory. Alternatively, we may attempt to identify and transact with a new commercialization partner, but there can be no assurance that we would be able to identify a suitable partner or transact on terms that are favorable to us.

We may not realize the benefits of strategic alliances that we may form in the future or of potential future product acquisitions or licenses.

We may desire to form additional strategic alliances, create joint ventures or collaborations, enter into licensing arrangements with third parties or acquire products or business, in each case that we believe will complement or augment our existing business. These relationships or transactions, or those like them, may require us to incur nonrecurring and other charges, increase our near- and long-term expenditures, issue securities that dilute our existing stockholders, reduce the potential profitability of the products that are the subject of the relationship or disrupt our management and business. In addition, we face significant competition in seeking appropriate strategic alliances and transactions and the negotiation process is time-consuming and complex and there can be no assurance that we can enter into any of these transactions even if we desire to do so. Moreover, we may not be successful in our efforts to establish a strategic alliance or other alternative arrangements for any future product candidates and programs because our research and development pipeline may be insufficient, our product candidates and programs may be deemed to be at too early a stage of development for collaborative effort and third parties may not view our product candidates and programs as having the requisite potential to demonstrate a positive benefit/risk profile. Any delays in entering into new strategic alliances argements related to our product candidates could also delay the development and commercialization of our product candidates and reduce their competitiveness even if they reach the market.



If we license products or acquire businesses, we may not be able to realize the benefit of these transactions if we are unable to successfully integrate them with our existing operations and company culture. We cannot be certain that, following an acquisition or license, we will achieve the financial or strategic results that would justify the transaction.

Risks Related to Our Intellectual Property

If we are unable to obtain and maintain sufficient intellectual property protection for our product candidates, or if the scope of the intellectual property protection is not sufficiently broad, our ability to commercialize our product candidates successfully and to compete effectively may be adversely affected.

We rely upon a combination of patents, trademarks, trade secrets and confidentiality agreements – both that we own or possess or that are owned or possessed by our partners that are in-licensed to us – to protect the intellectual property related to our technology and product candidates. When we refer to "our" technologies, inventions, patents, patent applications or other intellectual property rights, we are referring to both the rights that we own or possess as well as those that we in-license, many of which are critical to our intellectual property protection and our business. For example, the product candidates and platform technology we have licensed from our partners are protected primarily by patent or patent applications of our partners that we have licensed and as confidential know-how and trade secrets. If the intellectual property that we rely on is not adequately protected, competitors may be able to use our technologies and erode or negate any competitive advantage we may have.

The patentability of inventions and the validity, enforceability and scope of patents in the biotechnology field is uncertain because it involves complex legal, scientific and factual considerations, and it has in recent years been the subject of significant litigation. Moreover, the standards applied by the U.S. Patent and Trademark Office (USPTO) and non-U.S. patent offices in granting patents are not always applied uniformly or predictably. For example, there is no uniform worldwide policy regarding patentable subject matter or the scope of claims allowable in biotechnology patents.

There is no assurance that all potentially relevant prior art relating to our patents and patent applications is known to us or has been found in the instances where searching was done. We may be unaware of prior art that could be used to invalidate an issued patent or prevent a pending patent application from issuing as a patent. There also may be prior art of which we are aware, but which we do not believe affects the validity or enforceability of a claim of one of our patents or patent applications, which may, nonetheless, ultimately be found to affect the validity or enforceability of such claim. As a consequence of these and other factors, our patent applications may fail to result in issued patents with claims that cover our product candidates in the U.S. or in other countries.

Even if patents have issued or do successfully issue from patent applications, and even if these patents cover our product candidates, third parties may challenge the validity, enforceability or scope thereof, which may result in these patents being narrowed, invalidated or held to be unenforceable. No assurance can be given that if challenged, our patents would be declared by a court to be valid or enforceable.

Even if unchallenged, our patents and patent applications or other intellectual property rights may not adequately protect our intellectual property, provide exclusivity for our product candidates or prevent others from designing around our claims. The possibility exists that others will develop products on an independent basis which have the same effect as our product candidates and which do not infringe our patents or other intellectual property rights, or that others will design around the claims of patents that we have had issued that cover our product candidates. If the breadth or strength of protection provided by our patents and patent applications with respect to our product candidates is threatened, it could jeopardize our ability to commercialize our product candidates and dissuade companies from collaborating with us.

We may also desire to seek a license from a third party who owns intellectual property that may be useful for providing exclusivity for our product candidates, or for providing the ability to develop and commercialize a product candidate in an unrestricted manner. There is no guarantee that we will be able to obtain a license from such a third party on commercially reasonable terms, or at all.

In addition, the USPTO and various foreign governmental patent agencies require compliance with a number of procedural, documentary, fee payment and other similar provisions during the patent application process. While an inadvertent lapse can in many cases be cured by payment of a late fee or by other means in accordance with the applicable rules, there are situations in which noncompliance can result in abandonment or lapse of the patent application, resulting in partial or complete loss of patent rights in the relevant jurisdiction.

We and our partners have filed a number of patent applications covering our product candidates or methods of using or making those product candidates. We cannot offer any assurances about which, if any, patents will be issued with respect to these pending patent applications, the breadth of any such patents that are ultimately issued or whether any issued patents will be found invalid and unenforceable or will be threatened by third parties. Because patent applications in the U.S. and most other countries are confidential for a period of time after filing, and some remain so until issued, we cannot be certain that we or our partners were the first to file any patent application related to a product candidate. We or our partners may also become involved in proceedings regarding our patents, including patent infringement lawsuits, interference or derivation proceedings, oppositions, and *inter partes* and post-grant review proceedings before the USPTO the European Patent Office and other non-U.S. patent offices.

Even if granted, patents have a limited lifespan. In the U.S., the natural expiration of a patent generally occurs 20 years after it is filed. Although various extensions may be available if certain conditions are met, the life of a patent and the protection it affords is limited. If we encounter delays in our clinical studies or in obtaining regulatory approvals, the period of time during which we could exclusively market any of our product candidates under patent protection, if approved, could be reduced. Even if patents covering our product candidates are obtained, once the patent life has expired for a product, we may be vulnerable to competition from biosimilar products, as we may be unable to prevent competitors from entering the market with a product that is similar or identical to our product candidates.

Furthermore, the research resulting in certain of our licensed patent rights and technology was funded by the U.S. government. As a result, the government has certain rights to these patent rights and technology. When new technologies are developed with government funding, the government generally obtains certain rights in any resulting patents, including a non-exclusive license authorizing the government to practice the invention for or on behalf of the U.S. These rights may permit the government to disclose confidential information on which we rely to third parties and to exercise march-in rights to use or allow third parties to use our licensed technology. The government can exercise its march-in rights if it determines that action is necessary because we fail to achieve practical application of the government-funded technology, because action is necessary to alleviate health or safety needs, to meet requirements of federal regulations or to give preference to U.S. industry. In addition, our rights in any inventions that result from government-funded research may be subject to certain requirements to manufacture products embodying these inventions in the U.S.

If we are sued for infringing the intellectual property rights of third parties, the resulting litigation could be costly and time-consuming and could prevent or delay our development and commercialization efforts.

Our commercial success depends, in part, on us and our partners not infringing the patents and proprietary rights of third parties. There is a substantial amount of litigation and other adversarial proceedings, both within and outside the U.S., involving patent and other intellectual property rights in the biotechnology and pharmaceutical industries, including patent infringement lawsuits, interference or derivation proceedings, oppositions, and *inter partes* and post-grant review proceedings before the USPTO and non-U.S. patent offices. Numerous U.S. and non-U.S. issued patents and pending patent applications owned by third parties exist in the fields in which we are developing and may develop our product candidates. As the biotechnology and pharmaceutical industries expand and more patents are issued, the risk increases that our product candidates may be subject to claims of infringement of third parties; patent rights, as it may not always be clear to industry participants, including us, which patents cover various types of products or methods of use. The coverage of patents is subject to interpretation by the courts, and the interpretation is not always uniform or predictable.

Third parties may assert infringement claims against us based on existing or future intellectual property rights, alleging that we are employing their proprietary technology without authorization. There may be third-party patents or patent applications with claims to materials, formulations, methods of manufacture or methods for treatment related to the use or manufacturing of our product candidates that we failed to identify. For example, patent applications covering our product candidates could have been filed by others without our knowledge, since these applications generally remain confidential for some period of time after their filing date. Even pending patent applications that have been published, including some of which we are aware, could be later amended in a manner that could cover our product candidates or their use or manufacture. In addition, we may have analyzed patents or patent applications of third parties that we believe are relevant to our activities and believe that we are free to operate in relation to any of our product candidates or our activities infringing their claims.

If we or our partners are sued for patent infringement, we would need to demonstrate that our product candidates, products and methods either do not infringe the patent claims of the relevant patent or that the patent claims are invalid, and we may not be able to do this. Proving that a patent is invalid is difficult and even if we are successful in the relevant proceedings, we may incur substantial costs and the time and attention of our management and scientific personnel could be diverted from other activities. If any issued third-party patents were held by a court of competent jurisdiction to cover aspects of our materials, formulations, methods of manufacture or methods for treatment, we could be forced, including by court order, to cease developing, manufacturing or commercializing the relevant product candidate until the relevant patent expired. Alternatively, we may desire or be required to obtain

a license from such third party in order to use the infringing technology and to continue developing, manufacturing or marketing the infringing product candidate. However, we may not be able to obtain any required license on commercially reasonably terms, or at all. Even if we were able to obtain a license, the rights may be nonexclusive, which could result in our competitors gaining access to the same intellectual property licensed to us.

We may face claims that we misappropriated the confidential information or trade secrets of a third party. If we are found to have misappropriated a third party's trade secrets, we may be prevented from further using these trade secrets, which could limit our ability to develop our product candidates.

Defending against intellectual property claims could be costly and time consuming, regardless of the outcome. Thus, even if we were to ultimately prevail, or to settle before a final judgment, any litigation could burden us with substantial unanticipated costs. In addition, litigation or threatened litigation could result in significant demands on the time and attention of our management team, distracting them from the pursuit of other company business. During the course of any intellectual property litigation, there could be public announcements of the results of hearings, rulings on motions, and other interim proceedings in the litigation and these announcements may have negative impact on the perceived value of our product candidates, programs or intellectual property. In the event of a successful intellectual property claim against us, we may have to pay substantial damages, including treble damages and attorneys' fees if we are found to have willfully infringed a patent, or to redesign our infringing product candidates, which may be impossible or require substantial time and monetary expenditure. In addition to paying monetary damages, we may lose valuable intellectual property rights or personnel and the parties making claims against us may obtain injunctive or other equitable relief, which could impose limitations on the conduct of our business. We may also elect to enter into license agreements in order to settle patent infringement claims prior to litigation, and any of these license agreements may require us to pay royalties and other fees that could be significant. As a result of all of the foregoing, any actual or threatened intellectual property claim could prevent us from developing or commercializing a product candidate or force us to cease some aspect of our business operations.

We may not be able to protect our intellectual property rights throughout the world.

Filing, prosecuting, enforcing and defending patents on all of our product candidates in all countries throughout the world would be prohibitively expensive. Our intellectual property rights in certain countries outside the U.S. may be less extensive than those in the U.S. In addition, the laws of certain foreign countries do not protect intellectual property rights to the same extent as laws in the U.S. Consequently, we and our partners may not be able to prevent third parties from practicing our inventions in countries outside the U.S., or from selling or importing infringing products made using our inventions in and into the U.S. or other jurisdictions. Competitors may use our technologies in jurisdictions where we have not obtained patent protection or where we do not have exclusive rights under the relevant patents to develop their own products and, further, may export otherwise-infringing products to territories where we and our partners have patent protection but where enforcement is not as strong as that in the U.S. These infringing products may compete with our product candidates in jurisdictions where we do not have exclusive rights or where we do not have exclusive rights under the relevant patents, or our patent claims and other intellectual property rights may not be effective or sufficient to prevent them from so competing.

Many companies have encountered significant problems in protecting and defending intellectual property rights in foreign jurisdictions. The legal systems of certain countries, particularly certain developing countries, do not favor the enforcement of patents and other intellectual property protection, particularly those relating to biopharmaceuticals, which could make it difficult for us and our partners to stop the infringement of our patents or marketing of competing products in violation of our intellectual property rights generally. Proceedings to enforce our patent rights in foreign jurisdictions could result in substantial costs and divert our attention from other aspects of our business, could put our patents at risk of being invalidated or interpreted narrowly, could put our patent applications at risk of not issuing, and could provoke third parties to assert claims against us or our partners. We or our partners may not prevail in any lawsuits that we or our licensors initiate, and even if we or our licensors are successful the damages or other remedies awarded, if any, may not be commercially meaningful.

We have in-licensed a significant portion of our intellectual property from our partners. If we breach any of our license agreements with these partners, we could lose the ability to continue the development and potential commercialization of one or more of our product candidates.

We hold rights under license agreements with our partners, including MSK, QIMR Berghofer and Moffitt that are important to our business. Our discovery and development platform is built, in part, around patent rights in-licensed from our partners. Under our existing license agreements, we are subject to various obligations, including diligence obligations with respect to development and commercialization activities, payment obligations upon achievement of certain milestones and royalties on product sales. If there is any conflict, dispute, disagreement or issue of nonperformance between us and our counterparties regarding our rights or obligations under these license agreements, including any conflict, dispute or disagreement arising from our failure to satisfy diligence or payment obligations, we may be liable to pay damages and our counterparties may have a right to terminate the affected license. The

termination of any license agreement with one of our partners would materially adversely affect our ability to utilize the intellectual property that is subject to that license agreement in our drug discovery and development efforts, our ability to enter into future collaboration, licensing and/or marketing agreements for one or more affected product candidates and our ability to commercialize the affected product candidates. Furthermore, a disagreement under any of these license agreements may harm our relationship with the partner, which could have negative impacts on other aspects of our business.

We may become involved in lawsuits to protect or enforce our intellectual property, which could be expensive, time-consuming and unsuccessful and have a material adverse effect on the success of our business.

Third parties may infringe our patents or misappropriate or otherwise violate our intellectual property rights. Our patent applications cannot be enforced against third parties practicing the technology claimed in these applications unless and until a patent issues from the applications, and then only to the extent the issued claims cover the technology. In the future, we or our partners may elect to initiate legal proceedings to enforce or defend our or our partners' intellectual property rights, to protect our or our partners' trade secrets or to determine the validity or scope of our intellectual property rights. Any claims that we or our partners assert against perceived infringers could also provoke these parties to assert counterclaims against us or our partners alleging that we or our partners infringe their intellectual property rights or that our intellectual property rights are invalid.

Interference or derivation proceedings provoked by third parties, brought by us or our partners, or brought by the USPTO or any non-U.S. patent authority may be necessary to determine the priority of inventions or matters of inventorship with respect to our patents or patent applications. We or our partners may also become involved in other proceedings, such as reexamination or opposition proceedings, *inter partes* review, post-grant review or other pre-issuance or post-grant proceedings in the USPTO or its foreign counterparts relating to our intellectual property or the intellectual property of others. An unfavorable outcome in any of these proceedings could require us or our partners to cease using the related technology and commercializing our product candidates, or to attempt to license rights to it from the prevailing party. Our business could be harmed if the prevailing party does not offer us or our partners a license on commercially reasonable terms if any license is offered at all. Even if we or our licensors obtain a license, it may be non-exclusive, thereby giving our competitors access to the same technologies licensed to us or our licensors. In addition, if the breadth or strength of protection provided by our patents and patent applications is threatened, it could dissuade companies from collaborating with us to license, develop or commercialize current or future product candidates.

Any intellectual property proceedings can be expensive and time-consuming. Our or our partners' adversaries in these proceedings may have the ability to dedicate substantially greater resources to prosecuting these legal actions than we or our partners can. Accordingly, despite our or our partners' efforts, we or our partners may not be able to prevent third parties from infringing upon or misappropriating our intellectual property rights, particularly in countries where the laws may not protect our rights as fully as in the U.S. Even if we are successful in the relevant proceedings, we may incur substantial costs and the time and attention of our management and scientific personnel could be diverted from other activities. We could be found liable for monetary damages, including treble damages and attorneys' fees, if we are found to have willfully infringed a patent. In addition, in an infringement proceeding, a court may decide that one or more of our patents is invalid or unenforceable, in whole or in part, or may refuse to stop the other party from using the technology at issue on the grounds that our patents do not cover the technology in question. An adverse result in any litigation proceeding could put one or more of our patents at risk of being invalidated, held unenforceable or interpreted narrowly.

Furthermore, because of the substantial amount of discovery required in connection with intellectual property litigation, there is a risk that some of our confidential information could be compromised by disclosure during this type of litigation. In addition, there could be public announcements of the results of hearings, motions or other interim proceedings or developments.

If we are unable to protect the confidentiality of our trade secrets and other proprietary information, the value of our technology could be materially adversely affected and our business could be harmed.

In addition to seeking the protection afforded by patents, we rely on trade secret protection and confidentiality agreements to protect proprietary know-how that is not patentable or that we elect not to patent, processes for which patents are difficult to enforce, and other elements of our technology, discovery and development processes that involve proprietary know-how, information or technology that is not covered by patents. The T-cell immunotherapy product candidates and platform technology we have licensed from our partners are protected primarily as confidential know-how and trade secrets. Any disclosure to or misappropriation by third parties of our confidential proprietary information could enable competitors to quickly duplicate or surpass our technological achievements, including by enabling them to develop and commercialize products substantially similar to or competitive with our product candidates, thus eroding our competitive position in the market.

Trade secrets can be difficult to protect. We seek to protect our proprietary technology and processes, in part, by entering into confidentiality agreements and invention assignment agreements with our employees, consultants, CMOs, and outside scientific advisors, contractors and collaborators. These agreements are designed to protect our proprietary information. Although we use reasonable efforts to protect our trade secrets, our employees, consultants, contractors, or outside scientific advisors might intentionally or inadvertently disclose our trade secrets or confidential, proprietary information to competitors. In addition, competitors may otherwise gain access to our trade secrets or independently develop substantially equivalent information and techniques. If any of our confidential proprietary information to compete with us, which could harm our competitive position.

Enforcing a claim that a third party illegally obtained and is using any of our trade secrets is expensive and time consuming, and the outcome is unpredictable. In addition, the laws of certain foreign countries do not protect proprietary rights such as trade secrets to the same extent or in the same manner as the laws of the U.S. Misappropriation or unauthorized disclosure of our trade secrets to third parties could impair our competitive advantage in the market and could materially adversely affect our business, results of operations and financial condition.

Risks Related to Commercialization of Our Product Candidates

Our commercial success depends upon attaining significant market acceptance of our product candidates, if approved, among physicians, patients, healthcare payors and the medical community, including hospitals and outpatient clinics.

Even if we obtain regulatory approval for any of our product candidates that we may develop or acquire in the future, the product may not gain market acceptance among physicians, healthcare payors, patients or the medical community that supports our product development efforts, including hospitals and outpatient clinics. Market acceptance of any of our product candidates for which we receive approval depends on a number of factors, including:

- the efficacy and safety of the product candidates as demonstrated in clinical studies;
- the clinical indications and patient populations for which the product candidate is approved;
- acceptance by physicians and patients of the drug as a safe and effective treatment;
- the administrative and logistical burden of treating patients;
- the ability to identify in a timely manner the appropriate patients who will benefit from specific therapy;
- the consideration of novel cellular therapies by physicians, hospitals and third-party payors;
- the potential and perceived advantages of product candidates over alternative treatments;
- the safety of product candidates seen in a broader patient group, including its use outside the approved indications;
- any restrictions on use together with other medications;
- the prevalence and severity of any side effects;
- product labeling or product insert requirements of the FDA or other regulatory authorities;
- the timing of market introduction of our products as well as competitive products;
- the development of manufacturing and distribution processes for our product candidates;
- the cost of treatment in relation to alternative treatments;
- the availability of coverage and adequate reimbursement from, and our ability to negotiate pricing with, third-party payors and government authorities;
- relative convenience and ease of administration; and
- the effectiveness of our sales and marketing efforts and those of our collaborators.

Even if we are able to commercialize our product candidates, the products may not receive coverage and adequate reimbursement from third-party payors in the U.S. and in other countries in which we seek to commercialize our products, which could harm our business.

Our ability to commercialize any product successfully will depend, in part, on the extent to which coverage and adequate reimbursement for these products and related treatments will be available from government health administration authorities, private health insurers and other organizations.

Government authorities and third-party payors, such as private health insurers and health maintenance organizations, determine which medications they will cover and establish reimbursement levels. A primary trend in the healthcare industry is cost containment. Government authorities and third-party payors have attempted to control costs by limiting coverage and the amount of reimbursement for particular medications. Increasingly, third-party payors are requiring that drug companies provide them with predetermined discounts from list prices and are challenging the prices charged for medical products. Third-party payors may also seek additional clinical evidence, beyond the data required to obtain regulatory approval, demonstrating clinical benefits and value in specific patient populations before covering our products for those patients. We cannot be sure that coverage and adequate reimbursement will be available for any product that we commercialize and, if reimbursement is available, what the level of reimbursement will be available for any product that we commercialize and, and higher patient copayments or other administrative burdens could lead to reduced demand from patients or healthcare professionals. This could particularly be the case in a challenging economic climate. Coverage and reimbursement may impact the demand for, or the price of, any product candidate for which we obtain regulatory approval, and ultimately our ability to successfully commercialize any product candidate for which we obtain regulatory approval.

There may be significant delays in obtaining coverage and reimbursement for newly approved drugs, and coverage may be more limited than the purposes for which the drug is approved by the FDA or comparable foreign regulatory authorities. Moreover, eligibility for coverage and reimbursement does not imply that any drug will be paid for in all cases or at a rate that covers our costs, including research, development, manufacture, sale and distribution. Interim reimbursement levels for new drugs, if applicable, may also not be sufficient to cover our costs and may only be temporary. Reimbursement rates may vary according to the use of the drug and the clinical setting in which it is used, may be based on reimbursement levels already set for lower cost drugs and may be incorporated into existing payments for other services. Net prices for drugs may be reduced by mandatory discounts or rebates required by government healthcare programs or private payors and by any future relaxation of laws that presently restrict imports of drugs from countries where they may be sold at lower prices than in the U.S. Coverage and reimbursement policies for drug products can differ significantly from payor to payor as there is no uniform policy of coverage and reimbursement for drug products among third-party payors in the U.S. often rely upon Medicare coverage policy and payment limitations in setting their own reimbursement policies. There may be significant delays in obtaining coverage and reimbursement as the process of determining coverage and reimbursement is often time consuming and costly which will require us to provide scientific and clinical support for the use of our products to each payor separately, with no assurance that coverage or adequate reimbursement will be obtained. It is difficult to predict at this time what government authorities and third-party payors will decide with respect to coverage and reimbursement for our drug products. Our inability to prayor deproducts and private payors for any approved products tha

Current and future legislation, including potentially unfavorable pricing regulations or other healthcare reform initiatives, may increase the difficulty and cost for us to obtain regulatory approval of and commercialize our product candidates and affect the prices we may obtain.

The regulations that govern, among other things, regulatory approvals, coverage, pricing and reimbursement for new drug products vary widely from country to country. In the U.S. and some foreign jurisdictions, there have been a number of legislative and regulatory changes and proposed changes regarding the healthcare system that could prevent or delay regulatory approval of our product candidates, restrict or regulate post-approval activities and affect our ability to successfully sell any product candidates for which we obtain regulatory approval. In particular, in March 2010, the Patient Protection and Affordable Care Act, as amended by the Health Care and Education Reconciliation Act, collectively, the Affordable Care Act, was enacted, which substantially changed the way healthcare is financed by both governmental and private insurers, and continues to significantly impact the U.S. pharmaceutical industry. The Affordable Care Act and its implementing regulations, among other things, addressed a new methodology by which rebates owed by manufacturers under the Medicaid Drug Rebate Program are calculated for certain drugs and biologics, including our product candidates, increased the minimum Medicaid rebates owed by manufacturers under the Medicaid Drug Rebate Program, extended the Medicaid Drug Rebate Program to utilization of prescriptions of individuals enrolled in Medicaid managed care organizations, subjected manufacturers to new annual fees and taxes for certain branded prescription drugs, provided incentives to programs that increase the federal government's comparative effectiveness research and established a new Medicare Part D coverage gap discount program.

Other legislative changes have been proposed and adopted in the U.S. since the Affordable Care Act was enacted. In August 2011, the Budget Control Act of 2011, among other things, created measures for spending reductions by the U.S. Congress. A Joint Select Committee on Deficit Reduction, tasked with recommending a targeted deficit reduction of at least \$1.2 trillion for the years 2013 through 2021, was unable to reach required goals, thereby triggering the legislation's automatic reduction to several government programs. This includes aggregate reductions of Medicare payments to providers of 2% per fiscal year, which went into effect in April 2013, and, due to subsequent legislative amendments, will remain in effect into 2031 unless additional Congressional action is taken. Though COVID-19 relief legislation suspended the 2% Medicare sequester from May 1, 2020 through March 31, 2022, sequestration started again on April 1, 2022. From April 1 to June 30, 2022, payment for Medicare fee-for-service claims will be adjusted

downwards by 1%; beginning July 1, 2022, the payment will be adjusted downwards by 2%. In January 2013, the American Taxpayer Relief Act of 2012 (the ATRA) was enacted which, among other things, further reduced Medicare payments to several providers, including hospitals and outpatient clinics, and increased the statute of limitations period for the government to recover overpayments to providers from three to five years.

There have been judicial and Congressional challenges to numerous elements of the Affordable Care Act, as well as efforts by both the executive and legislative branches of the federal government to repeal or replace certain aspects of the Affordable Care Act. In addition, the U.S. Congress has considered legislation that would repeal or repeal and replace all or part of the Affordable Care Act. While the U.S. Congress has not passed comprehensive repeal legislation, it has enacted laws that modify certain provisions of the Affordable Care Act, such as removing penalties, starting January 1, 2019, for not complying with the Affordable Care Act's individual mandate to carry health insurance, eliminating the implementation of certain mandated fees, and increasing the point-of-sale discount that is owed by pharmaceutical manufacturers who participate in Medicare Part D. On June 17, 2021, the U.S. Supreme Court dismissed a legal challenge to the law brought by several states arguing that, without the individual mandate, the entire Affordable Care Act was unconstitutional. The Supreme Court dismissed the lawsuit without ruling on the merits of the states' constitutionality arguments. While the legal challenge to the Affordable Care Act was pending, on January 28, 2021, President Biden issued an executive order to initiate a special enrollment period from February 15, 2021 through August 15, 2021 for purposes of obtaining health insurance coverage through the Affordable Care Act marketplace. The executive order also instructed certain governmental agencies to review and reconsider their existing policies that create unnecessary barriers to obtaining access to health insurance coverage through Medicaid or the Affordable Care Act. In the future, there may be additional challenges and/or amendments to the Affordable Care Act and our business.

There have been, and likely will continue to be, legislative and regulatory proposals at the foreign, federal and state levels directed at broadening the availability of healthcare and containing or lowering the cost of healthcare. We cannot predict the initiatives that may be adopted in the future. The continuing efforts of governments, insurance companies, managed care organizations and other payors of healthcare services to contain or reduce costs of healthcare, including by imposing price controls, may adversely affect the demand for our product candidates for which we obtain regulatory approval and our ability to set a price that we believe is fair for our products. Any reduction in reimbursement from Medicare or other government programs may result in a similar reduction in payments from private payors.

Legislative and regulatory proposals have been made to expand post-approval requirements and restrict sales and promotional activities for pharmaceutical products. We cannot be sure whether additional legislative changes will be enacted, or whether the U.S. or foreign regulations, guidance or interpretations will be changed, or what the impact of these changes on the regulatory approvals of our product candidates, if any, may be. In the U.S., the EU and other potentially significant markets for our product candidates, government authorities and third-party payors are increasingly attempting to limit or regulate the price of medical products and services, particularly for new and innovative products and therapies, which has resulted in lower average selling prices for certain products in certain markets. In the U.S., there have been several Congressional inquiries and proposed and enacted federal and state legislation designed to, among other things, bring more transparency to drug pricing, review the relationship between pricing and manufacturer patient programs, and reform government program reimbursement methodologies for drugs. For example, the Consolidated Appropriations Act of 2021 included several drug price reporting and transparency measures, such as a new requirement for certain Medicare plans to develop tools to display Medicare Part D prescription drug benefit information in real time and for group and health insurance issuers to report information on pharmacy benefit and drug costs to the Secretaries of Health and Human Services, Labor, and the Treasury.

There have also been administrative developments in the U.S. related to drug pricing. For example, included in the Consolidated Appropriations Act of 2021 were several drug price reporting and transparency measures, such as a new requirement for certain Medicare plans to develop tools to display Medicare Part D prescription drug benefit information in real time and for group and health insurance issuers to report information on pharmacy benefit and drug costs to the Secretaries of the Departments of Health and Human Services, Labor and the Treasury. Additionally, on July 9, 2021, President Biden issued an Executive Order to promote competition in the U.S. economy that included several initiatives addressing prescription drugs. Among other provisions, the Executive Order stated that the Biden administration will "support aggressive legislative reforms that would lower prescription drugs, including by allowing Medicare to negotiate drug prices, by imposing inflation caps, and through other related reforms." In response to the Executive Order, on September 9, 2021, the Department of Health and Human Services issued a Comprehensive Plan for Addressing High Drug Prices that identified potential legislative policies and administrative tools that Congress and the agency can pursue in order to make drug prices more affordable and equitable, improve and promote competition throughout the prescription drug industry, and foster scientific innovation. Most recently, on February 2, 2022, the Biden Administration signaled its continued commitment to the Cancer Moonshot initiative, which was initially launched in 2016. In its announcement, the administration noted that its new goals under the initiative

include addressing inequities in order to ensure broader access to cutting-edge cancer therapeutics and investing in a robust pipeline for new treatments. At the state level, legislatures have increasingly passed legislation and implemented regulations designed to control pharmaceutical and biological product pricing, including price or patient reimbursement constraints, discounts, restrictions on certain product access and marketing cost disclosure and transparency measures, and, in some cases, to encourage importation from other countries and bulk purchasing. Furthermore, the increased emphasis on managed healthcare in the U.S. and on country and regional pricing and reimbursement controls in the EU will put additional pressure on product pricing, reimbursement and usage, which may adversely affect our future product sales. These pressures can arise from rules and practices of managed care groups, other insurers, judicial decisions and governmental laws and regulations related to Medicare, Medicaid and healthcare reform, pharmaceutical reimbursement policies and pricing in general. Further, it is possible that additional governmental action is taken in response to the COVID-19 pandemic.

In addition, there is significant uncertainty regarding the reimbursement status of newly approved healthcare products. We may need to conduct expensive pharmacoeconomic studies in order to demonstrate the cost-effectiveness of our products. If third-party payors do not consider our products to be cost-effective compared to other therapies, the payors may not cover our products after approved as a benefit under their plans or, if they do, the level of payment may not be sufficient to allow us to sell our products on a profitable basis.

Price controls may be imposed in foreign markets, which may adversely affect our future profitability.

In some countries, particularly member states of the EU and the U.K., the pricing of prescription drugs is subject to governmental control. In these countries, pricing negotiations with governmental authorities can take considerable time after receipt of regulatory approval for a product. In addition, there can be considerable pressure by governments and other stakeholders on prices and reimbursement levels, including as part of cost containment measures. Political, economic and regulatory developments may further complicate pricing negotiations, and pricing negotiations may continue after reimbursement has been obtained. Reference pricing used by various EU member states and parallel distribution, or arbitrage between low-priced and high-priced member states, can further reduce prices. In some countries, we, or our collaborators, may be required to conduct a clinical study or other studies that compare the cost-effectiveness of our product candidates to other available therapies in order to obtain or maintain reimbursement or pricing approval. Publication of discounts by third-party payors or authorities may lead to further pressure on the prices or reimbursement levels within the country of publication and other countries. If reimbursement of our products is unavailable or limited in scope or amount, or if pricing is set at unsatisfactory levels, our business could be adversely affected.

We face substantial competition, which may result in others discovering, developing or commercializing products before or more successfully than we do.

We face competition from numerous pharmaceutical and biotechnology enterprises, as well as from academic institutions, government agencies and private and public research institutions for our current product candidates. Our commercial opportunities will be reduced or eliminated if our competitors develop and commercialize products that are safer, more effective, have fewer side effects or are less expensive than any products that we may develop. Additionally, our commercial opportunities will be reduced or eliminated if novel upstream products or changes in treatment protocols reduce the overall incidence or prevalence of our current or future target diseases. Competition could result in reduced sales and pricing pressure on our product candidates, if approved by applicable regulatory authorities. In addition, significant delays in the development of our product candidates could allow our competitors to bring products to market before us and impair any ability to commercialize our product candidates.

There are currently no FDA- or EMA-approved products for the treatment of EBV+ PTLD. However, some marketed products and therapies are used off-label in the treatment of EBV+ PTLD, such as rituximab and combination chemotherapy regimens. In addition, a number of companies and academic institutions are developing product candidates for EBV+ PTLD and other EBV-driven diseases including: Viracta Therapeutics, Inc., which is conducting a pivotal, Phase 2 clinical study for nanatinostat (formerly named tractinostat, or VRx-3996) in combination with antiviral drug valganciclovir in relapsed/refractory EBV+ lymphomas; AlloVir (formerly known as ViraCyte), which has completed a Phase 2 clinical study for Viralym-M (ALVR105), an allogeneic, multi-virus T-cell product that targets six viruses in allogeneis HSCT recipients with ≥ 1 treatment-refractory infection, including EBV, and is conducting a pivotal study for Virus-Associated Hemorrhagic-Cystitis, as well as initiated a Phase 3 trial for the prevention of BKV, CMV, AdV, EBV, HHV06 and JCV in post-allogeneic HSCT patients; and Tessa Therapeutics Pte Ltd., is conducting a Phase 2 clinical study of its autologous CD30 CAR T in CD30+ classical Hodgkin Lymphoma and funding a Phase 1, investigator-sponsored study at Baylor College of Medicine, for its allogeneic CD30-CAR EBVST product candidate.

Competition in the MS market is high with at least 20 therapies, including four generics or bioequivalents, approved in the U.S. and EU for the treatment of various forms of MS, including clinically isolated syndrome, relapsing-remitting MS (RRMS), secondary progressive MS (SPMS) and primary progressive MS (PPMS). There are many competitors in the MS market, including major multi-national fully-integrated pharmaceutical companies and established biotechnology companies. Most recently, Ponvory (S1P modulator), marketed by Johnson & Johnson, and Kesimpta[®] (anti-CD20 monoclonal antibody), marketed by Novartis, were approved in the U.S. and/or EU for the treatment of relapsing forms of MS.

There are numerous development candidates in Phase 3 studies for both relapsing and/or progressive forms of MS and additional novel agents could be approved in either or both indications in the future including TG Therapeutics' anti-CD20 monoclonal antibody ublituximab (estimated PDUFA 09/2022), Merck KGaA's Bruton's tyrosine kinase (BTK) inhibitor, evobrutinib, Roche's BTK inhibitor, fenebrutinib, Sanofi's BTK inhibitor, tolebrutinib and AB Science's tyrosine kinase inhibitor, masitinib. Medicinova is planning to initiate a Phase 3 study of its PDE inhibitor, ibudilast (MN166) in non-active SPMS.

There are currently six autologous CAR T therapies approved in the U.S. and/or EU: Novartis' Kymrial® (tisagenlecleucel), Gilead/Kite's Yescarta® (axicabtagene ciloleucel) and TecartusTM (brexucabtagene autoleucel) and Bristol-Myers Squibb's Breyanzi® (lisocabtagene maraleucel) and Abecma (idecabtagene vicleucel) with 2seventy bio and Johnson & Johnson and Legend Biotech's CaryktiTM (ciltacabtagene autoleucel). There are many CAR-mediated cell therapies in development, and, although the majority are autologous, they include allogeneic and off-the-shelf cell therapies. There are multiple allogeneic CAR platforms being developed with differences in approaches to minimize instances of donor cells recognizing the patient's body as foreign or rejection of the donor cells by the patient's body. These approaches include the use of geneediting to remove or inhibit the TCR and the use of cell types without a TCR. The majority of clinical stage allogeneic CAR programs utilize alpha beta T cells as the cell type and gene editing of the T-cell receptor and HLA as the preferred technology approach, however, other strategies are also in development. It is possible that some of these other approaches will have more favorable characteristics than the approach we utilize, which would result in them being favored by potential partners or customers over our products. Depending on the diseases that we target in the future, we may face competition from both autologous and allogeneic CAR therapies and other modalities (e.g., small molecules, antibodies) in the indication of interest.

Many of the approved or commonly used drugs and therapies for our current or future target diseases, includingEBV+ PTLD and MS, are well established and are widely accepted by physicians, patients and third-party payors. Some of these drugs are branded and subject to patent protection, and other drugs and nutritional supplements are available on a generic basis. Insurers and other third-party payors may encourage the use of generic products or specific branded products. We expect that, if any of our product candidates are approved, they will be priced at a significant premium over competitive generic products. Absent differentiated and compelling clinical evidence, pricing premiums may impede the adoption of our products over currently approved or commonly used therapies, which may adversely impact our business. In addition, many companies are developing new therapeutics, and we cannot predict what the standard of care will become as our products continue in clinical development.

Many of our competitors or potential competitors have significantly greater established presence in the market, financial resources and expertise in research and development, manufacturing, preclinical testing, conducting clinical studies, obtaining regulatory approvals and marketing approved products than we do, and as a result may have a competitive advantage over us. Smaller or early-stage companies may also prove to be significant competitors, including through collaborative arrangements with large and established companies or if they are acquired by larger companies. These third parties compete with us in recruiting and retaining qualified scientific, commercial and management personnel, establishing clinical study sites and patient registration for clinical studies, as well as in acquiring technologies and technology licenses complementary to our programs or advantageous to our business.

As a result of these factors, these competitors may obtain regulatory approval of their products before we are able to obtain patent protection or other intellectual property rights, which will limit our ability to develop or commercialize our product candidates. Our competitors may also develop drugs that are safer, more effective, more widely used and cheaper than ours, and may also be more successful than us in manufacturing and marketing their products. These appreciable advantages could render our product candidates obsolete or noncompetitive before we can recover the expenses of development and commercialization.

We expect the product candidates we develop will be regulated as biological products (biologics) and therefore they may be subject to competition sooner than anticipated.

The Biologics Price Competition and Innovation Act of 2009 (BPCIA) was enacted as part of the Affordable Care Act to establish an abbreviated pathway for the approval of biosimilar and interchangeable biological products. The regulatory pathway establishes legal authority for the FDA to review and approve biosimilar biologics, including the possible designation of a biosimilar as "interchangeable" based on its similarity to an approved biologic. Under the BPCIA, an application for a biosimilar product cannot be approved by the FDA until 12 years after the reference product was approved under a BLA. During this 12-year period of exclusivity, another company may still market a competing version of the reference product if the FDA approves a full BLA for the competing product containing the sponsor's own preclinical data and data from adequate and well-controlled clinical trials to demonstrate the safety, purity and potency of their product. The law is complex and is still being interpreted and implemented by the FDA. As a result, its ultimate impact, implementation, and meaning are subject to uncertainty. While it is uncertain when processes

intended to implement BPCIA may be fully adopted by the FDA, any of these processes could have a material adverse effect on the future commercial prospects for our biological products.

We believe that any of the product candidates we develop that is approved in the U.S. as a biological product under a BLA should qualify for the 12-year period of exclusivity. However, there is a risk that this exclusivity could be shortened due to congressional action or otherwise, or that the FDA will not consider the subject product candidates to be reference products for competing products, potentially creating the opportunity for generic competition sooner than anticipated. Moreover, the extent to which a biosimilar, once approved, will be substituted for any one of the reference products in a way that is similar to traditional generic substitution for non-biological products is not yet clear, and will depend on a number of marketplace and regulatory factors that are still developing.

In addition, the approval of a biologic product biosimilar to one of our products could have a material adverse impact on our business as it may be significantly less costly to bring to market and may be priced significantly lower than our products.

If we are unable to establish sales and marketing capabilities or enter into agreements with third parties to market and sell our product candidates, we may be unable to generate any revenue from the sale of our products.

We are at an early stage of establishing an organization that will be responsible for the sale, marketing and distribution of pharmaceutical products and the cost of establishing and maintaining such an organization may exceed the cost-effectiveness of doing so. In order to market any products that may be approved by the FDA and comparable foreign regulatory authorities, we must build our sales, marketing, managerial and other non-technical capabilities or make arrangements with third parties to perform these services. There are significant risks involved in building and managing a sales organization, including our ability to hire, retain and incentivize qualified individuals, generate sufficient sales leads, provide adequate training to sales and marketing personnel and effectively manage a geographically dispersed sales and marketing team. Any failure or delay in the development of our internal sales, marketing and distribution capabilities, or entering into agreements with third parties to market and sell our products, would adversely impact the commercialization of these products. We may be competing with many companies that currently have extensive and well-funded sales and marketing operations. Without a sufficiently scaled, appropriately timed and trained internal commercial organization or the support of a third party to perform sales and marketing functions, we may be unable to compete successfully against these more established companies.

We may need to grow the size of our organization, and we may experience difficulties in managing this growth.

As of March 31, 2022, we had 570 employees, 136 of which were ATOM facility employees that transitioned to FDB at the time the Fujifilm Transaction closed on April 4, 2022. We may expand our organization in the future to support our continuing development activities and potential commercialization of our product candidates. In particular, we may need to add substantial numbers of additional personnel and other resources to support our development and potential commercialization of our product candidates. As our development and commercialization plans and strategies continue to develop, or as a result of any future acquisitions, our need for additional managerial, operational, sales, marketing, financial and other resources will increase. Our management, personnel and systems currently in place may not be adequate to support this future growth. Future growth would impose significant added responsibilities on members of management, including:

- managing our preclinical and clinical studies effectively;
- identifying, recruiting, maintaining, motivating and integrating additional employees, including the additional personnel needed to support continued development and potential commercialization of our product candidates;
- managing our internal development efforts effectively while complying with our contractual obligations to licensors, licensees, contractors and other third parties;
- improving our managerial, development, operational, information technology, and finance systems; and
- expanding our facilities.

As our operations expand, we will also need to manage additional relationships with various strategic partners, suppliers and other third parties. Our future financial performance and our ability to commercialize our product candidates and to compete effectively will depend, in part, on our ability to manage any future growth effectively. To that end, we must be able to manage our development efforts and preclinical and clinical studies effectively and hire, train and integrate additional management, research and development, manufacturing, administrative and sales and marketing personnel. Our failure to accomplish any of these tasks could prevent us from successfully growing our company.



Risks Related to Ownership of Our Common Stock

Our stock price has been and will likely continue to be volatile and may decline regardless of our operating performance.

Our stock price has fluctuated in the past and can be expected to be volatile in the future. From January 1, 2020 through March 31, 2022, the reported sale price of our common stock has fluctuated between \$4.52 and \$28.20 per share. The stock market in general and the market for biotechnology companies in particular have experienced extreme volatility that has often been unrelated to the operating performance of particular companies. In particular, the ongoing COVID-19 pandemic has further heightened the volatility of the stock market for biopharmaceutical companies. As a result of this volatility, investors may experience losses on their investment in our common stock. The market price of our common stock may be influenced by many factors, including the following:

- the success of competitive products or technologies;
- regulatory actions with respect to our product candidates or products or our competitors' product candidates or products;
- actual or anticipated changes in our growth rate relative to our competitors;
- announcements by us or our competitors of significant acquisitions, strategic partnerships, joint ventures, collaborations or capital commitments;
- results of clinical studies of our product candidates or those of our competitors;
- regulatory or legal developments in the U.S. and other countries;
- developments or disputes concerning patent applications, issued patents or other proprietary rights;
- the recruitment or departure of key personnel;
- the level of expenses related to any of our product candidates or clinical development programs;
- the results of our efforts to in-license or acquire additional product candidates or products;
- actual or anticipated changes in estimates as to financial results, development timelines or recommendations by securities analysts;
- variations in our financial results or those of companies that are perceived to be similar to us;
- fluctuations in the valuation of companies perceived by investors to be comparable to us;
- inconsistent or unusual trading volume levels of our shares or derivatives thereof;
- announcement or expectation of additional financing efforts;
- sales of our common stock by us, our insiders or our other stockholders;
- changes in the structure of healthcare payment systems;
- market conditions in the pharmaceutical and biotechnology sectors;
- general economic, industry and market conditions; and
- the other risks described in this "Risk Factors" section.

In addition, the stock markets in general, and the markets for biotechnology and pharmaceutical stocks in particular, have experienced significant volatility that has often been unrelated to the operating performance of particular companies, including in connection with the ongoing COVID-19 pandemic, which has resulted in decreased stock prices for many companies. For example, negative publicity regarding drug pricing and price increases by pharmaceutical companies has negatively impacted, and may continue to negatively impact, the markets for biotechnology and pharmaceutical stocks. Likewise, as a result of significant changes in U.S. social, political, regulatory and economic conditions or in laws and policies governing foreign trade and healthcare spending and delivery, including the possible repeal and/or replacement of all or portions of the Affordable Care Act or changes in tariffs and other restrictions on free trade stemming from U.S. and foreign government policies, or for other reasons, the financial markets for biotechnology and pharmaceutical stocks. These market fluctuations may adversely affect the trading price of our common stock.

In the past, class action litigation has often been instituted against companies whose securities have experienced periods of volatility in market price. Any such litigation brought against us could result in substantial costs and divert management's attention and resources, which could result in delays of our clinical studies or commercialization efforts.



Our principal stockholders own a significant percentage of our stock and will be able to exert significant control over matters subject to stockholder approval.

Our principal stockholders own a significant portion of our outstanding common stock. These stockholders may be able to determine the outcome of all matters requiring stockholder approval. For example, these stockholders may be able to control elections of directors, amendments of our organizational documents, or approval of any merger, sale of assets, or other major corporate transaction. This may prevent or discourage unsolicited acquisition proposals or offers for our common stock. The interests of our significant stockholders may not always coincide with the interests of other stockholders and they may act in a manner that advances their best interests and not necessarily those of other stockholders, including seeking a premium value for their common stock, and might affect the market price for our common stock.

Sales of a substantial number of shares of our common stock in the public market could cause our stock price to fall.

Sales of a substantial number of shares of our common stock in the public market could occur at any time. These sales, or the perception in the market that the holders of a large number of shares intend to sell shares, could reduce the market price of our common stock. Moreover, certain holders of shares of our common stock will have rights, subject to certain conditions, to require us to file registration statements covering their shares or to include their shares in registration statements that we may file for ourselves or other stockholders. We have registered and intend to continue to register all shares of common stock that we may issue under our equity compensation plans. Once we register these shares, they can be freely sold in the public market upon issuance, subject to volume limitations applicable to affiliates.

We have incurred and will continue to incur increased costs as a result of being a public company and our management expects to devote substantial time to public company compliance programs.

As a public company, we have incurred and will continue to incur significant legal, accounting and other expenses. We are subject to the reporting requirements of the Exchange Act, which require, among other things, that we file with the SEC annual, quarterly and current reports with respect to our business and financial condition. In addition, the Sarbanes-Oxley Act, as well as rules subsequently adopted by the SEC and The Nasdaq Stock Market to implement provisions of the Sarbanes-Oxley Act, impose significant requirements on public companies, including requiring establishment and maintenance of effective disclosure and financial controls and changes in corporate governance practices. Further, pursuant to the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, the SEC has adopted and will adopt additional rules and regulations, such as mandatory "say on pay" voting requirements, that now apply to us. Stockholder activism, the current political environment and the potential for future regulatory reform may lead to substantial new regulations and disclosure obligations, which may lead to additional compliance costs and impact the manner in which we operate our business in ways we cannot currently anticipate.

The rules and regulations applicable to public companies have substantially increased our legal and financial compliance costs and make some activities more timeconsuming and costly. To the extent these requirements divert the attention of our management and personnel from other business concerns, they could have a material adverse effect on our business, financial condition and results of operations. The increased costs will decrease our net income or increase our net loss and may require us to reduce costs in other areas of our business or increase the prices of our products or services.

Because we do not anticipate paying any cash dividends on our capital stock in the foreseeable future, capital appreciation, if any, will be the sole source of potential gain for our stockholders.

We have never declared or paid cash dividends on our capital stock. We currently intend to retain all of our future earnings, if any, to finance the growth and development of our business. In addition, the terms of any future debt agreements may preclude us from paying dividends. As a result, capital appreciation, if any, of our common stock will be the sole source of gain for our stockholders for the foreseeable future.

Future sales and issuances of our common stock or rights to purchase common stock, including pursuant to our equity incentive plans, could result in additional dilution of the percentage ownership of our stockholders and could cause our stock price to fall.

We expect that significant additional capital will be needed in the future to continue our planned operations. To raise capital, we may sell substantial amounts of common stock or securities convertible into or exchangeable for common stock in one or more transactions at prices and in a manner we determine from time to time. These future issuances of common stock or common stock-related securities, together with the exercise of outstanding options or warrants, and any additional shares issued in connection with acquisitions or in-licenses, if any, may result in material dilution to our investors. Such sales may also result in material dilution to our existing stockholders, and new investors could gain rights, preferences and privileges senior to those of holders of our common stock. To the extent equity valuations, including the trading price of our common stock, are depressed as a result of economic disruptions

and uncertainty concerning the COVID-19 pandemic or other factors, the potential magnitude of this dilution will increase.Pursuant to our equity incentive plans, our compensation committee is authorized to grant equity-based incentive awards to our employees, non-employee directors and consultants. Future grants of RSUs, options and other equity awards and issuances of common stock under our equity incentive plans will result in dilution and may have an adverse effect on the market price of our common stock.

Some terms of our charter documents and Delaware law may have anti-takeover effects that could discourage an acquisition of us by others, even if an acquisition would be beneficial to our stockholders and may prevent attempts by our stockholders to replace or remove our current management.

Our amended and restated certificate of incorporation (Certificate of Incorporation) and amended and restated bylaws (Bylaws), as well as Delaware law, could make it more difficult for a third party to acquire us or increase the cost of acquiring us, even if doing so would benefit our stockholders, or remove our current management. These include terms that:

- permit our board of directors to issue up to 20,000,000 shares of preferred stock, with any rights, preferences and privileges as they may designate;
- provide that all vacancies on our board of directors, including as a result of newly created directorships, may, except as otherwise required by law, be filled by
 the affirmative vote of a majority of directors then in office, even if less than a quorum;
- establish that our board of directors is divided into three classes, with each class serving three-year staggered terms, which makes it more difficult to replace a
 majority of our directors in a short period of time;
- require that any action to be taken by our stockholders must be effected at a duly called annual or special meeting of stockholders and not be taken by written consent;
- provide that stockholders seeking to present proposals before a meeting of stockholders or to nominate candidates for election as directors at a meeting of stockholders must provide advance notice in writing, and also specify requirements as to the form and content of a stockholder's notice;
- not provide for cumulative voting rights, thereby allowing the holders of a majority of the shares of common stock entitled to vote in any election of directors to elect all of the directors standing for election; and
- provide that special meetings of our stockholders may be called only by our board of directors, the chairperson of our board of directors or our chief executive officer.

Any of the factors listed above may frustrate or prevent any attempts by our stockholders to replace or remove our current management by making it more difficult for stockholders to replace members of our board of directors, who are responsible for appointing the members of our management.

In addition, because we are incorporated in Delaware, we are governed by Section 203 of the Delaware General Corporation Law, which may discourage, delay or prevent someone from acquiring us or merging with us whether or not it is desired by or beneficial to our stockholders. Under Delaware law, a corporation may not, in general, engage in a business combination with any holder of 15% or more of its capital stock unless the holder has held the stock for three years or, among other things, the board of directors has approved the transaction. Any term of our Certificate of Incorporation or Bylaws or Delaware law that has the effect of delaying or deterring a change in control could limit the opportunity for our stockholders to receive a premium for their shares of our common stock and could also affect the price that some investors are willing to pay for our common stock.

If securities or industry analysts do not publish research or publish inaccurate or unfavorable research about our business, our stock price and trading volume could decline.

The trading market for our common stock will depend in part on the research and reports that securities or industry analysts publish about us and our business. In the event securities or industry analysts who cover us downgrade our stock or publish unfavorable research about us or our business, our stock price would likely decline. If one or more of these analysts cease coverage of our company or fail to publish reports on us regularly, demand for our stock could decrease, which might cause our stock price and trading volume to decline.



General Risk Factors

Our future success depends on our ability to retain our executive officers and to attract, retain and motivate qualified personnel.

We are highly dependent upon our executive officers and other key employees and the loss of the services of any of our executive officers or other key employees, including scientific, technical or management personnel, could impede the achievement of our corporate objectives.

Our success depends on our ability to recruit, retain, manage and motivate our employees. Although we enter into employment agreements or offer letters with our employees, these documents provide for "at-will" employment, which means that any of our employees could leave our employment at any time, with or without notice. Competition for skilled personnel in our industry and geographic regions is intense and may limit our ability to hire and retain qualified personnel on acceptable terms or at all. To induce valuable employees to remain at our company, in addition to salary and cash incentives, we have provided equity awards that vest over time. The value to employees of equity awards may be significantly affected by movements in our stock price that are beyond our control and may at any time be insufficient to counteract more lucrative offers from other companies.

There can be no assurance that we will achieve all of the anticipated benefits of the Fujifilm Transaction and we could face unanticipated challenges.

We may not realize some or all of the anticipated benefits from the Fujifilm Transaction and we may encounter post-closing risks, including associated with the provision of (i) certain transition services to FDB by us and (ii) the provision of services to us by FDB pursuant to the Fujifilm MSA. We may experience increased difficulty and loss of institutional knowledge as a result of the transfer of ATOM Facility employees to FDB in connection with the Fujifilm Transaction which could harm our business. During the transition period, the Fujifilm Transaction will require significant time and resources from us which may disrupt our business and distract management from other responsibilities, which may result in losses on disposition or continued financial involvement in the ATOM Facility, including through indemnification or other financial arrangements, which could adversely affect our financial results.

Our relationships with customers and third-party payors will be subject to applicable anti-kickback, fraud and abuse, privacy and other laws and regulations, which could expose us to criminal sanctions, civil penalties, contractual damages, reputational harm and diminished profits and future earnings.

Healthcare providers, including physicians, and third-party payors will play a primary role in the recommendation and prescription of any product candidates for which we obtain regulatory approval. Our current and future arrangements with third-party payors and customers may expose us to broadly applicable fraud and abuse and other healthcare laws and regulations that may constrain the business or financial arrangements and relationships through which we conduct research and would market, sell and distribute our products. As a biopharmaceutical company, even though we do not and will not control referrals of healthcare services or bill directly to Medicare, Medicaid or other third-party payors, federal and state healthcare laws and regulations pertaining to fraud and abuse and patients' rights are and will be applicable to our business. If we obtain FDA approval of any of our product candidates and begin commercializing those products in the United States, our potential exposure under such laws will increase significantly, and our costs associated with compliance with such laws are also likely to increase. These laws may impact, among other things, our current activities with principal investigators and research patients, as well as proposed and future sales, marketing and education programs, distribution agreements, discounting, commission compensation, certain patient support offerings, and other business arrangements generally. In addition, the approval and commercialization of any of our product candidates outside the United States will also likely subject us to foreign equivalents of the healthcare laws mentioned here, among other foreign laws. Restrictions under applicable federal and state healthcare laws and regulations that may affect certain business arrangements and our ability to operate include, but are not limited to, the following:

- the federal Anti-Kickback Statute, a criminal law that governs, for example, our marketing practices, educational programs, pricing policies, and relationships with healthcare providers or other entities, by prohibiting, among other things, persons from knowingly and willfully soliciting, offering, receiving or providing remuneration, directly or indirectly, overtly or covertly, in cash or in kind, to induce or reward, or in return for, either the referral of an individual for, or the purchase, order or recommendation of, any good or service, for which payment may be made under a federal healthcare program such as Medicare and Medicaid;
- the FDCA and PHSA, which prohibit the misbranding and adulteration of biological products that are regulated as drugs, and which regulate the marketing of biological products;
- federal civil and criminal false claims laws, including the civil False Claims Act, which can be enforced through civil whistleblower or qui tam actions, and
 civil monetary penalty laws impose criminal and civil penalties against individuals or entities for knowingly presenting, or causing to be presented, to the
 federal government, including the Medicare and Medicaid programs, claims for payment or approval that are false or fraudulent or making a false statement to
 avoid, decrease or conceal an obligation to pay money to the federal government;

- provisions enacted under the federal Health Insurance Portability and Accountability Act of 1996 (HIPAA) impose criminal and civil liability for knowingly
 and willfully executing or attempting to execute, a scheme or artifice to defraud any healthcare benefit program and also impose criminal liability for, among
 other things, knowingly and willfully falsifying, concealing or covering up a material fact or making any materially false statements in connection with the
 delivery of or payment for healthcare benefits, items or services;
- HIPAA, as amended by HITECH also imposes obligations, including mandatory contractual terms, with respect to safeguarding the privacy, security and
 transmission of individually identifiable health information held by covered entities and their business associates and their subcontractors that use, disclose or
 otherwise process individually identifiable health information;
- the federal Physician Payments Sunshine Act, implemented as the Open Payments Program, which requires certain manufacturers of drugs, devices, biologics
 and medical supplies to report annually to CMS information related to payments and other transfers of value to physicians (defined to include doctors, dentists,
 optometrists, podiatrists and chiropractors), teaching hospitals, physician assistants, nurse practitioners, clinical nurse specialists, anesthesiologist assistants,
 certified registered nurse anesthetists and certified nurse midwives, as well as and ownership and investment interests held by physicians and their immediate
 family members and applicable group purchasing organizations;
- state and foreign laws and regulations that are analogous to, and may be broader in scope than, the federal laws and regulations described in this risk factor, such as state anti-kickback and false claims laws, may apply to sales or marketing or other arrangements and claims involving healthcare items or services reimbursed by non-governmental third-party payors, including private insurers; and
- state laws that require pharmaceutical companies to comply with the pharmaceutical industry's voluntary compliance guidelines and the relevant compliance
 guidance promulgated by the federal government; some state laws require drug manufacturers to report information regarding pricing and marketing
 information related to payments and other transfers of value to physicians and other healthcare providers as well as those that require the registration of
 pharmaceutical sales representatives; and some other state laws require the protection of the privacy and security of health information, which may differ from
 each other in significant ways and often are not preempted by HIPAA.

Efforts to ensure that our business arrangements with third parties comply with applicable healthcare laws and regulations will involve substantial costs. It is possible that governmental authorities will conclude that our business practices do not comply with current or future statutes, regulations or case law involving applicable healthcare laws and regulations. If our operations are found to be in violation of any of these laws or any other governmental regulations, we may be subject to significant civil, criminal and administrative penalties, damages, fines, imprisonment, exclusion from government-funded healthcare programs, such as Medicare and Medicaid, disgorgement, additional reporting requirements or oversight if we become subject to a corporate integrity agreement or similar agreement, and the curtailment or restructuring of our operations, reputational harm, contractual damages, and diminished profits and future earnings, any of which could adversely affect our ability to operate our business and our results of operations. If any physicians or other healthcare providers or entities with whom we expect to do business are found to not be in compliance with applicable laws, they may be subject to significant criminal, civil or administrative sanctions, including exclusions from government-funded healthcare programs.

Our employees may engage in misconduct or other improper activities, including noncompliance with regulatory standards and requirements, which could cause significant liability for us and harm our reputation.

We are exposed to the risk of employee fraud or other misconduct, including intentional failures to comply with FDA regulations or similar regulations of comparable foreign regulatory authorities, provide accurate information to the FDA or comparable foreign regulatory authorities, comply with manufacturing standards we have established, comply with federal and state healthcare fraud and abuse laws and regulations and similar laws and regulations established and enforced by comparable foreign regulatory authorities, report financial information or data accurately or disclose unauthorized activities to us. Employee misconduct could also involve the improper use of information obtained in the course of clinical studies, which could result in regulatory sanctions and serious harm to our reputation. It is not always possible to identify and deter employee misconduct, and the precautions we take to detect and prevent this activity may not be effective in controlling unknown or unmanaged risks or losses or in protecting us from governmental investigations or other actions or lawsuits stemming from a failure to be in compliance with such laws or regulations. If any such actions are instituted against us, and we are not successful in defending ourselves or asserting our rights, those actions could have a significant impact on our business and results of operations, including the imposition of significant fines or other sanctions.

Product liability lawsuits against us could cause us to incur substantial liabilities and to limit commercialization of any products that we may develop.

We face an inherent risk of product liability exposure related to the testing of our product candidates in human clinical studies and will face an even greater risk if we commercially sell any products that we may develop. Product liability claims may be brought against us by subjects enrolled in our clinical studies, patients, healthcare providers or others using, administering or selling our products. If we cannot successfully defend ourselves against claims that our product candidates or products caused injuries, we could incur substantial liabilities. Regardless of merit or eventual outcome, liability claims may result in:

- decreased demand for any product candidates or products that we may develop;
- clinical holds or termination of clinical study sites or entire study programs;
- injury to our reputation and significant negative media attention;
- withdrawal of clinical study participants;
- significant costs to defend the related litigation;
- substantial monetary awards to study subjects or patients;
- loss of revenue;
- · diversion of management and scientific resources from our business operations; and
- the inability to commercialize any products that we may develop.

We currently hold product liability insurance coverage at a level that we believe is customary for similarly situated companies and adequate to provide us with insurance coverage for foreseeable risks, but which may not be adequate to cover all liabilities that we may incur. We may not be able to maintain insurance coverage at a reasonable cost or in an amount adequate to satisfy any liability that may arise. We intend to expand our insurance coverage for products to include the sale of commercial products if we obtain regulatory approval for our product candidates in development, but we may be unable to obtain commercially reasonable product liability insurance for any products that receive regulatory approval. Large judgments have been awarded in class action lawsuits based on drugs that had unanticipated side effects. A successful product liability claim or series of claims brought against us, particularly if judgments exceed our insurance coverage, could decrease our cash and adversely affect our business.

If we and our third-party manufacturers fail to comply with environmental, health and safety laws and regulations, we could become subject to fines or penalties or incur costs that could have a material adverse effect on the success of our business.

We and our third-party manufacturers are subject to numerous environmental, health and safety laws and regulations, including those governing laboratory procedures and the handling, use, storage, treatment and disposal of hazardous materials and wastes. Our and our CMOs operations involve the use of hazardous and flammable materials, including chemicals and biological materials. Our and our CMOs operations also produce hazardous waste products. We generally contract with third parties for the disposal of these materials and wastes. We cannot eliminate the risk of contamination or injury from these materials. In the event of contamination or injury resulting from our or our thirdparty manufacturers' use of hazardous materials, we could be held liable for any resulting damages, and any liability could exceed our resources. We also could incur significant costs associated with civil or criminal fines and penalties.

Although we maintain workers' compensation insurance to cover us for costs and expenses we may incur due to injuries to our employees resulting from the use of hazardous materials with a policy limit that we believe is customary for similarly situated companies and adequate to provide us with insurance coverage for foreseeable risks, this insurance may not provide adequate coverage against potential liabilities. We do not maintain insurance for environmental liability or toxic tort claims that may be asserted against us in connection with our storage or disposal of biological or hazardous materials.

In addition, we may incur substantial costs in order to comply with current or future environmental, health and safety laws and regulations. These current or future laws and regulations may impair our research, development or production efforts. Failure to comply with these laws and regulations also may result in substantial fines, penalties or other sanctions, which could adversely affect our business, financial condition, results of operations and prospects.



The actual or perceived failure by us, our customers, or vendors to comply with increasingly stringent laws, regulations and contractual obligations relating to privacy, data protection, and data security could harm our reputation, and subject us to significant fines and liability.

We are or may become subject to numerous domestic and foreign laws and regulations regarding privacy, data protection, and data security, the scope of which is changing, subject to differing applications and interpretations and may be inconsistent among countries, or conflict with other rules. We are also subject to the terms of our contractual obligations to customers and third parties related to privacy, data protection, and data security. The actual or perceived failure by us, our customers, our vendors, or other relevant third parties to address or comply with these laws, regulations, and obligations could increase our compliance and operational costs, expose us to regulatory scrutiny, actions, fines and penalties, cause regulators to reject, limit or disrupt our clinical trial activities, result in reputational harm, lead to a loss of customers, reduce the use of our products, result in litigation and liability, and otherwise cause a material adverse effect on our business, financial condition, and results of operations.

For example, the EU adopted the General Data Protection Regulation (EU) 2016/679 (EU GDPR), in May 2018 which has direct effect in all EU member states and has extraterritorial effect where organizations outside of the EU *inter alia* process personal information of individuals in the EU in relation to the offering of goods or services to those individuals ("targeting test") or monitoring of their behavior ("monitoring test"). As such, the EU GDPR applies to us to the extent we are processing personal information in the context of an establishment in an EU Member State or we meet the requirements of either the targeting test or the monitoring test. The EU GDPR impose onerous and comprehensive privacy, data protection, and data security obligations onto controllers and processors, including, as applicable: (i) contractual privacy, data protection, and data security commitments, including the requirement to implement appropriate technical and organizational measures to safeguard personal information; (ii) establishing means for individuals to exercise their data protection requirements to supervisory authorities without undue delay (and no later than 72 hours where feasible) and/or concerned individuals; (vi) high standards for obtaining valid consent from data subjects; (vii) obligations to consider data protection as any new products or services are developed; and (viii) the provisions of more detailed privacy notices for clinical trial subjects and investigators. The EU GDPR also provides that EU member states may introduce further laws and regulations limiting the processing of genetic, biometric, or health data, which could limit our ability to collect, use and share EU data, cause our compliance costs to increase, require us to change our practices, adversely impact our business, and harm our financial condition.

The EU GDPR also restricts the transfer of personal information from the European Economic Area (EEA) to the United States and other countries that the European Commission does not recognize as having "adequate" data protection laws unless the parties to the transfer have implemented an appropriate data transfer mechanism in accordance with the EU GDPR. Data protection laws in the U.K. (as discussed below) and Switzerland impose similar restrictions. One of the primary mechanisms allowing U.S. companies to import personal information from the EU have been certification to the EU-U.S. Privacy Shield and Swiss-U.S. Privacy Shield frameworks. However, the EU-U.S. Privacy Shield and Court of Justice of the EU (CJEU) and subsequent guidance required additional compliance efforts to analyze international data flows and take steps to ensure adequate protections in addition to those provided under the Standard Contractual Clauses (or SCCs). Similarly, the Swiss-U.S. Privacy Shield Framework was declared as inadequate by the Swiss Federal Data Protection and Information from the EU have been released requiring additional compliance and implementation of the Laws been agreed to in principle by EU and US authorities, which promises to offer another mechanism for lawful international transfers of personal information from the EU have been released requiring additional compliance and implementation afforms for personal information for personal information transfers to the U.S. and other countries, we will face increased exposure to regulatory actions, substantial fines, and injunctions against processing or transfer information from the EU, and we may be required to increase demand for our products and services as our customers that are subject to the EU GDPR may seek alternatives that do not involve personal information transfers out of the EU.

Assisting our customers, partners, and vendors in complying with the EU GDPR, or complying with the EU GDPR ourselves, may cause us to incur substantial operational costs or require us to change our business practices. There may also be a risk that the measures will not be implemented correctly or that individuals within the business will not be fully compliant with the required procedures.

The EU GDPR imposes significant fines for serious non-compliance of up to the greater of $\in 20$ million or four percent of consolidated global turnover and restrictions or prohibitions on data processing, which could impair our ability to do business in the EU, reduce demand for our services and adversely impact our business and results of operations. The EU GDPR also confers a private

right of action on data subjects and consumer associations to lodge complaints with supervisory authorities, seek judicial remedies and obtain compensation for damages resulting from violations of the EU GDPR.

Further, following the United Kingdom's exit from the EU, known as Brexit, the EU GDPR's obligations continue to apply to the United Kingdom in substantially unvaried form under the so called "UK GDPR" by virtue of section 3 of the European Union (Withdrawal) Act 2018. The UK GDPR exists alongside the UK Data Protection Act 2018 which implements certain derogations in the UK GDPR into UK law. Under the UK GDPR, companies established in the United Kingdom and companies not established in the United Kingdom but who process personal information in relation to the offering of goods or services to individuals in the United Kingdom, or to monitor their behavior will be subject to the UK GDPR – the requirements of which are (at this time) largely aligned with those under the EU GDPR and as such, may lead to similar compliance and operational costs with potential fines of up to £17.5 million or 4% of global turnover. As a result, we are potentially exposed to two parallel data protection regimes, each of which authorizes fines and the potential for divergent enforcement actions. It should also be noted that the new EU SCCs do not automatically apply in the UK since Brexit, and the UK Government has not yet formally acknowledged the new EU SCCs, i.e., as a valid data transfer mechanism under the UK GDPR. Indeed, on 11 August 2021, the UK Information Commissioner's Office (ICO) launched a public consultation on its draft international data transfer agreement and guidance. This included the publication of a draft UK addendum that can be used with the new EU SCCs – however, this is yet to be finalized and as such, for the time being transfers from the UK to a third country should continue to be made in reliance on the "old" SCC.

Other countries outside of Europe continue to enact or are considering enacting similar cross-border data transfer restrictions and laws requiring local data residency, which could increase the cost and complexity of delivering our services and operating our business. For example, Brazil recently enacted the General Data Protection Law (Lei Geral de Proteção de Dados Pessoais or LGPD) (Law No. 13,709/2018), which broadly regulates the processing of personal information and imposes compliance obligations and penalties comparable to those of the EU GDPR.

Regulation of privacy, data protection and data security has also become more stringent in the United States. For example, the California Consumer Privacy Act (CCPA), which took effect on January 1, 2020, gives California residents expanded rights to access and delete their personal information, opt out of certain personal information sharing, and receive detailed information about how their personal information is used. The CCPA provides for civil penalties for violations, as well as a private right of action for data breaches that is expected to increase data breach litigation. The CCPA may increase our compliance costs and potential liability. The CCPA will be expanded substantially on January 1, 2023, when the California Privacy Rights Act of 2020 (CPRA) becomes fully operative. The CPRA will, among other things, give California residents the ability to limit use of certain sensitive personal information, further restrict the use of cross-contextual advertising, establish restrictions on the retention of personal information, expand the types of data breaches subject to the CCPA's private right of action, provide for increased penalties for CPRA violations concerning California residents the ability comprehensive privacy laws that will also go into effect in 2023, and several more are considering their own versions of privacy legislation, demonstrating a strong trend towards more stringent state privacy, data protection and data security legislation in the U.S., which could increase our potential liability and adversely affect our business.

Compliance with U.S. and foreign privacy, data protection, and data security laws and regulations could cause us to incur substantial costs or require us to change our business practices and compliance procedures in a manner adverse to our business. Moreover, complying with these various laws could require us to take on more onerous obligations in our contracts, restrict our ability to collect, use and disclose data, or in some cases, impact our ability to operate in certain jurisdictions. Failure to comply with U.S. and foreign privacy, data protection, and data security laws and regulations could result in government enforcement actions (which could include civil or criminal penalties), private litigation and/or adverse publicity and could negatively affect our operating results and business. Claims that we have violated individuals' privacy rights, failed to comply with privacy, data protection, and data security laws, or breached our contractual obligations, even if we are not found liable, could be expensive and time consuming to defend, could result in adverse publicity and could have a material adverse effect on our business, financial condition, and results of operations.

If our security measures are compromised, or our information technology systems or those of our vendors, and other relevant third parties fail or suffer security breaches, loss or leakage of data, and other disruptions, this could result in a material disruption of our services, compromise sensitive information related to our business, harm our reputation, trigger our breach notification obligations, prevent us from accessing critical information, and expose us to liability or other adverse effects to our business.

In the ordinary course of our business, we may collect, process, and store proprietary, confidential, and sensitive information, including personal information (including health information), intellectual property, trade secrets, and proprietary business information owned or controlled by ourselves or other parties. It is critical that we do so in a secure manner to maintain the confidentiality, integrity, and availability of such information. We face several risks relative to protecting this critical information,

including loss of access risk, inappropriate use or disclosure, inappropriate modification, and the risk of our being unable to adequately monitor, audit and modify our controls over our critical information. This risk extends to the third-party service providers who handle elements of our operations.

We, our partners, our CROs, our CMOs, and other business vendors on which we rely depend on information technology and telecommunication systems for significant elements of our operations, including, for example, systems handling human resources, financial reporting and controls, regulatory compliance and other infrastructure operations. Notwithstanding the implementation of security measures, given the size and complexity of our information technology systems and those of our third-party vendors and other contractors and consultants, and the increasing amounts of proprietary, confidential and sensitive information that they maintain, such information technology systems are potentially vulnerable to breakdown, service interruptions, system malfunction, natural disasters, terrorism, war and telecommunication and electrical failures, as well as security breaches from inadvertent or intentional actions by our personnel, third-party vendors, contractors, consultants, business partners, and/or other third parties, or from cyber-attacks by malicious third parties (including the deployment of harmful malware, ransomware, denial-of-service attacks, social engineering, and other means to affect service reliability and threaten the confidentiality, integrity, and availability of information), which may compromise our system infrastructure, or that of our third-party vendors and other contractors and consultants, or lead to data leakage. The risk of a security breach or disruption, particularly through accidental actions or omissions by trusted insiders, cyber-attacks or cyber intrusions, including by computer hackers, viruses, foreign governments, and cyber terrorists, has generally increased as the number, intensity, and sophistication of attempted attacks and intrusions from around the world have increased. Additionally, the increased usage of computers operated on home networks due to the shelter-in-place or similar restrictions related to the COVID-19 pandemic may make our systems more susceptible to security breaches. For example, in March 2021, MSK provided notice that MSK was one of many customers impacted by a data breach at Accellion, Inc., which provides a document-sharing system. MSK subsequently notified us that certain documents related to one of our discontinued programs were subject to the breach, which compromise we deemed immaterial. Although we take measures to protect sensitive data from unauthorized access, use or disclosure, we and our third party service providers frequently defend against and respond to cyber attacks, and our information technology and infrastructure may be vulnerable to attacks by hackers or viruses or breached due to personnel error, malfeasance, or other malicious or inadvertent disruptions. Any such breach or interruption could compromise our networks and the information stored there could be accessed by unauthorized parties, manipulated, publicly disclosed, lost, or stolen.

Failures or significant downtime of our information technology or telecommunication systems or those used by our third-party service providers could cause significant interruptions to our operations, including preventing us from conducting tests or research and development activities and preventing us from managing the administrative aspects of our business. For example, the loss of clinical study data from completed, ongoing or planned clinical studies could result in delays in our regulatory approval efforts and significantly increase our costs to recover or reproduce the data. In addition, sophisticated operating system software and applications that we procure from third parties may contain defects in design or manufacture, including vulnerabilities, "bugs" and other problems that could unexpectedly interfere with the operation of our networks, system, or our processing of personal information or other data. To the extent that any disruption or security breach results in a loss of or damage to our data or applications, or inappropriate disclosure of confidential or proprietary information, we could incur liability, the further development of our product candidates could be delayed and our business could be otherwise adversely affected.

We may not be able to anticipate all types of security threats, and we may not be able to implement preventative measures effective against all such security threats. We also may not be effective in responding to, containing or mitigating the risks of an attack. The techniques used by cyber criminals change frequently, may not be recognized until launched, and can originate from a wide variety of sources, including outside groups such as external service providers, organized crime affiliates, terrorist organizations, hostile foreign governments or agencies, or cybersecurity researchers. To the extent that any disruption or security breach were to result in a loss of, or damage to, our data or applications, or those of our third-party vendors and other contractors and consultants, or inappropriate disclosure of confidential or proprietary information, we could incur liability and reputational damage and the further development and commercialization of our products and services could be delayed.

The costs related to significant security breaches or disruptions could be material and could exceed the limits of the cybersecurity insurance we maintain, if any, against such risks. If the information technology systems of our third-party vendors and other contractors and consultants become subject to disruptions or security breaches, we may have insufficient recourse against such third parties and may have to expend significant resources to mitigate the impact of such an event, and to develop and implement protections to prevent future events of this nature from occurring.

We cannot assure you that our data protection efforts and our investment in information technology will prevent significant breakdowns, data leakages, breaches in our systems, or those of our third-party vendors and other contractors and consultants, or other cyber incidents that could have a material adverse effect upon our reputation, business, operations, or financial condition. For example, if such an event were to occur and cause interruptions in our operations, or those of our third-party vendors and other contractors and consultants, it could result in a material disruption of our programs and the development of our services and

technologies could be delayed. Furthermore, significant disruptions of our internal information technology systems or those of our third-party vendors and other contractors and consultants, or security breaches could result in the loss, misappropriation, and/or unauthorized access, use, or disclosure of, or the prevention of access to, confidential information (including trade secrets or other intellectual property, proprietary business information, and personal information), which could result in financial, legal, business, and reputational harm to us. For example, any such event that leads to unauthorized access, use, or disclosure of personal information, including personal information regarding our customers or employees, could harm our reputation directly, compel us to comply with federal and/or state breach notification laws and foreign law equivalents, subject us to mandatory corrective action, and otherwise subject us to liability under laws and regulations that protect the privacy and security of personal information, which could result in significant legal and financial exposure and reputational damages that could potentially have an adverse effect on our business.

Although we take measures to protect sensitive data from unauthorized access, use or disclosure, our information technology and infrastructure may be vulnerable to attacks by hackers or viruses or breached due to personnel error, malfeasance, or other malicious or inadvertent disruptions. Any such breach or interruption could compromise our networks and the information stored there could be accessed by unauthorized parties, manipulated, publicly disclosed, lost, or stolen.

Any such access, breach, or other loss of information could result in legal claims or proceedings, liability under domestic or foreign privacy, data protection and data security laws such as HIPAA and HITECH, and penalties. Notice of certain security breaches must be made to affected individuals, the Secretary of HHS, and for extensive breaches, notice may need to be made to the media or state attorneys general. Such notice could harm our reputation and our ability to compete. Although we have implemented security measures, such data is currently accessible through multiple channels, and there is no guarantee we can protect our data from breach. Unauthorized access, loss or dissemination could also damage our reputation or disrupt our operations, including our ability to conduct our analyses, conduct research and development activities, collect, process and prepare company financial information, and manage the administrative aspects of our business.

Penalties for violations of these laws vary. For instance, penalties for failure to comply with a requirement of HIPAA and HITECH vary significantly, and include significant civil monetary penalties and, in certain circumstances, criminal penalties with fines up to \$250,000 per violation and/or imprisonment. A person who knowingly obtains or discloses individually identifiable health information in violation of HIPAA may face a criminal penalty of up to \$50,000 and up to one-year imprisonment. The criminal penalties increase if the wrongful conduct involves false pretenses or the intent to sell, transfer or use identifiable health information for commercial advantage, personal gain or malicious harm.

Further, various states, such as California and Massachusetts, have implemented similar privacy laws and regulations, such as the California Confidentiality of Medical Information Act, that impose restrictive requirements regulating the use and disclosure of health information and other personally identifiable information. These laws and regulations are not necessarily preempted by HIPAA, particularly if a state afford greater protection to individuals than HIPAA. Where state laws are more protective, we have to comply with the stricter provisions. In addition to fines and penalties imposed upon violators, some of these state laws also afford private rights of action to individuals who believe their personal information has been misused. California's patient privacy laws, for example, provide for penalties of up to \$250,000 and permit injured parties to sue for damages. Similarly, the CCPA allows consumers a private right of action when certain personal information is subject to unauthorized access and exfiltration, theft or disclosure due to a business' failure to implement and maintain reasonable security procedures. The interplay of federal and state laws may be subject to varying interpretations by courts and government agencies, creating complex compliance issues for us and data we receive, use and share, potentially exposing us to additional expense, adverse publicity and liability. Further, as regulatory focus on privacy issues continues to increase and laws and regulations concerning the protection of personal information expand and become more complex, these potential risks to our business could intensify. Changes in laws or regulations associated with the enhanced protection of certain types of sensitive data, for the treatment of genetic data, along with increased customer demands for enhanced data security infrastructure, could greatly increase our cost of providing our products, decrease demand for our products, reduce our revenues and/or subject us to additional liabilities.

Changes in tax laws or regulations that are applied adversely to us or our customers may have an adverse effect on our business, cash flows, financial condition or results of operations.

We are subject to income and non-income based taxes in the U.S. and various jurisdictions outside the U.S. Our business and financial condition could be adversely affected by changes in federal, state, local or international tax laws, changes in taxing jurisdictions' administrative interpretations, decisions, policies and positions, changes in accounting principles, applicability of withholding taxes, and changes to our business operations. For example, US legislations such as the Tax Act, the Coronavirus Aid, Relief, and Economic Security Act (CARES Act), and the American Rescue Act, made significant changes to the corporate tax rate, the potential realization of net deferred tax assets relating to our operations, taxation of foreign earnings, and deductibility of expenses, and could have a material impact on our financial position or results of operations.



Our ability to use net operating loss carryforwards and certain tax assets to offset future taxable income or taxes may be subject to certain limitations.

Our ability to use our federal and state net operating losses (NOLs) and certain other tax attributes to offset potential future taxable income and related income taxes that would otherwise be due is dependent upon our generation of future taxable income, and we cannot predict with certainty when, or whether, we will generate sufficient taxable income to use all of our NOLs or other tax attributes.

As of December 31, 2021, we had significant U.S. federal and state NOLs due to prior period losses. Under the Tax Cuts and Jobs Act (the Tax Act), federal NOLs generated in tax years beginning on or after January 1, 2018 may be carried forward indefinitely, but the utilization of such federal NOLs is limited to 80% of current year taxable income. The CARES Act temporarily suspends this 80% taxable income limitation, allowing an NOL carryforward to fully offset taxable income in tax years beginning before January 1, 2021. It is uncertain if, and to what extent, various states will conform to the Tax Act or the CARES Act. Neither the Tax Act nor the CARES Act had a material impact to our financial statements.

In addition, under Section 382 of the Internal Revenue Code of 1986, as amended (the Code), our ability to utilize these NOLs and other tax attributes, such as federal tax credits, in any taxable year may be limited if we have experienced an "ownership change". Generally, a Section 382 ownership change occurs if one or more stockholders or groups of stockholders who owns at least 5% of a corporation's stock increases its ownership by more than 50 percentage points over its lowest ownership percentage within a three-year testing period. Similar rules may apply under state tax laws. We performed a Section 382 analysis of transactions in our stock through December 31, 2021 and concluded that we have experienced ownership changes since inception that we believe under Section 382 of the Code will result in limitations on our ability to use certain pre-change NOLs and credits. In addition, we may experience subsequent ownership changes as a result of future equity offerings or other changes in the ownership of our stock, some of which are beyond our control. As a result, the amount of the NOLs and tax credit carryforwards presented in our financial statements could be limited and, in the case of NOLs generated before January 1, 2018 may expire unused. Any such material limitation or expiration of our NOLs may harm our future operating results by effectively increasing our future tax obligations. Similar provisions of state tax law may also apply to limit the use of accumulated state tax attributes. Regulatory changes, such as suspensions on the use of NOLs, or other unforeseen reasons, may cause our existing tax attributes to expire, decrease in value or otherwise be unavailable to offset future income tax liabilities.

Business disruptions could seriously harm our future revenues and financial condition and increase our costs and expenses.

Our operations could be subject to earthquakes, power shortages, telecommunications failures, water shortages, floods, hurricanes, typhoons, fires, extreme weather conditions, medical epidemics and other natural or man-made disasters or business interruptions, for which we are predominantly self-insured. Two of our corporate locations are located in California, an area prone to earthquakes and fires. The occurrence of any of these business disruptions could seriously harm our operations and financial condition and increase our costs and expenses. We rely on third-party manufacturers to produce our product candidates. Our ability to obtain clinical supplies of product candidates could be disrupted, if the operations of these suppliers are affected by a man-made or natural disaster or other business interruption, including, for example, the ongoing COVID-19 pandemic.

Item 6. Exhibits

Exhibit		Incorporated by Reference				Filed
No.	Description of Exhibit	Form	File No.	Exhibit	Filing Date	Herewith
2.1+	Asset Purchase Agreement, dated as of January 26, 2022, by and between Atara Biotherapeutics, Inc., FUJIFILM Diosynth Biotechnologies California, Inc., and for certain limited purposes, FUJIFILM Holdings America Corporation	8-K	001-36548	2.1	4/4/2022	
3.1	Amended and Restated Certificate of Incorporation of Atara Biotherapeutics, Inc.	S-1	333-196936	3.2	6/20/2014	
3.2	Amended and Restated Bylaws of Atara Biotherapeutics, Inc.	S-1	333-196936	3.4	6/20/2014	
10.1+	Master Services and Supply Agreement dated as of January 26, 2022 by and between Atara Biotherapeutics, Inc., and FUJIFILM Diosynth Biotechnologies California, Inc.					х
31.1	Certification by Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002					Х
31.2	Certification by Principal Financial and Accounting Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002					Х
32.1(1)	<u>Certifications of Chief Executive Officer and Principal Financial and</u> <u>Accounting Officer pursuant to 18 U.S.C Section 1350 as adopted</u> <u>pursuant to Section 906 of The Sarbanes-Oxley Act of 2002</u>					Х
101.INS	Inline XBRL Instance Document – the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document					Х
101.SCH	Inline XBRL Taxonomy Extension Schema Document					Х
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document					Х
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document					Х
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document					Х
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.					Х
104	The cover page from the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2022, formatted in Inline XBRL.					Х

+ Portions of this exhibit have been omitted as being both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

(1) The certifications attached as Exhibit 32.1 accompany this Quarterly Report on Form 10-Q pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, and shall not be deemed "filed" by the Registrant for purposes of Section 18 of the Securities Exchange Act of 1934, as amended.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, Atara Biotherapeutics, Inc. has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: May 5, 2022

ATARA BIOTHERAPEUTICS, INC.

By: /s/ Pascal Touchon

Pascal Touchon President and Chief Executive Officer (Duly Authorized Officer and Principal Executive Officer)

By: /s/ Utpal Koppikar

Utpal Koppikar Chief Financial Officer (Duly Authorized Officer and Principal Financial and Accounting Officer)

73

[***] = Certain information contained in this document, marked by brackets, has been omitted because it is both not material and would be competitively harmful if publicly disclosed.

Exhibit 10.1



Diesynth biotechnologies



(1) FUJIFILM DIOSYNTH BIOTECHNOLOGIES CALIFORNIA, INC.

AND

(2) ATARA BIOTHERAPEUTICS, INC.

Partners for *Life*

1.	DEFINITIONS AND INTERPRETATION	2
2.	FRAMEWORK AND SCOPE OF SERVICES	12
3.	TERM	12
4.	PERFORMANCE OF PROGRAM	13
5.	QUALITY AND REGULATORY MATTERS	16
6.	CONFORMING BATCHES AND NON-CONFORMING BATCHES	17
7.	DELIVERY, TITLE, RISK, STORAGE AND INVENTORY	20
8.	PRICE AND PAYMENT	21
9.	WARRANTIES	23
10.	INDEMNIFICATION & LIMITATION OF LIABILITY	25
11.	INTELLECTUAL PROPERTY	28
12.	CONFIDENTIALITY	29
13.	CHANGE	32
14.	DELAY, CANCELLATION, TERMINATION AND CONSEQUENCES	33
15.	FORCE MAJEURE	37
16.	DISPUTE RESOLUTION	38
17.	AUDIT	39
18.	NOTICES	40
19.	EXPORT/IMPORT CONTROLS AND SANCTIONS COMPLIANCE	40
20.	MODERN SLAVERY AND CORRUPTION	41
21.	ASSIGNMENT AND SUB-CONTRACTING	42
22.	GENERAL	43
23.	GOVERNING LAW	44
Schedule 1	- Charges	1
Schedule 2	- Forecasting and Orders	2
Schedule 3	- Addresses for Notice	3
Schedule 4	- Atara Deliverables	4
Schedule 5	- Statement of Work Template	5
		1

THIS AGREEMENT IS BETWEEN

- (1) **FUJIFILM DIOSYNTH BIOTECHNOLOGIES CALIFORNIA, INC.,** incorporated and registered in Delaware whose principal place of business is at 200 Summit Lake Drive, Valhalla, NY 10595 ("Fujifilm"); and
- (2) ATARA BIOTHERAPEUTICS, IN Corporated and registered in Delaware with a principal place of business office at 611 Gateway Boulevard, Suite 900, South San Francisco, CA 94080, USA ("Atara").

WHEREAS

- (A) Fujifilm is a biopharmaceutical contract development and manufacturing organization and, Atara and Fujifilm are agreeing to enter into this Agreement under which Fujifilm is appointed to carry out agreed clinical and commercial services for Atara in relation to certain of Atara's Products, including manufacturing, testing, storage, AL&P, new product introductions and supply of Products to Atara, as more specifically set forth herein.
- (B) Capitalized terms used in these Recitals are defined below.

NOW, THEREFORE, in consideration of the respective covenants, representations, warranties and agreements set forth herein, the parties hereto agree as follows:

AGREED TERMS

1. DEFINITIONS AND INTERPRETATION

1.1 In this Agreement the following words have the following meanings unless inconsistent with the context:

"Affiliate"	means, with respect to any Person, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, such first Person; "control," including the terms "controlled by" and "under common control with," means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, as trustee or executor, as general partner or managing member, by contract or otherwise;
"AL&P"	assembly, secondary packaging, labeling, Fujifilm disposition of Finished Drug Product and preparations for shipment pick up;
"Applicable Laws"	all applicable and binding ordinances, rules, regulations, laws, requirements, guidances and court orders of any kind whatsoever which apply (i) in the case of Fujifilm: to the provision of the Fujifilm Services in the jurisdiction in which a Program is being performed, including all applicable cGMP; and (ii) in the case of Atara: to the provision and receipt of the Fujifilm Services and that relate to the treatment/use/consumption of the Product in the applicable jurisdiction (including storage, import, export, transport, marketing, promotion, processing, packaging, distribution, sale and/or use), as amended from time to time including all applicable cGMP;

"APA"	means the Asset Purchase Agreement between Atara and Fujifilm dated January 26, 2022.
"Atara Deliverable(s)"	one or more of the deliverables set forth on Schedule 4 to be supplied by Atara to Fujifilm for the performance of a Program and identified in its corresponding SOW;
"Atara Release"	Atara quality's review of the Batch Documentation and final approval to release of the Product, following Fujifilm Disposition, in accordance with the Quality Agreement;
"Atara-Supplied Consumable(s)"	the Product-specific Consumable(s) agreed by Atara to be supplied to Fujifilm for the performance of a Program following a Consumables Delay in such Program pursuant to clause 14.1.2;
"Authorized Third Party"	has the meaning given to it in clause 12.112.1.3;
"Background IP"	all Intellectual Property Rights owned or controlled by a party, whether solely or jointly, prior to the Effective Date or arising after the Effective Date independently of the Fujifilm Services performed under this Agreement and without reliance upon, use of or reference to any Confidential Information or Background IP of the other party;
"Batch"	a specific quantity of a given Product that is intended to be of uniform character and quality, within specified limits, and is produced under cGMP conditions during the same cycle of manufacture and from the same donor material as defined by the applicable QA Documents. For clarity and, where applicable, Batches obtained from the same donor material but from different cycles of manufacture constitute separate (i.e. not the same) Batches;
"Batch Documentation"	has the meaning given to it in clause 6.4;
"Batch Fee"	the agreed Batch Fee set out in Part 3 of Schedule 1;

"Business Day"	in relation to notices given under this Agreement (including any Scope of Work) a day other than a Saturday, Sunday or any day on which the commercial banks located in Thousand Oaks or San Francisco, California, USA, are authorized or obligated by law to be closed;
"Calendar Year"	a period of twelve (12) consecutive calendar months corresponding to the calendar year commencing on the first day of January and on the last day of December, except that the first Calendar Year of this Agreement shall commence on the Effective Date and the final Calendar Year shall end on the actual date of expiry or termination of this Agreement (which may not be 31 st December of the final Calendar Year);
"Certificate of Analysis"	means a document signed by an authorized representative of Fujifilm, describing Product Specifications for, and testing methods applied to, the Product, and the results of such testing;
"Certificate of Compliance"	means a document signed by an authorized representative of Fujifilm, certifying that a particular Batch meets the Product Specifications and was manufactured in accordance with cGMP and the Process including an assessment of Deviations raised during the manufacture;
"cGMP"	means all current good manufacturing practice applicable to the manufacture of Product(s) that are promulgated by the applicable Regulatory Authority, as may be amended from time to time, and as are applicable to the Product(s), including: Current Good Manufacturing Practice as defined in (i) those sections applicable within the FDA Regulations 21 CFR Part 210, 211, 600, 601 and 610 and (ii) the rules governing medicinal products in the European Union and the United Kingdom. Eudralex Volume 4 – Guidelines for good manufacturing practices for medicinal products for human and veterinary use. Part I – Basic Requirements for Medicinal Products. Part II – Basic Requirements for Active Substances used as Starting Materials and ICHQ7;
"cGMP Batch"	a Batch identified in a Scope of Work which is intended to be manufactured during a manufacturing stage and subject to Fujifilm Disposition in each case in accordance with cGMP;
"Change"	has the meaning given to it in clause 13;

"Charges"	has the meaning set out in clause8.1.1;
"Commercially Reasonable Efforts"	with respect to the activities within the Fujifilm Services, the carrying out of such activities using efforts and resources consistent with the efforts and resources used by a biopharmaceutical contract and development manufacturing organization of similar size and financial resources in the same or similar circumstances;
"Competitor"	means: (i) [***] contract development and manufacturing organizations [***]; and (ii) [***] cell therapy contract development and manufacturing organizations [***], but in each case, shall exclude [***], in each case, [***] as reasonably and mutually agreed by the parties;
"Confidential Information"	the terms of this Agreement, any Scope of Work, and all information (in whatever form) in respect of the business of each of party and its Affiliates including any ideas, business methods, finance, prices, business, financial, marketing or development plans, products or services, know-how or other matters connected with products or services manufactured and/or marketed, customer lists or details, computer systems and software, which is (in each case) provided to, or obtained by, one party from the other; for clarity, Confidential Information of Atara includes Atara Deliverables, Atara Background IP and Atara Foreground IP; and all information of third parties that Atara has an obligation to keep confidential. Confidential Information of Fujifilm includes Fujifilm Background IP, Fujifilm Foreground IP and all information of third parties that Fujifilm has an obligation to keep confidential;
"Conforming Batch"	a cGMP Batch which has been produced in accordance with cGMP, the Process and which meets the Product Specification;
"Consumable"	a consumable item used or intended for use in the performance of a Program other than Atara-Supplied Consumables or Atara Deliverables, including reagents (including analytical reagents), raw materials , packaging components, chromatography resins, filters, filtration membranes, media, buffer bags, refold bags, tubing, hoses, disposable analytical test kits, in-process measurement probes, columns (including analytical columns) and disposable containers;

"Consumables Delay"	has the meaning given to it in clause 14.1.2;
"Atara Foreground IP"	all Foreground IP that [***];
"[***]"	[***];
"Commercial License"	has the meaning given to it in the definition of "Regulatory Application" in this clause 1;
"Delay"	has the meaning given to it in clause 14.1.1;
"Demonstration Batch"	a non-cGMP Batch which is manufactured for demonstration purposes and which is not intended for human use;
"Deviation"	a cGMP deviation as detailed in the Quality Agreement;
"Effective Date"	as mutually agreed by the parties in writing;
"Engineering Batch"	a Batch that is manufactured in a cGMP Facility at scale using the Process but which is not intended for human use;
"Existing Products"	Tab-cel [®] , ATA188, ATA3219, ATA3271, BLCLs and PBMCs, provided that, whilst EBV reagent, CAR vectors, BLCLs and PBMCs are intermediates, they are treated as their own "Product" for the purposes of this Agreement;
"Facility"	Fujifilm's manufacturing facility located at 2430 Conejo Spectrum Street, Thousand Oaks, CA 91320 or such other Fujifilm facility specifically identified and agreed to by Atara in a Scope of Work;
"Foreground IP"	all Intellectual Property Rights conceived, developed, generated or reduced to practice during the Term either [***];
"Fujifilm Disposition"	the Stage during which (i) the Product is tested for compliance versus the Product Specification; (ii) all production instruction and analytical records relating to cGMP manufacture of each cGMP Batch prepared by Fujifilm are reviewed; and (iii) a Fujifilm recommendation for Product release or reject is made; in each case as applicable;
"Fujifilm Foreground IP"	all Foreground IP that (a) is not Atara Foreground IP and (b) is generally applicable to the manufacturing or analysis of pharmaceutical products;
"Fujifilm Expression Technology"	the Background IP of Fujifilm that is its proprietary manufacturing, expression or purification technologies, including:
	(a)an expression system within the scope of international patent application PCT/GB2007/000351 (the "pAVEway™ Expression System");
	(b)expression technology within the scope of international patent application PCT/GB2014/000165 (the "Apollo™ Expression Technology");
	(c)Fujifilm's proprietary alcohol oxidase yeast Pichia pastoris expression system (the "Yeast Expression System");
	(d)Fujifilm's proprietary HEK293 cell lines; and
	(e)Fujifilm's proprietary Vero cell lines;
	6

the services described in the applicable Scope of Work executed by the parties, that may include:

(a)manufacturing and testing (manufacturing/processing, in process and release testing, environmental monitoring, material management);

(b)storage (including, storage, material management, including receipt and issuance to manufacturing, QC testing or for AL&P and shipment;

(c)AL&P;

- (d)new production introductions (tech transfer, engineering and project management oversight, equipment qualification and validation, batch record and SOP generation, training, operator qualification, process validation); and
- (e)additional support and testing (additional release testing, stability testing, sample management, investigational testing, method support and transfer, NC/CAPA/CC investigation and coordination);

a conscious and voluntary disregard of the need to use reasonable care, which is reasonably likely to cause foreseeable grave injury or harm to persons, property, or both;

"Gross Negligence"

"Indemnify"	to indemnify, defend and hold harmless;
"Intellectual Property Right"	any intellectual property rights and interests including patents, utility models, designs, design rights, copyright (including rights in software), decryption rights, database rights, trade marks, rights pursuant to passing off, service marks, business and trade names, domain names, know-how, results, data, databases, formulations, compounds, rights in biological or chemical materials, rights under data exclusivity laws, rights under unfair competition laws, topography rights, inventions, rights in confidential information (including technical and commercial trade secrets); supplementary protection certificates and image rights, and rights of a similar or corresponding character in any part of the world, in each case whether registered or not and including any application for registration and renewals or extensions of such rights in any country in the world and whether subsisting now or in the future;
"Inventory"	has the meaning given to it in clause 7.1.2;
[***]	[***]
"Liabilities"	any (i) liabilities of any nature, whether accrued, absolute, contingent or otherwise and whether in contract, tort (including negligence) or otherwise; (ii) losses, costs, damages, fines or expenses (including reasonable legal fees); and (iii) claim, demand, proceeding, action or cause of action including those by third parties; in each case howsoever arising. "Liability" shall be construed accordingly;
"Material Review Board" or "MRB"	a cross-functional committee established pursuant to the Quality Agreement, led by quality assurance, that consists of representation from program management, quality control (as applicable), and manufacturing, for clarity, Atara representation is required on any such committee;
"Modifications"	a modification to a Facility; or equipment (including Process specific qualification and installation of existing equipment), required in order to perform a Process and detailed in the applicable Scope of Work (as applicable);
"Non-Conforming Batch"	a cGMP Batch which has not been produced in accordance with cGMP and/or does not meet the Product Specification [***];
	8

"Order"	has the meaning given to it in Schedule 2;
"Pass Through Costs"	has the meaning given to it in Part 3 of Schedule 1;
"Pass Through Related Services"	ad hoc services (not otherwise covered by the Charges) to be performed by a Third Party, which may include: Subcontracted Work (not otherwise covered by the Charges); and Process Specific Equipment; undertaking of Modifications and disposal of Special Waste by Fujifilm for a Program, in each case, as specifically identified in a SoW;
"Pass Through Services Costs"	has the meaning given to it in Part 3 of Schedule 1;
"Person"	means an individual, corporation, partnership, limited liability company, limited liability partnership, syndicate, person, trust, association, organization or other entity, including any governmental authority, and including any successor, by merger or otherwise, of any of the foregoing.
"Process"	the particular process used, or to be used (including all steps and activities) in the Fujifilm Services, including manufacture of the Product, as set forth in the Process Specification and the Quality Agreement;
"Process Specification"	the Process operating parameters and specifications as documented in (i) pre- successful grant of a Commercial License: the QA Documents which have been agreed by the parties for cGMP Batch production and (ii) following grant of a Commercial License: the QA Documents which have been agreed by the parties for cGMP Batch production and the applicable Commercial License (for example the Biologics License Application or a Marketing Authorization Application);
"Processing Room Fees"	has the meaning given to it in Part 1 of Schedule Error! Reference source not found.;
"Process-Specific Equipment"	an item of equipment set forth in a SoW, which (a) the parties mutually agree is required by Fujifilm to operate the Process and is specific to the Process; and (b) as of the SoW Effective Date, is not already used by Fujifilm in its Facilities or otherwise for other customer(s) of Fujifilm;
"Product"	the particular product or substance (cells, compound, reagent, intermediate or molecule) created during and as a result of performing the Process, including the Existing Products. The specific Product of a Program will be identified in the applicable Scope of Work for such Program;

"Product Specification"	the product specification for Product which is documented in a QA Document;
"Program"	a program of Fujifilm Services for the Product, as more specifically set out in an SOW, to be carried out by Fujifilm in accordance with the terms of this Agreement;
"Program Manager"	on a Program-by-Program basis, the manager appointed by each of Fujifilm and Atara under the applicable SoW for such Program;
"QA Documents"	the documents produced and approved in accordance with the Quality Agreement or any cGMP documents agreed by the parties in writing;
"Quality Agreement"	the quality agreement entered into by the parties (as may be amended from time to time) containing quality assurance provisions for the manufacture of Product to be supplied under this Agreement;
"[***]"	[***];
"Regulatory Application"	means, with respect to a Product, any submission to a Regulatory Authority of any appropriate regulatory application specific to the Product, including any submission to a Regulatory Authority and any supplement or amendment thereto, and further including a Biologics License Application or a Marketing Authorization Application. Once granted a Regulatory Application shall be known as a " Commercial License " in this Agreement;
"Regulatory Authority"	the U.S. Food and Drug Administration, the European Medicines Agency, the Medicines & Healthcare products Regulatory Agency, and any successor to any such entities and any other regulatory authority mutually agreed by the parties and named in the Quality Agreement;
"Scope of Work" or "SoW"	on a Program-by-Program basis, the document setting out the detail of the Fujifilm Services to be undertaken by Fujifilm for Atara in connection with such Program and executed by the parties, a form of which is attached hereto as Schedule 5;
"SoW Effective Date"	for each Scope of Work, the date that the Scope of Work is fully signed by both parties;
"Special Waste"	Hazardous Materials as mutually agreed in writing or as otherwise identified in a SoW;

"Stage"	a stage of a Program as described in a SoW. For the purposes of clause 10, a sub- stage described in a SoW (for example a Stage that is clearly identified as a sub- stage and including a suffix such as Stage 1A) shall be deemed to be an independent Stage in its own right;
"Subcontracted Work"	work subcontracted by Fujifilm under clause 21.3;
"Tax"	all duty, sales, use or excise taxes, charges, levies, assessments and other fees of any kind imposed by any governmental or other authority (including, value added tax, sales tax or any other similar type of turnover tax) excluding taxes based on the income of Fujifilm;
"Term"	has the meaning given to in clause 3.1; and
"Willful Misconduct"	a knowing violation of a reasonable and uniformly enforced rule or policy, or intentionally doing that which should not be done or intentionally failing to do that which should be done, or knowing that injury will probably result or recklessly disregarding the possibility that injury may result.

- 1.2 In this Agreement (except where the context otherwise requires) any words following the terms "including", "include", " for example" or any similar expression are by way of illustration and emphasis only and shall not limit the generality or extent of any other words or expr essions.
- 1.3 Insofar as this Agreement or a Scope of Work obliges any party to this Agreement to negotiate, take action or to do something, that party shall conduct such communications, negotiations, take such action or do such thing in good faith acting professionally and, in the case of Fujifilm, using Commercially Reasonable Efforts to achieve the result contemplated in this Agreement. There shall be a general obligation on the parties to act in good faith and with professionalism in relation to the matters contemplated in this Agreement.
- 1.4 In the case of conflict or ambiguity between terms of the main body of this Agreement, any Schedule to this Agreement or any other terms in any Scope of Work, the order of priority shall be as follows: (i) the main body of the Agreement; (ii) the Schedules to the Agreement; and (iii) the main body of the Scope of Work unless a Scope of Work specifically varies a provision of the Agreement or a Schedule to the Agreement by reference to the provision it is amending in which case the Scope of Work shall take precedence in that instance.
- 1.5 In the case of conflict or ambiguity between theterms of this Agreement or any specific Scope of Work and the terms of the QA Documents, the terms of the QA Documents shall prevail solely in relation to cGMP quality matters subject to clause 10.11.
- 1.6 Where a defined term is used in clause 10 (Indemnification & Limitation of Liability) it shall retain its meaning even when the entire word that is a defined term is in capitals.
- 1.7 Where a provision requires agreement, consent and/or approval of a party, such agreement, consent and/or approval, unless expressly stated otherwise, may not be unreasonably withheld, delayed or conditioned.

2. FRAMEWORK AND SCOPE OF SERVICES

2.1 <u>Framework</u>

- 2.1.1 This Agreement establishes (i) a framework to enable Atara to order Fujifilm Services from Fujifilm and (ii) the general terms and conditions applicable to Fujifilm's performance of the Fujifilm Services and, in connection therewith, each Program for Atara.
- 2.1.2 This Agreement is structured so that a separate, numbered, Scope of Work shall be entered into by the parties for the provision of a Program.
- 2.1.3 The provisions of this Agreement shall apply to each Scope of Work and no Scope of Work shall be effective or binding on any party until it has been signed by an authorized representative of each contracting party.
- 2.1.4 Nothing in this Agreement or any Scope of Work shall oblige any party to enter into any Scope of Work and each Scope of Work constitutes a separate contract that is deemed a part of this Agreement.

2.2 <u>Scope of the Fujifilm Services</u>

- 2.2.1 The scope of this Agreement covers ad hoc services (including new product introductions activities), clinical supply of Product, commercial supply of Product, and additional services such as AL&P and storage.
- 2.2.2 On a Program-by-Program basis, if Atara requires Fujifilm Services or Pass Through Related Services, the parties will enter into a Scope of Work under this Agreement. Each Scope of Work shall clearly identify the type of Fujifilm Services and/or Pass Through Services to be provided).

2.3 Forecasting and Orders

2.3.1 During the Term, Atara will submit to Fujifilm the Rolling Forecasts (as defined in Schedule 2) as more specifically described in Schedule 2 and the parties will follow the procedures in connection with such Rolling Forecasts set forth in Schedule 2.

2.4 <u>Co-operative Management</u>

- 2.4.1 Each party will be responsible for its internal decision making process and for reasonably informing the other party of decisions made affecting any or all of the Fujifilm Services in a regular and timely manner.
- 2.4.2 Without limiting the foregoing, the parties will establish the joint committees/teams set forth in Schedule 6 to advise the parties and to execute on certain matters pertaining to the delivery of Fujifilm Services.

3. TERM

3.1 This Agreement shall come into force on the Effective Date and will continue for an initial term of ten (10) years (the **T'erm**").

3.2 [***]:

3.2.1 [***]; and

3.2.2 [***].

- 3.3 Each Scope of Work will take effect from the SoW Effective Date and shall continue until the earlier of:
 - 3.3.1 the date specified in such Scope of Work, or if no such date is specified, the date the Program, or part of the Program referred to in such Scope of Work is completed; or
 - 3.3.2 termination of this Agreement or the relevant Scope of Work in accordance with the terms of this Agreement.

PERFORMANCE OF PROGRAM

4.1 Fujifilm:

4.

- 4.1.1 shall perform the Fujifilm Services and Pass Through Related Services (if any) and carry out each Program, or parts of each Program, in accordance with:
 - (a) the terms of this Agreement and any Scope of Work for such Program,
 - (b) Applicable Laws,
 - (c) the Quality Agreement and cGMP (in both cases when applicable), and
 - (d) the Process Specification for the applicable cGMP Batch (if any),
- 4.1.2 [***] Notwithstanding the foregoing, if Fujifilm does supply Non-Conforming Batches to Atara hereunder then clause 6 shall apply and in the case of [***], these shall be resolved in accordance with clause 6.6).
- 4.2 The parties agree that it shall not be considered a breach of this Agreement or any Scope(s) of Work, by Fujifilm if an objective of a Program is not achieved provided that Fujifilm has complied with itsobligations set out in clause 4.1. Notwithstanding any contrary provisions in this Agreement or any Scope(s) of Work, subject to Fujifilm's compliance with clause 4.1.1 and 4.1.2, the parties acknowledge and agree that the Fujifilm Services to be performed during the Program prior to the [***] are by their nature developmental and Fujifilm cannot (and consequently does not) guarantee to Atara the achievement of a successful outcome for the Program, production of Conforming Batches or production of a specified volume of Product.
- 4.3 Each Scope of Work contains assumptions on which Fujifilm's ability to perform a Program depends. If an assumption set out in the Scope of Work proves to be incorrect or actual circumstances differ from an assumption (including if such assumption cannot be met at such time as Fujifilm reasonably requires to enable it to perform its obligations) then the parties shall agree a Change to account for the change in assumption.

- 4.4 Atara shall:
 - 4.4.1 provide Atara Deliverables within the timeframe(s) set out in the SoW;
 - 4.4.2 meet all its obligations and responsibilities under this Agreement, any Scopes of Work and the Quality Agreement; and
 - 4.4.3 comply with Applicable Laws.
- 4.5 Fujifilm's performance of Fujifilm Services and/or delivery of each Batch depends upon Atara's compliance with clause 4.4. Fujifilm will not be in breach of any obligation under this Agreement or a Scope of Work to the extent Fujifilm's failure to meet that obligation is a direct result of Atara's breach of performance of any obligations under clause 4.4.
- 4.6 Each party shall ensure that its staff involved in the delivery of the Program do not bypass processes or controls (including operating and corporate policies and procedures and Facility-specific policies) put in place under the Quality Agreement, this Agreement or a Scope of Work by virtue of direct action enabled by their familiarity with the Facility and personnel.
- 4.7 Fujifilm will keep complete and accurate records (including [***]) of all work done by it under this Agreement, in form and substance as specified in the applicable Scope of Work, the Quality Agreement, and this Agreement (collectively, the 'Records"). All such Records will be the property of Atara. Except as required by or necessary to comply with Applicable Law, enforce its rights or perform its obligations under this Agreement, any Scope of Work or the Quality Agreement, Fujifilm will not transfer, deliver or otherwise provide any such Records to any party other than Atara, without the prior written consent of Atara. All original Records of the performance of the Fujifilm Services, including, the manufacture of Product under this Agreement will be retained and archived by Fujifilm in accordance with cGMP (if applicable), Applicable Law and the Quality Agreement, but in no case for less than a period of [***] following completion of the applicable Program. Atara will be provided electronic copies of completed Records via an Atara-specific Fujifilm-hosted data room. For the sake of clarity, [***] in connection with Atara's review of Records via the Atara-specific Fujifilm-hosted data room in connection with the Release of a Batch or Fujifilm Disposition [***]. Upon Atara's request [***], Fujifilm will promptly provide Atara with hard copies of such Records following written request therefor. [***] after completion of a Program, and upon Atara's written request, all of the aforementioned records will be sent to Atara or Atara's designee [***]; provided, however, that Atara may elect to have such Records retained in Fujifilm's archives for an additional period of time at a reasonable charge to Atara. Fujifilm will retain copies of all of such Records and Atara will provide Fujifilm full access to all original Records if Fujifilm is required or compelled to furnish any of such original Records in connection with a regulatory inspection by any Authority or otherwise in connection with Applicable Law.
- 4.8 All Fujifilm Services shall be performed at the Facility. Fujifilm shall notify Atara of and shall obtain Atara's prior written consent (such consent not to be unreasonably withheld, conditioned or delayed) to any proposed change in the location of the Facility or use of any replacement or additional facility for the performance of the Fujifilm Services under this Agreement.
- 4.9 Except as otherwise provided in a Scope of Work, Fujifilm shall be responsible, [***], for performing validation of the Facility, the equipment in the Facility and cleaning and maintenance processes employed in the Process in accordance with cGMP (if applicable), the Quality Agreement, and the applicable Scope of Works.

- 4.10 Fujifilm shall be responsible, [***], for obtaining and maintaining throughout the Term, any Facility or other licenses or permits, and any regulatory and government approvals necessary for the performance of the Fujifilm Services under this Agreement (but for the avoidance of doubt not Product specific applications, for example IND or BLA). At Atara's reasonable request, Fujifilm shall, within [***], provide Atara with copies of all such relevant licenses, permits and approvals that are ordinarily required by the relevant Regulatory Authorities and Atara requires in connection with its Regulatory Applications. Subject to the obligations of confidentiality set out in clause 12, Atara will have the right to use any and all information contained in such licenses, permits and approvals to the extent required in connection with its Regulatory Applications, regulatory approval, and clinical and commercial development of Product.
- 4.11 As specified in the Batch Documentation, on a Program-by-Program basis, Fujifilm will take and retain, for such period and in such quantities as may be required by cGMP (if applicable) and the Quality Agreement, samples of Product from the Process produced under such Program. Further, upon Atara's written request, Fujifilm will submit such samples to Atara and/or contract testing laboratories specified by Atara in the applicable SoW for such Program or Quality Agreement (but for the avoidance of doubt Fujifilm shall be entitled to retain the samples set forth in the applicable sampling plan required to be retained at the Facility in accordance cGMP and shall not be obliged to submit all such samples to Atara and/or contract testing laboratories).
- 4.12 During the Term, and in accordance with the Quality Agreement and Fujifilm's then current person-in-plant SOP for the Facility, Fujifilm will permit Atara or its duly authorized representatives to oversee the performance of the Fujifilm Services related to and including cGMP manufacture of Product at the Facility, including the manufacturing of any Batch; provided, that, Atara will (a) use only common areas and those areas of the Facility where the Fujifilm Services are performed and (b) will not enter or attempt to access any areas indicated by Fujifilm as accessible to authorized personnel only. Notwithstanding the foregoing, during the [***] the Term, Fujifilm agrees to permit up to [***] Atara representatives to be at the Facility to observe up to two different Processes (or parts of a Process), for up to [***], unless otherwise agreed by Fujifilm. Atara will and will cause its duly authorized representatives to comply with the Fujifilm's reasonable instructions and/or monitoring policies (as the same may be amended from time to time) at all times any Atara representatives are in the Facility.
- 4.13 In the event of Product recalls (as determined in accordance with the Quality Agreement), Fujifilm and Atara shall reasonably cooperate to enable Atara to complete the recall and shall thereafter [***]. The costs and expenses of conducting any recall shall be treated as follows:
 - (a) If any Product is recalled solely as a result of the supply by Fujifilm of Product that was not manufactured in compliance with the terms of this Agreement or the Quality Agreement or of Fujifilm's negligence, Gross Negligence or Willful Misconduct, [***]. [***]. If Atara and/or its Affiliates elect to utilize a Third Party to conduct a recall, Atara shall notify Fujifilm of such intent and take into account Fujifilm's comments and suggestions regarding such Third Party in good faith. Upon selection of a Third Party by Atara Fujifilm will work with and reasonably cooperate with such Third Party as directed by Atara.
 - (b) If each party has contributed to the cause of a recall, [***].
 - (c) All recalls of Product other than those described in clauses (a) or (b) above shall [***]. Fujifilm will work with and reasonably cooperate with Atara in accordance

with the Quality Agreement and any Third Party assisting in the conduct of a recall, as designated and directed by Atara, at Atara's sole cost.

4.14 During the Term, Atara may request, and Fujifilm shall provide, technology transfer services as contemplated and in accordance with clause 14.6.2(c).

5. QUALITY AND REGULATORY MATTERS

5.1 <u>Quality Agreement</u>

- 5.1.1 The parties will execute the Quality Agreement on or prior to the Effective Date
- 5.1.2 Atara acknowledges that Fujifilm shall not commence any cGMP activity under this Agreement until the Quality Agreement is executed by both parties.

5.2 <u>Regulatory Assistance</u>

- 5.2.1 Atara shall provide Fujifilm with a copy of the relevant Facility related sections of Module 3 of Atara's Chemistry, Manufacturing and Controls section of any submission to a Regulatory Authority supporting Atara's regulatory filing activities for the applicable Product or Process which relates to or contains information about the Process; the Facility (including Fujifilm equipment); and/or the Fujifilm Services ("CMC Section") in accordance with the Quality Agreement. [***] Atara shall reasonably consider Fujifilm's comments to a CMC Section in relation to any information regarding, or impacting, Fujifilm including any information regarding the Process, equipment, controls and analytics or any information provided to Atara by Fujifilm related to or in accordance with the Quality Agreement. In the event the CMC Section materially impacts the Fujifilm Services in a SoW, the parties shall discuss in good faith and mutually agree to a Change to such SoW and any Delay arising as a result of such impact shall be deemed to be an Atara Delay.
- 5.2.2 During a Program, Atara may request assistance from Fujifilm in respect of the CMC Section. Such assistance will be documented in a Scope of Work together with the Charges payable for such assistance. However, no advice or assistance given by Fujifilm shall be deemed to be or construed as a guarantee that a Product will receive regulatory approval.
- 5.2.3 Fujifilm will provide one electronic (PDF) copy of any documents which may be reasonably required by Atara in support of its regulatory filing activities within [***] of Atara's request (email being sufficient). If Atara requires copies of the laboratory notebooks, provision of these will be subject to discussion and agreement by the parties and agreement of an additional fee associated with copying.

5.2.4

- (a) Atara shall have the exclusive right and responsibility for determining regulatory strategy, decisions and actions relating to each Program and each Product, including any Regulatory Application.
- (b) Fujifilm shall have the exclusive right and responsibility for determining regulatory strategy, decisions and actions to the extent relating to:

- 1. the Facility (including in particular utilities and equipment);
- 2. Fujifilm's quality systems, policies and internal procedures;
- 3. any requirement imposed on Fujifilm by a Regulatory Authority; or
- 4. any other commitments to Third Party customers made by Fujifilm after the Effective Date,

(each a "Fujifilm Regulatory Responsibility").

- 5.2.5 Atara acknowledges and agrees that the Fujifilm Quality Assurance team reserves the right to conduct the Fujifilm Disposition of the Product to Atara in accordance with the Quality Agreement. Fujifilm acknowledges and agrees that the Atara Quality Assurance team reserves the right to perform the Atara Release of the Product in accordance with its quality management systems.
- 5.2.6 In the event Atara makes any change to its regulatory filings, including any Regulatory Application(s), which may have an impact on any Fujifilm Regulatory Responsibility or Fujifilm's performance of its obligations under this Agreement, Atara shall provide Fujifilm an opportunity to review and comment on such changes by providing Fujifilm [***] prior written notice together with a copy of such changes. Atara shall reasonably consider Fujifilm's comments to such changes to the CMC Section of its regulatory filings to the Regulatory Authorities in relation to any information regarding, or impacting, Fujifilm including any information regarding the Process, equipment, controls and analytics or any information provided to Atara by Fujifilm related to or in accordance with the Quality Agreement. In the event such changes to the CMC Section materially impacts the Fujifilm Services in a SoW, the parties shall discuss in good faith and mutually agree to a Change to such SoW and any Delay arising as a result of such impact shall be deemed to be an Atara Delay .

6. CONFORMING BATCHES AND NON-CONFORMING BATCHES

- 6.1 Each cGMP Batch will be determined to be a Conforming Batch or a Non-Conforming Batch in accordance with the Quality Agreement.
- 6.2 As set forth in the applicable Scope of Work, each cGMP Batch of Product will be sampled and tested by Fujifilm, a Permitted Subcontractor, or its Affiliates, including conducting stability testing as per the provisions of the Quality Agreement, against the Product Specification, and Fujifilm will review the documentation relating to the manufacture of the Batch and will assess if the manufacture has taken place in compliance with cGMP (if applicable) and the Process as summarized in the Certificate of Analysis.
- 6.3 If during cGMP manufacturing Fujifilm discovers any Deviation, Fujifilm will inform Atara in accordance with the reporting requirements set out in the Quality Agreement. For the avoidance of doubt, Atara acknowledges that the occurrence of a Deviation does not automatically mean that a Batch is a Non-Conforming Batch.
- 6.4 In respect of Conforming Batches, Fujifilm will complete Fujifilm Disposition, and issue to Atara a Certificate of Analysis, a Certificate of Compliance and a complete copy of Batch

records and applicable analytical data (collectively, the "Batch Documentation"). The provisions of clauses 6.5 to 6.9 shall apply to Non-Conforming Batches only.

6.5

If a Batch is determined to be a Non-Conforming Batch (either duringujifilm Disposition or Atara Release), then an investigation on such Non-Conforming Batch will be undertaken in accordance with the Quality Agreement and clauses 6.6 and 6.9 hereof (as applicable).

- 6.6 Within the period specified in the Quality Agreement after Atara's receipt of the Batch Documentation (the Product Acceptance Period"), Atara shall conduct the activities constituting the Atara Release to determine, by review of the Batch Documentation, whether or not Atara agrees that the given Batch is a Conforming Batch. Fujifilm will provide timely answers to information requests and resolution issues arising from Atara's review of such Batch Documentation. If within the Product Acceptance Period Atara makes a reasonable determination that such Batch is a Non-Conforming Batch, Atara shall have the right to initiate the investigational process as set forth in the Quality Agreement with respect to that Batch and shall so notify Fujifilm in writing within the Product Acceptance Period. Such written notice shall specify, in sufficient detail, the manner in which Atara believes the Batch fails to conform to the Product Specification. If Atara does not submit written notice of initiation of the investigational process with respect to a given Batch within the Product Acceptance Period, such Batch will be deemed to have satisfied the Atara Release. Atara shall notify Fujifilm within [***] following discovery of any [***] Defect. If Atara notifies Fujifilm in writing that it has discovered a [***] Defect[***] of the relevant Batch, Atara shall provide all available data and documentation in its possession and control to Fujifilm to support its determination, and if Fujifilm discovers a [***] Defect in a Batch, it shall provide Atara all available documentation regarding such discovery and any other supporting data and documentation Atara may reasonably request. If it is agreed by the parties (failure to agree triggering the process described in clause 6.9) that the cause of the Batch being a Non-Conforming Batchis a failure by Fujifilm to comply with clause 4.1.1, then the remediation procedure set out in clause 6.8 shall be applied.
- 6.7 If a Batch is a Non-Conforming Batch and the cause of that Batch being a Non-Conforming Bat<u>db not</u> a failure by Fujifilm to comply with clause 4.1.1 then [***]. Any further work in relation to the Non-Conforming Batch (such as analysis of the Batch) or manufacture of a replacement cGMP Batch shall be carried out [***] to be agreed in writing by the parties in a Change.
- 6.8 If a Batch is a Non-Conforming Batch and the cause of the Batch being a Non-Conforming Bat<u>db</u> a failure by Fujifilm to comply with clause 4.1.1 then:
 - 6.8.1 Fujifilm shall use Commercially Reasonable Efforts to manufacture a replacement cGMP Batch **(Keplacement Batch**'') as soon as is reasonably practicable [***]; and
 - 6.8.2 if the Non-Conforming Batch is produced before the [***]:
 - (a) the Pass Through Costs and Pass Through Services Cost relating to the Replacement Batch shall be payable by [***]; and
 - (b) in the event a Replacement Batch which is a Conforming Batch cannot be produced within [***] of the original planned Delivery Date for such Batch then [***] shall apply.

- 6.8.3 If the the Non-Conforming Batch is produced after the[***]:
 - (a) the Pass Through Costs and Pass Through Services Cost relating to the Replacement Batch shall be borne by [***]; and
 - (b) in the event a Replacement Batch which is a Conforming Batch cannot be produced within [***] of the original planned Delivery Date for such Batch then [***] shall apply.
- 6.8.4 If Atara requests delivery of a Non-Conforming Batch, the parties shall agree in writing (in a Change) on fair consideration payable for that Non-Conforming Batch . Fujifilm agrees to deliver a Non-Conforming Batch to Atara on the express condition that it (i) will not be used for human or clinical trials; (ii) will be labeled as "Not for Human Use"; and (iii) is subject to Atara's indemnity given under clause 10.10.4. Atara acknowledges and agrees that it shall use any material produced in a Non-Conforming Batch at its own risk.
- 6.9 If the parties cannot agree if a Batch is a Conforming Batch or a Non-Conforming Batch and/or if the cause of a Batch being a Non-Conforming Batch is not agreed by the parties, then this clause 6.9 shall apply:
 - 6.9.1 the parties will first exhaust the investigation/resolution options set out in the Quality Agreement including reference to the MRB under the Quality Agreement and transparent disclosure of all applicable reports and analysis on which their respect opinion is based;
 - 6.9.2 if the MRB is unable to resolve this matter then the documentation related to the applicable Batch will be reviewed by an independent cGMP consultant acceptable to both parties (acting reasonably). The result of such independent review will be binding for both parties solely for the purpose of determining whether the Batch is a Non-Conforming Batch and/or if a Non-Conforming Batch is caused by a failure by Fujifilm to comply with clause 4.1.1;
 - 6.9.3 if the independent cGMP consultant finds that the Batch is not a Non-Conforming Batch or that the Non-Conforming Batch was not caused by a failure by Fujifilm to comply with clause 4.1.1, Atara will pay Fujifilm for the Batch in question in accordance with clause 6.7 plus the reasonable cost to Fujifilm of the investigation;
 - 6.9.4 if the independent cGMP consultant finds that the Batch is a Non-Conforming Batch and that the Non-Conforming Batch is caused by a failure by Fujifilm to comply with clause 4.1.1, the remedial procedure set out in clause 6.8 will be applied; and
 - 6.9.5 all fees and expenses of the independent cGMP consultant incurred in making such determination, will be paid by the party against whom the determination is made. If, however, the independent cGMP consultant cannot establish the root cause of the non-conformance, then [***].

7. DELIVERY, TITLE, RISK, STORAGE AND INVENTORY

7.1 <u>Delivery to Atara</u>

- 7.1.1 Delivery by Fujifilm to Atara, or Atara's designee, of any material in connection with a Program including any quantity of Product manufactured during the Program, any Process-Specific Equipment and return of anyAtara Deliverables or any materials delivered to Fujifilm under a SoW ("Materials") will be made Ex Works the Facility (Incoterms 2020) and clauses 7.1 to 7.4 shall apply to such Materials. Fujifilm shall package the relevant Materials ready for shipment in accordance with Atara's reasonable instructions and in accordance with the Quality Agreement and the relevant Scope of Works for AL&P (if applicable). Fujifilm shall provide all shipment documentation set forth in the Scope of Work and the Quality Agreement.
- 7.1.2 Delivery of Materials will occur on completion of Fujifilm Disposition and delivery to Atara of all Batch Documentation (the "**Delivery Date**") at which point Fujifilm shall deliver all Materials ordered by Atara under this Agreement, into inventory ("**Inventory**") at the Facility and the same shall be stored at the Facility whilst in Inventory in accordance with the applicable Material specifications and Quality Agreement. In no event will Fujifilm be required to store any Material in Inventory beyond the Term. In the event of any early termination of this Agreement, the parties shall mutually agree on the final disposition (i.e. delivery or destruction) of any Materials remaining in Inventory prior to the effective date of such termination in accordance with Part 3 of Schedule 1.
- 7.1.3 Physical delivery of Materials to Atara (or its designee) out of Inventory will occur in accordance with an indicative delivery schedule which shall be agreed by the parties' Program teams in advance. Unless otherwise expressly agreed by the parties in writing, Fujifilm will not make Product available for collection out of Inventory by Atara until both: (a) Fujifilm Disposition is complete; and (b) Atara's quality assurance team has approved such Product for Atara Release in a written notice in accordance with clause 6.6. Materials for which a license is required under clause 11.3 will not be made available for collection by Atara until a license has been signed by the parties.

7.2 <u>Risk and Title</u>

- 7.2.1 Risk in Materials shall pass to Atara on the Delivery Date (save for risk inProcess-Specific-Equipment in relation to which risk shall pass as set forth in Section 5.2 of Part 3 of Schedule 1.
- 7.2.2 Title to the Product shall pass to Atara on the Delivery Date.

7.3 <u>Delivery to Fujifilm</u>

- 7.3.1 Delivery of any materials which Atara is required to supply to Fujifilm pursuant to a SoWshall be delivered to Fujifilm DDP, the Facility (Incoterms 2020). Risk in those materials remains with Atara until delivered at the Facility subject to the limitations of liability set out in clause 10 and subject to any specific exemptions described in a SoW.
- 7.3.2 Atara will at all times retain title to and ownership of the Atara Deliverables, and the Atara-Supplied Consumables.

7.4 <u>Inventory</u>

- 7.4.1 Fujifilm will store the Consumables, Atara Deliverables, the Atara-Supplied Consumables, Products (including intermediates (PBMC, BLCL, vectors) of Product), Inventory, and any materials delivered to Fujifilm under a SoW and any work in process at each and every stage of the Process in accordance with cGMP and the Quality Agreement (in both cases if applicable).
- 7.4.2 Atara Deliverables, the Atara-Supplied Consumables, Product (including intermediates (PBMC, BLCL, vectors) of Product), Inventory, and any materials delivered to Fujifilm under clause a SoW and any work in process at each and every stage of the Process shall be stored in accordance with the Quality Agreement, the Facility's segregation policies, and in such a way as to be able at all times to clearly distinguish such materials from products and materials belonging to Fujifilm, or held by it for a third party's account.
- 7.4.3 Fujifilm shall ensure that such Atara-Supplied Consumables, Atara Deliverables and Product (including intermediates (PBMC, BLCL, vectors) of Product)), Inventory, any materials delivered to Fujifilm under a SoW, and any work in process are free and clear of any liens or encumbrances. Fujifilm will at all times take such reasonable measures as are required to protect the Atara-Supplied Consumables, Atara Deliverables, Product (including intermediates (PBMC, BLCL, vectors) of Product)), Inventory, any materials delivered to Fujifilm under a SoW, and any work in process from loss, damage and theft at all stages of the Process. Fujifilm will immediately notify Atara if at any time it believes any Product (including intermediates (PBMC, BLCL, vectors) of Product), Atara-Supplied Consumables, or Atara Deliverables, Inventory or any work in process have been damaged, lost or stolen.
- 7.4.4 During the term of this Agreement, Fujifilm shall provide to Atara real time inventory visibility for: (a) intermediates (PBMC, BLCL, vectors), components of Product; and (b) Product (Drug Product and Finished Drug Product).

8. PRICE AND PAYMENT

- 8.1 Atara shall pay to Fujifilm:
 - 8.1.1 the "Charges" being:
 - (a) Processing Room Fees;
 - (b) the fees for the Fujifilm Services as set out in the applicable Scope(s) of Work; and
 - 8.1.2 the Pass Through Costs and Pass Through Services Costs;
- 8.2 Fujifilm may invoice Atara for the Charges in respect of each Program in accordance with the terms set out in the Scope of Work (as applicable) and Schedule 1.
- 8.3 Atara shall pay each undisputed invoice issued to it by Fujifilm within [***] of the date of invoice, in full and in cleared funds in USD by electronic transfer to the financial institution specified in the relevant invoice.

- 8.4 The Charges are exclusive of any Tax which may apply and which shall be payable by Atara to Fujifilm at the rate prescribed by law.
- 8.5 If there is a change in the rate of Tax payable or in the Tax treatment of some or all of the Fujifilm Services, Pass Through Related Services or the Product, including due to a change of law or practice or interpretation of the existing legislation or revised determination of a relevant tax legislation or tax practice, then the appropriate party, as required by applicable law shall be responsible for such Tax (for clarity, the foregoing does not relieve Atara of its responsibility for such Taxes on Consumables purchases by Fujifilm under this Agreement or Pass Through Related Services that will be passed through to Atara pursuant to the terms of this Agreement) which becomes due on that supply and any fees and/or interest which is being levied in relation to the outstanding sums and/or non-payment.

The responsible party, as required by applicable law, shall: be responsible for the collection, remittance and payment of any or all Taxes, in respect of the purchase, importation, exportation, sale or other distribution of any materials delivered to Atara by Fujifilm in connection with a Program

- 8.5.1 Atara shall:
- 8.5.2 be responsible for payment of Taxes on purchases by Atara of Fujifilm Services, the purchase of Consumables under this Agreement and Pass Through Related Services from Fujifilm; and
- 8.5.3 make all payments under this Agreement without withholding or deduction of, or in respect of, any Tax unless required by law.
- 8.6 The parties shall exercise their reasonable efforts to ensure that any withholding Taxes imposed are reduced as far as possible under the provisions of any applicable tax treaty and shall cooperate in filing any forms required for such reduction. Each party shall cooperate with the other and furnish the other party with appropriate documents, including Tax documentation, to secure application of the most favorable rate of withholding Tax under Applicable Law (or exemption from such withholding tax payments, as applicable).
- 8.7 Without prejudice to any other right or remedy that it may have, if Atara fails to pay any undisputed Charges, in whole or in part, to Fujifilm on the due date for payment:
 - 8.7.1 Atara shall pay interest on the overdue amount at the rate of [***] or, if lower, the maximum applicable legal rate. Such interest shall be payable in respect of the period from the due date until actual payment of the overdue amount (whether before or after judgment) in accordance with clause 8.3; and
 - 8.7.2 (except where Atara has complied with its obligations in clause 8.9 below) Fujifilm will notify Atara in writing that it has not paid undisputed Charges on the due date for payment. On expiry of the [***] following such written notice, Fujifilm may issue a second written notice stating that it will immediately suspend work on the Program in respect of which payment is overdue if payment is not made within [***] of such second written notice. If payment has not been made within such [***] period Fujifilm may suspend such work without further notice until the undisputed payment in such notices has been made in full.
 - 8.8 If Atara disputes the payment of any Charges or a part of them, Atara shall:

- 8.8.1 notify Fujifilm of the disputed amount including giving reasonable details of the dispute as soon as reasonably practicable after receipt of the invoice but in any event within [***] of receipt of such invoice; and
- 8.8.2 pay the amount of Charges not in dispute in accordance with clause 8.3,

and the dispute shall be dealt with under the dispute resolution process set out in clause 16.

8.9 A party shall not be entitled to withhold, set off or reduce payment of any amounts payable under this Agreement to the other party by any amounts which it claims are owed to it by such other party under this Agreement or any other agreement.

9. WARRANTIES

- 9.1 Fujifilm represents and warrants to Atara that:
 - 9.1.1 It has the full power and right to enter into this Agreement and that there are no outstanding agreements, assignments, licenses, encumbrances or rights of any kind held by other parties, private or public, that are inconsistent with the provisions of this Agreement.
 - 9.1.2 The execution and delivery of this Agreement by Fujifilm has been authorized by all requisite corporate action and this Agreement is and will remain a valid and binding obligation of Fujifilm, enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors.
 - 9.1.3 Each Program will be performed, in all material respects, [***] by individuals who are [***].
 - 9.1.4 It and its Affiliates [***] shall perform the Fujifilm Services and carry out each Program, or parts of each Program, in accordance with [***] and it (i) shall [***]; (ii) shall store Product and Inventory [***]; and (iii) [***]. The warranties set forth in this clause 9.1.4 shall not apply to any Materials that Atara or a Third Party had misused, neglected, improperly handled, altered, abused or used for any purpose other than the one for which it was manufactured or other conditions beyond the control of Fujifilm or its agents.
 - 9.1.5 Unencumbered title to all Product (other than encumbrances not caused by Fujifilm, its Affiliates, or its permitted subcontractors) will be conveyed to Atara upon delivery.
 - 9.1.6 It and its Affiliates have obtained, and shall maintain all requisite material licenses, authorizations and approvals required by federal, state or local government authorities, including the FDA and any other applicable Regulatory Authority, to provide the Fujifilm Services in accordance with this Agreement, and (ii) to its knowledge, there are no pending or uncorrected citations or adverse conditions noted in any inspection of the Facility, which would cause Product to be misbranded or adulterated within the meaning of the U.S. Food, Drug & Cosmetics Act, or other applicable laws.
 - 9.1.7 To the best of its knowledge: Fujifilm, its Affiliates and subcontractors involved in the manufacture, processing or handling of Product, and each of their respective

officers and directors, as applicable, and any person used by Fujifilm or its subcontractors involved in the manufacture, processing or handling of Product under this Agreement: (a) have not been debarred and are not subject to a pending debarment pursuant to section 306 of the United States Food, Drug and Cosmetic Act, 21 U.S.C. § 335a; (b) are not ineligible to participate in any federal and/or state healthcare programs or federal procurement or non-procurement programs (as that term is defined in 42 U.S.C. 1320a-7b(0); (c) are not disqualified by any government or regulatory authorities from performing specific services, and are not subject to a pending disqualification proceeding; and (d) have not been convicted of a criminal offense related to the provision of healthcare items or services and are not subject to any such pending action. Fujifilm will notify Atara promptly upon becoming aware if Fujifilm, its subcontractors involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product, or any person involved in the manufacture, processing or handling of Product under this Agreement, or any of their respective officers or directors, as applicable, is so debarred, ineligible, disqualified or convicted.

- 9.2 Atara represents and warrants to Fujifilm that:
 - 9.2.1 it has the full power and right to enter into this Agreement and that there are no outstanding agreements, assignments, licenses, encumbrances or rights held by other parties, private or public, that are inconsistent with the provisions of this Agreement;
 - 9.2.2 the execution and delivery of this Agreement by Atara has been authorized by all requisite corporate action and this Agreement is and will remain a valid and binding obligation of Atara, enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors;
 - 9.2.3 it will not use, sell, market, promote, advertise, vend, trade, offer, or commercialize Product into any regulatory jurisdiction unless and until it receives the necessary (i) Product approvals from the applicable Regulatory Authority; (ii) express written permission from the applicable Regulatory Authority prior to receipt of Product approvals; or (iii) unless otherwise permitted under Applicable Law prior to receipt of Product approvals in a jurisdiction; and
 - 9.2.4 to the best of its knowledge: Atara, its subcontractors involved in the manufacture, processing or handling of Product, and each of their respective officers and directors, as applicable, and any person used by Fujifilm or its subcontractors involved in the manufacture, processing or handling of Product under this Agreement: (a) have not been debarred and are not subject to a pending debarment pursuant to section 306 of the United States Food, Drug and Cosmetic Act, 21 U.S.C. § 335a; (b) are not ineligible to participate in any federal and/or state healthcare programs or federal procurement or non-procurement programs (as that term is defined in 42 U.S.C. 1320a-7b(0); (c) are not disqualified by any government or regulatory authorities from performing specific services, and are not subject to a pending disqualification proceeding; and (d) have not been convicted of a criminal offense related to the provision of healthcare items or services and are not subject to any such pending action. Fujifilm will notify Atara promptly upon becoming aware if Fujifilm, its subcontractors involved in the manufacture, processing or handling of Product, or any person involved in the

manufacture, processing or handling of Product under this Agreement, or any of their respective officers or directors, as applicable, is so debarred, ineligible, disqualified or convicted.

10. INDEMNIFICATION & LIMITATION OF LIABILITY

- 10.1 Nothing in this Agreement limits or excludes the liability of any party to the other for any liability that is not permitted to be limited or excluded by law and clauses 10.2 to 10.8 are expressly agreed to be subject to this clause 10.1.
- 10.2 SUBJECT TO THE APPLICABLE PROVISIONS OF THIS CLAUANDOEXCEPT IN RESPECT OF THE FOLLOWING PROVISIONS/OBLIGATION: [***] DUE IN ACCORDANCE WITH THIS AGREEMENT AND THE APPLICABLE SCOPE OF WORK:
 - 10.2.1 WITH RESPECT TO ATARA'S [***]; AND
 - 10.2.2 IN RESPECT OF ANY AND ALL [***].
- 10.3 SUBJECT TO THE APPLICABLE PROVISIONS OF THIS CLAUSE 10 AND EXCEPT IN RESPECT OF THE FOLLOWING PROVISIONS/OBLIGA [***] FUJIFILM'S TOTAL LIABILITY, WHETHER OR NOT ARISING, IN CONTRACT, TORT (INCLUDING NEGLIGENCE OR BREACH STATUTORY DUTY), MISREPRESENTATION, RESTITUTION OR OTHERWISE ARISING UNDER THIS AGREEMENT, A SCOPE OF WORK C CONNECTION WITH THE PERFORMANCE OR CONTEMPLATED PERFORMANCE OF THIS AGREEMENT OR A SCOPE OF WORK SHALL CIRCUMSTANCES BE LIMITED AS FOLLOWS:
 - 10.3.1 EXCEPT AS SET FORTH IN CLAUSE 10.3.3 BELOW, [***]; AND
 - 10.3.2 EXCEPT AS SET FORTH IN CLAUSE 10.3.3 BELOW, AND NOTWITHSTANDING ANY PROVISION TO THE CONTRARY IN CLA 10.3[***]; AND
 - 10.3.3 IN RESPECT OF ANY AND ALL LIABILITY ARISING UNDER OR IN CONNECTION WITH [***]; AND
 - 10.3.4 IN RESPECT OF ANY AND ALL LIABILITY RELATING TO THIS AGREEMENT (INCLUDING UNDER ALL SCOPE OF WORKS I HEREUNDER), FUJIFILM'S TOTAL AGGREGATE LIABILITY TO ATARA SHALL BE LIMITED [***].
- 10.4 EXCEPT IN RESPECT OF THE FOLLOWING PROVISIONS/OBLIGATIONS WHICH SHALL REMAIN UNCAPPED: (I) OF BREACH BY EITHER OF CLAUSES 11 OR 12; (II) LIABILITY ARISING UNDER CLAUSES 10.9.3 AND 10.10.5, UNDER NO CIRCUMSTANCES SHALL EITHER PART LIABLE, WHETHER IN CONTRACT, TORT (INCLUDING NEGLIGENCE), FOR BREACH OF STATUTORY DUTY OR OTHERWISE, ARISING UI OR IN CONNECTION WITH THIS AGREEMENT FOR: LOSS OF PROFIT; LOSS OF BUSINESS; DEPLETION OF GOODWILL; LOSS OF ANTICI SAVINGS; LOSS OR CORRUPTION OF DATA OR INFORMATION; OR ANY SPECIAL, INDIRECT, CONSEQUENTIAL OR PURE ECONOMIC COSTS, DAMAGES, CHARGES OR EXPENSES.
- 10.5 Liability for Non-Conforming Batches:
 - 10.5.1 THE PROVISIONS OF CLAUSE 6 SHALL APPLY TO NON-CONFORMING BATCHES AND FUJIFILM SHALL HAVE NO LIABILI RESPECT OF NON-CONFORMING BATCHES

EXCEPT TO COMPLY WITH CLAUSE6 INCLUDING UNDER CLAUSE 9.1.4 (PRODUCT WARRANTY).

- 10.5.2 FUJIFILM GIVES NO, AND DISCLAIMS ANY, WARRANTIES, UNDERTAKINGS OR SIMILAR TERMS WHATSOEVER (WHETHER . COMPLIANCE WITH CGMP OR OTHERWISE) IN RESPECT OF NON-CONFORMING BATCHES DELIVERED TO ATARA PURSUANT TO CLAUSE 6, OR THE USE BY ATARA OF SUCH NON-CONFORMING BATCHES.
- 10.5.3 Atara uses any material produced in a Non-Conforming Batch at its own risk and shall undertake such tests as are necessary in order to satisfy itself that such materials are fit for the purposes for which Atara proposes to use such materials.

10.6 Disclaimer and Liability for Demonstration and Engineering Batches

- 10.6.1 FUJIFILM GIVES NO, AND DISCLAIMS ANY, WARRANTIES, UNDERTAKINGS OR SIMILAR TERMS WHATSOEVER IN RESPECTIVE DEMONSTRATION BATCHES OR ENGINEERING BATCHES OR THE USE BY ATARA OF A DEMONSTRATION BATCH OF DEMONSTRATION BATCH.
- 10.6.2 [***], FUJIFILM SHALL HAVE NO LIABILITY TO ATAIRACONNECTION WITH DEMONSTRATION BATCHES OR ENGINEERING BATCHES OR THE USE BY ATARA OF THE DEMONSTRATION BATCHES OR ENGINEERING BATCHES.
- 10.6.3 Atara uses any material produced in a Demonstration Batch or Engineering Batch at its own risk and shall undertake such tests as are necessary in order to satisfy itself that such materials are fit for the purposes for which Atara proposes to use such materials. Atara expressly agrees that Product produced pursuant to a Demonstration Batch or Engineering Batch is not suitable, and will not be used, for human consumption or use or in clinic trials.
- 10.7 FUJIFILM GIVES NO, AND DISCLAIMS ANY, WARRANTIES, UNDERTAKINGS OR SIMILAR TERMS WHATSOEVER IN RESPECT OF ANY A OR ASSISTANCE GIVEN BY FUJIFILM IN CONNECTION WITH THE USE OF THE PRODUCT IN OR AS A PRODUCT (INCLUDING ADVI ASSISTANCE RELATED TO ANY REGULATORY APPROVAL).
- 10.8 ALL WARRANTIES, CONDITIONS AND OTHER TERMS, EXPRESS (OTHER THAN THOSE SET OUT IN THIS AGREEMENT) OR IM STATUTORY, CUSTOMARY OR OTHERWISE WHICH BUT FOR THIS CLAUSE 10 WOULD OR MIGHT SUBSIST IN FAVOR OF, ATHER (TO THE FULLEST EXTENT PERMITTED BY LAW) EXCLUDED FROM THIS AGREEMENT INCLUDING, IN PARTICULAR, ANWAWRANEDES RELATING TO MERCHANTABILITY, FITNESS FOR A PARTICULAR USE AND NON-INFRINGEMENT.
- 10.9 Fujifilm shall Indemnify Atara, its Affiliates and their respective directors, officers, employees, and agents, and their respective successors and permitted assigns (collectively "Atara Indemnitees"), from any and all third-party claims, actions, causes of action, liabilities, losses, damages, costs or expenses, including reasonable attorneys' fees (collectively "Claims"), which directly arise out of or relate to:

10.9.1 [***];

10.9.2 [***];

26

10.9.3 Fujifilm's use of Fujifilm's Background IP or Fujifilm Foreground IP in performing a Program infringes a third party's Intellectual Property Rights, except to the extent Atara is liable under clause 10.10.5;

in each case, except to the extent Atara is liable under clause 10.10 and expressly subject to clauses 10.3 and 10.4.

10.10 Atara shall Indemnify Fujifilm, its Affiliates and their respective directors, officers, employees, and agents, and their respective successors and permitted assigns (collectively "Fujifilm Indemnitees"), harmless from any and all Claims which directly arise out of or relate to:

10.10.1 [***];

10.10.2 [***];

- 10.10.3 use or resale of Product manufactured or delivered hereunder by or on behalf of Atara or its Affiliates [***];
- 10.10.4 use of Non-Conforming Batches delivered to Atara at its request pursuant to clause 6;
- 10.10.5 Fujifilm's use of Atara Deliverables (except to the extent manufactured by Fujifilm) or Atara's Intellectual Property Rights, in each case, in accordance Atara's written instructions and with this Agreement; and
- 10.10.6 use of Demonstration Batches or Engineering Batches delivered to Atara hereunder;

in each case, except to the extent Fujifilm is liable under clause 10.9 and expressly subject to clauses 10.2 and 10.4.

- 10.11 No Claims incurred pursuant to the Quality Agreement may be made under the Quality Agreement by either party. Accordingly, performance of the Quality Agreement shall be deemed to be performance under this Agreement and as such any breach of the Quality Agreement shall be deemed to be a breach of this Agreement and all Claims shall be construed and limited in accordance with this Agreement.
- 10.12 Each party agrees to take all reasonable steps to mitigate any Liabilities that it may seek to claim from the other under or in connection with this Agreement including pursuant to any indemnity.
- 10.13 If a party is entitled to benefit from an indemnity (the **'Indemnified Party**'') from the other party (the **'Indemnifying Party**'') in accordance with this Agreement (an **"Indemnity Claim**''), the Indemnified Party shall notify the Indemnifying Party in writing of the Indemnity Claim (providing all necessary details) and the Indemnifying Party shall at its own expense conduct all negotiations and any litigation arising in connection with the Indemnity Claim provided always that:
 - 10.13.1 the Indemnifying Party shall consult theIndemnified Party on all substantive issues which arise during the conduct of such litigation and negotiations and shall take due and proper account of the interests of the Indemnified Party;
 - 10.13.2 the Indemnifying Party shall not settle or compromise the Indemnity Claim without the ndemnified Party's prior written consent (not to be unreasonably withheld, conditioned or delayed) and shall ensure that any settlement or compromise does

not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of the Indemnified Party;

- 10.13.3 the Indemnified Party shall not make any admissions or admit liability in relation to the Indemnity Claim or otherwise settle any Indemnity Claim without the written agreement of the Indemnifying Party; and
- 10.13.4 the Indemnified Party shall fully cooperate and assist the Indemnifying Party, at the Indemnifying Party's cost and expense (such expenses incurred by Indemnified Party to be reasonable, customary and documented), in relation to the Indemnity Claim (without limiting the extent of the indemnity).
- 10.14 Without prejudice to (i) the foregoing provisions of this clause 10 and (ii) a party's ability to self-insure (provided it has adequate financial means to do so): each party shall maintain adequate insurance with a reputable company to ensure that it can meet its liabilities under this Agreement.

11. INTELLECTUAL PROPERTY

- 11.1 Subject to the limited licenses of clauses 11.2 and 11.3, neither party shall acquire any right, title or interest in the other party's Background IP.
- 11.2 During the Term, Atara grants to Fujifilm a royalty-free, non-exclusive, non-sublicensable, non-transferable, license to use Atara's Background IP and Atara Foreground IP for the sole and exclusive purpose of performance of each Program at the Facility (or such other facility agreed to in writing by Atara). Atara warrants that, to Atara's knowledge, the use by Fujifilm (or its Authorized Third Parties) of Atara's Background IP or Atara Foreground IP in accordance with this clause 11.2 does not violate any Intellectual Property Rights of any third party and it will notify Fujifilm in writing should it become aware of any claims asserting such violation. For the avoidance of doubt, Atara Background IP and Atara Foreground IP arethe sole Confidential Information of Atara and Fujifilm shall keep Atara Background IP and Atara Foreground IP as used to protect Fujifilm Background IP, Fujifilm Foreground IP and Fujifilm Expression Technology, but, in any event, no less than a reasonable degree of care.
- 11.3 Fujifilm shall not use any Fujifilm Expression Technology in the manufacture of Product or deliver any materials (including any cell bank or cell paste) unless and until a license is granted to Atara in writing on terms to be agreed under the relevant Fujifilm Expression Technology. [***].
- 11.4 All title to and all rights and interest in any Atara Foreground IP shall vest in Atara. [***]
- 11.5 All title to and all rights and interest in any Fujifilm Foreground IP shall vest in Fujifilm. [***].
- 11.6 If requested to do so by the other party, each party shall at the expense of the requesting party execute all documents and do all such further acts as the requesting party may reasonably require to perfect the assignment under clause 11.4 or 11.5.
- 11.7 Fujifilm grants to Atara a royalty free, non-exclusive, perpetual, irrevocable worldwide license that is transferable and sublicensable under the Fujifilm Foreground IP for the exclusive purpose of developing, manufacturing, have manufactured, distributing, transporting, storing, offering for sale, selling, using and otherwise disposing of the Product. Fujifilm represents and

warrants to Atara that, to Fujifilm's knowledge, such use of the Fujifilm Foreground IP does not violate anyntellectual Property Rights of any third party and it will notify Atara in writing should it become aware of any claims asserting such violation.

- 11.8 Fujifilm represents and warrants to Atara that, to Fujifilm's knowledge,the performance of the Fujifilm Services and the Process will not violate any patent, trade secret or other proprietary or intellectual property rights of any third party and it will notify Atara in writing should it become aware of any claims asserting such violation. For clarity, Fujifilm makes no representations or warranties about the intellectual property status of any Atara Background IP or its use in connection with the Fujifilm Services or the Process.
- 11.9 If Atara elects to manufacture Product, or to have Product manufactured by a third party, then Fujifilm will provide to Atara or, subject to the execution of an appropriate confidentiality agreement, Atara's designee, all necessary manufacturing know-how, information, including documentation, technical assistance, materials and cooperation by appropriate employees of Fujifilm as Atara or its designee may reasonably require in order to manufacture Product. Atara will compensate Fujifilm for such assistance.

12. CONFIDENTIALITY

- 12.1 Each party (the "**Receiving Party**") receiving Confidential Information of the other party agrees with the other party (the "**Disclosing Party**"):
 - 12.1.1 to keep the Disclosing Party's Confidential Information confidential and exercise the same degree of care and safeguards with respect to such Confidential Information as used to maintain the confidentiality of the Receiving Party's own information of like character, but, in any event, no less than a reasonable degree of care;
 - 12.1.2 not to access or use the Disclosing Party's Confidential Information save for the purposes of:
 - (a) complying with its obligations and exercising its rights under this Agreement and each SoW.
 - 12.1.3 not to disclose the Disclosing Party's Confidential Information other than to the Receiving Party's:
 - (a) Affiliates that need to know the Confidential Information for the purpose of assisting such Receiving Party in performing its obligations under this Agreement or a SoW and are bound in writing to obligations of confidentiality at least as restrictive as the terms of this clause 12;
 - (b) officers and employees that need to know the Confidential Information for the purpose of assisting such Receiving Party in performing its obligations under this Agreement or a SoW and are bound in writing to obligations of confidentiality at least as restrictive as the terms of this clause 12;
 - (c) contractors and sub-contractors, professional advisers, consultants and agents who are engaged to advise the Receiving Party in connection with a Program or this Agreement and who have a bone fide need to know that Confidential Information to provide the contracted services and are bound in

writing to obligations of confidentiality at least as restrictive as the terms of this clause12; and

(d) solely where Atara is the Receiving Party, pursuant to clause 12.7;

the "Authorized Third Parties".

The obligations of each party relating to the other party's Confidential Information shall expire [***] after termination of this Agreement, provided that such obligations shall survive in perpetuity with respect to all trade secrets of a party so long as such party continues to protect such trade secrets as such. The disclosure of Confidential Information shall not constitute the grant of any license or any other rights, or generate any business arrangements, unless specifically set forth herein or in another writing.

- 12.2 The parties acknowledge and agree that the Disclosing Party may disclose its Confidential Information directly to the Receiving Party's Authorized Third Parties and that such disclosure shall be governed by this clause 12. The Receiving Party shall procure that each of its Authorized Third Parties keeps the Disclosing Party's Confidential Information confidential in accordance with this clause 12 and shall remain primarily liable to the Disclosing Party for any act or omission of any of its Authorized Third Parties.
- 12.3 The Receiving Party shall within [***] of receipt of the Disclosing Party's written request (including after termination of this Agreement and any SoW):
 - 12.3.1 deliver up to the Disclosing Party all items and copies of all or any Confidential Information of the Disclosing Party;
 - 12.3.2 expunge and/or make irretrievable all Confidential Information of the Disclosing Party from any computer or other similar device in which it is stored and, if further requested, certify in writing signed by an authorized representative that it has done the same (provided that this clause 12.3.2 shall not apply to automatically archived electronic files or electronic back-ups made in the ordinary course of business, on secured central servers, which cannot reasonably be deleted and such electronic files shall be retained subject to the obligations of confidence set out in this clause 12); and
 - 12.3.3 destroy all hard copies of notes, analyses or memoranda containing the Disclosing Party's Confidential Information (and, if further requested, certify in writing signed by an authorized representative that it has done the same)

provided that the Receiving Party shall be entitled to retain copies of the Confidential Information to enable it to monitor its obligations under this Agreement or which is required to be maintained by Applicable Laws or a Regulatory Authority subject always to the obligations of confidence under this Agreement.

- 12.4 Confidential Information shall not include information which:
 - 12.4.1 is, or becomes, generally available to the public other than as a direct or indirect result of the information being disclosed by the Receiving Party or its Authorized Third Parties in breach of this Agreement (except that any compilation of otherwise public information in a form not publicly known shall still be treated as Confidential Information);

- 12.4.2 was available to the Receiving Party on a non-confidential basis prior to disclosure by the DisclosingParty;
- 12.4.3 was, is, or becomes available to the Receiving Party on a non-confidential basis from a person who, to the Receiving Party's knowledge, is not under any confidentiality obligation in respect of that information;
- 12.4.4 was lawfully in the possession of the Receiving Party before the information was disclosed by the Disclosing Party;
- 12.4.5 is developed by or for the Receiving Party independently of the information disclosed by the Disclosing Party without use of, reference to, or reliance upon the Confidential Information of the Disclosing Party, as evidenced by the Receiving Party's written records; or
- 12.4.6 the Disclosing Party and the Receiving Party agree in writing is not confidential.
- 12.5 The Receiving Party may disclose Confidential Information of the Disclosing Party when required any governmental authority or agency in compliance with any federal, state and/or local laws and/or regulations, or pursuant to an order of a court of competent jurisdiction, but then only to the extent of such required disclosure and save that the Receiving Party shall, to the extent it is lawful to do so, give prompt notice to the Disclosing Party of any such potential disclosure and reasonably cooperates with Disclosing Party to allow the Disclosing Party a reasonable opportunity to seek a protective order or otherwise limit such disclosure.
- 12.6 Nothing in this clause 12 shall prevent a party from promptly making all disclosures and filings with government authorities as may, in its judgement be required or advisable in connection with the execution and delivery of this Agreement or the consummation of and the performance thereof the transactions contemplated hereby, including, without limitation, disclosures required by the rules and regulations of the Securities Exchange Commission ('SEC"), other government authority, or applicable stock exchange, provided that except where prohibited by Applicable Law or exigent circumstances, the Receiving Party takes reasonable efforts to provide the Disclosing Party at least [***] prior written notice of such disclosure (and the right to review and comment on the proposed disclosure), and discloses only that portion of the Confidential Information that the Receiving Party is legally required to disclose in the reasonable opinion of the Receiving Party's legal counsel opinion. To the extent that this Agreement may need to be filed by Atara with the SEC, Atara shall, prior to making any such filing with the SEC, provide Fujifilm and its counsel with a proposed redacted version of this Agreement which it intends to file with the SEC and to give due consideration to any comments provided by Fujifilm or its counsel and use reasonable efforts to obtain confidential treatment for such required disclosure. Following filing of the Agreement with the SEC, Atara may describe or refer to portions of the Agreement for which confidential information is not obtained from the SEC without the prior review or approval of Fujifilm.
- 12.7 In addition its rights under clauses 12.1.2 and 12.1.3, Atara may use and disclose Confidential Information of Fujifilm to the extent necessary in communications with existing or prospective Atara's investors, sources of finance, acquirers, collaboration partners, sublicensees and commercial partners provided that: (a) such recipients are under obligations of confidentiality obligations at least as restrictive as the terms of this clause 12; (b) none of the financial terms of the Agreement, a SoW or any proposals are disclosed to any such investor, sources of finance, acquirer, collaboration partner, sublicensee or commercial partner that is a

Competitor without the prior written consent of Fujifilm which shall not be unreasonably withheld, conditioned, or delayed (c) no disclosures are made to contract development and/or manufacturing organizations in the biopharmaceutical industry without Fujifilm's prior written consent; and (d) such recipients are treated as Authorized Third Parties for the pupposes of clause 12.2.

- 12.8 Each Party agrees that a breach of this clause 12 may result in irreparable harm to the Disclosing Party. In seeking enforcement of any of the other party's obligations under this clause 12, the Disclosing Party shall therefore be entitled to seek injunctive and other equitable relief to prevent or restrain the breach of this Agreement. Notwithstanding the foregoing, nothing in this Agreement shall be construed as prohibiting the Disclosing Party from pursuing any other remedies that may be available to it, whether at law or in equity.
- 12.9 To the extent that any Confidential Information may include materials subject to the attorney-client privilege, no party is waiving, and no party will be deemed to have waived or diminished, its attorney work-product protections, attorney-client privileges or similar protections and privileges as a result of disclosing any Confidential Information to any other party, its Affiliates or their respective representatives, regardless of whether the Disclosing Party has asserted or is or may be entitled to assert such privileges and protections. In furtherance of the foregoing, no party nor its Affiliates shall claim or contend, in any action involving any party, that any other party waived its attorney work-product protections, attorney-client privileges or similar protections and privileges with respect to any information, documents or other material (whether or not disclosed to any other party or its Affiliates) due to any party's disclosure of Confidential Information (including Confidential Information related to pending or threatened litigation) to any other party, its Affiliates or their respective representatives.

13. CHANGE

- 13.1 Subject to clause 13.2, if a party wishes to change (**Change**") any aspect of this Agreement or any Scope of Work (including if additional or different work is requested or required such as the production of a different number of Batches or if such work is required to be carried out at a different time or if actual circumstances differ from the assumptions set out in a Scope of Work (including if such assumptions cannot be met at all or in a timely fashion)) then the parties shall mutually agree upon the terms of the Change and the Change shall not be effective until the applicable Change document) is signed by each party. Fujifilm will use reasonable efforts to prepare a draft Change within [***] of agreeing with Atara the need for a Change.
- 13.2 In the event that there is a change to Applicable Law which comes into effect after the Effective Date that adversely affects, or is reasonably likely to adversely affect, production of Product by the Process when conducted in accordance with Fujifilm's standard operating procedures or methods, and within the declared constraints of the Facility, then the parties will enter into a mutually agreed upon Change to accommodate that change of Applicable Law; the cost of which shall be allocated as follows:
 - 13.2.1 if the change to Applicable Law specifically relates to the Product or Process, then Atara will be liable for the costs of the Change; and
 - 13.2.2 if the change to Applicable Law relates to the Facility as it is operated by Fujifilm across its customer base or the general manufacturing activities carried out by Fujifilm at the Facility, then Fujifilm will be liable for the costs of the Change; and

13.2.3 if the change to Applicable Law does not fall within either of clause \$3.2.1 or 13.2.2, the parties will discuss and agree on how those costs will be borne in the Change.

14. DELAY, CANCELLATION, TERMINATION AND CONSEQUENCES

- 14.1 <u>Delay</u>:
 - 14.1.1 [***].
 - 14.1.2 [***].
 - 14.1.3 [***].

14.2 <u>Termination for Cause</u>

- 14.2.1 Fujifilm or Atara shall be entitled to terminate this Agreement (and all Scope of Works made under it, together with any Orders made under the applicable Scope of Work(s)) in its entirety:
 - (a) in the case of Fujifilm as the terminating party, and without prejudice to Fujifilm's right to suspend performance under clause 8.8.2: if Atara fails to pay any undisputed amounts when due and payable to Fujifilm hereunder, and fails to make such undisputed payment with interest as calculated pursuant to clause 8.8.1, within [***] after receiving written notice of such payment failure from Fujifilm (provided such notice shall be sent only once the payment at issue becomes delinquent), Fujifilm shall provide Atara with a second written notice and if Atara continues to fail to make such undisputed payment with interest as calculated pursuant to clause 8.8.1, within [***] after receiving such second notice, Fujifilm may terminate immediately upon giving written notice to Atara;
 - (b) if the other party commits a material breach of its obligations or its representations and warranties under this Agreement (other than Atara's failure to make payment), and fails to start to diligently resolve issues caused by the breach within [***] after receipt of notice giving reasonable particulars of the breach and requiring it to be remedied (including the implementation of a resolution plan) with ultimate resolution of those issues caused by the breach within [***] of receipt of the original notice, such party may terminate this Agreement immediately upon written notice to the breaching party, provided, however, that such cure period shall be suspended during any time that a party seeks resolution of a dispute as to whether an alleged material breach occurred pursuant to clause 16. For the avoidance of doubt, a party's Gross Negligence and/or Willful Misconduct shall be deemed a material breach of this Agreement by such party if the terminating party so elects;
 - (c) if the other party takes any step or action in connection with its entering administration, provisional liquidation or any composition or arrangement with its creditors (other than in relation to a solvent restructuring), being wound up (whether voluntarily or by order of the court, unless for the purpose of a solvent restructuring), having a receiver appointed to any of its

assets or ceasing to carry on business or, if the step or action is taken in another jurisdiction, in connection with any analogous procedure in the relevant jurisdiction; or

- (d) upon [***] prior written notice, if the other party or the person controlling the other party has a change in control and the new controlling entity is reasonably considered by the party giving notice either, (i) in the case of Fujifilm giving notice only, to be its Competitor or (ii) in the case of either party giving notice, not to have reasonable financial creditworthiness. For the avoidance of doubt, any new controlling entity with a market capitalization of greater than [***] as of the effective date of any such change in control shall be deemed to have "reasonable financial creditworthiness".
- 14.2.2 If a party commits a material breach of a Scope of Work, the non-breaching party may give written notice to the other party, giving reasonable particulars of the breach and requiring it to be remedied. If the breaching party fails to start to diligently resolve issues caused by the breach within [***] after receipt of notice (including the implementation of a resolution plan) with ultimate resolution of those issues caused by the breach within [***] of receipt of the original notice, then the non-breaching party shall have the right, in its sole discretion, to immediately terminate that Scope of Work (provided, however, that the cure period shall be suspended during any time that a party seeks resolution of a dispute as to whether an alleged material breach occurred pursuant to clause 16).

14.3 <u>Termination by Atara for Convenience</u>

- 14.3.1 Atara may terminate this Agreement in its entirety for convenience upon giving Fujifilm [***] written notice in which case clause 14.4.1 shall apply.
- 14.3.2 Atara may terminate or cancel a Scope of Work in its entirety for convenience by giving not less than [***] written notice to Fujifilm in which case clause 14.4.2 shall apply.
- 14.3.3 Atara may terminate or cancel or one or more Stages of a Scope of Work for convenience by giving not less than [***] written notice to Fujifilm in which case clause 14.4.2 shall apply.

14.4 <u>Financial Consequences of Termination</u>

- 14.4.1 If Fujifilm terminates this Agreement in its entirety under clauses 14.2.1(a) or 14.2.1(b), or if Atara terminates this Agreement in its entirety under clause 14.3.1 (both such termination events being known as an "Atara Led Termination") then the following sums shall becomedue and payable in full within [***] of receipt of an invoice from Fujifilm (invoiced in accordance with the terms of this Agreement) relating to the same (for the avoidance of doubt there is no requirement to raise a new invoice for any sums for which an invoice has already been raised which shall remain payable under the terms of those invoices):
 - (a) The Charges that are due for the Fujifilm Services that have been performed;
 - (b) any amount of the [***] (which shall take into account monies due under clause 14.4.1(a); and

- (c) unpaid Pass Through Costs and Pass Through Services Costs(including any Pass Through Costs and Pass Through Services Costs which Fujifilm has already incurred or has committed to later incur under a non-cancellable order and unrecoverable costs and expenses not yet invoiced to Atara).
- 14.4.2 If Atara terminates a Scope of Work in its entirety or one or more Stages of a SoW for convenience under clause 14.3o2 14.3.3. Atara shall pay Fujifilm the Charges that are due for the Fujifilm Services that have been performed as of the date of such partial termination and any unpaid Pass Through Costs and Pass Through Services Costs (including any Pass Through Costs and Pass Through Services Costs which Fujifilm has already incurred or has committed to later incur under a noncancellable order and unrecoverable costs and expenses not yet invoiced to Atara, in the case of each of the foregoing, as of the date of such partial termination) within [***] of receipt of an invoice from Fujifilm relating to the same (for the avoidance of doubt there is no requirement to raise a new invoice for any sums for which an invoice has already been raised).
- 14.4.3 In the event that any termination event other than those described in clauses 14.4.1 and 14.4.2 occurs, then the following sums shall become due and payable in full within [***] of receipt of an invoice from Fujifilm relating to the same (for the avoidance of doubt there is no requirement to raise a new invoice for any sums for which an invoice has already been raised which shall remain payable under the terms of those invoices):
 - (a) the Charges that are due for the Fujifilm Services that have been properly performed shall become due and payable in accordance with this Agreement and the applicable SoW;
 - (b) Atara shall have no obligation to pay Fujifilm [***];
 - (c) unpaid Pass Through Costs and Pass Through Services Costs (including any Pass Through Costs and Pass Through Services Costs which Fujifilm has already incurred or has committed to later incur under a non-cancellable order and unrecoverable costs and expenses not yet invoiced to Atara) shall be due and payable in accordance with the terms of this Agreement and the applicable SoW; and
 - (d) subject to the entry into a SoW for the services and entry into suitable confidentiality agreement, Fujifilm shall assist Atara in transferring the manufacturing responsibility for the Product(s) to Atara, an Affiliate of Atara or a Third Party designee, as determined by Atara. Such assistance shall include transfer of the Processes, procedures and reasonable data required to manufacture such Product(s) in accordance with the Product Specifications, Process Specifications and Regulatory Authority guidelines but shall not include Fujifilm's Confidential Information or Background IP. Fujifilm will provide such transfer assistance (meaning, in brief, collation of Process documentation, analytical method summaries, provision of Bill of Materials and answering reasonable questions on the same) [***].

14.5 Additional Non-Financial Effects of Partial Termination with respect to SoW(s).

- 14.5.1 If a party exercises any of its rights of termination in respect of only one or more SoWs:
 - (a) this Agreement shall terminate in respect of those SoWs and the provisions of this Agreement relating to termination of this Agreement shall apply in relation to those SoWs;
 - (b) in all other respects this Agreement shall continue in full force and those SoWs in respect of which the party has terminated this Agreement will be deemed to be removed from the definition of the SoWs; and
 - (c) termination of any Scope of Work shall be without prejudice to the rights and remedies of any party which may have accrued up to the date of termination.
- 14.5.2 If a party exercises any of its rights of termination in respect of one or more Stage of a So\\but not the SoW in its entirety) the cancelled Fujifilm Services shall terminate but in all other respects the SoW shall continue in full force.

14.6 Additional Non-Financial Consequences of Termination of this Agreement

- 14.6.1 The termination of this Agreement in its entirety shall be without prejudice to the rights and remedies of any party which may have accrued up to the date of termination.
- 14.6.2 Upon termination of this Agreement in its entirety for any reason whatsoever:
 - (a) the relationship of the parties shall cease and any rights or licenses granted under or pursuant to this Agreement shall cease to have effect save as (and to the extent) expressly provided for in this clause 14.6;
 - (b) each Receiving Party shall return the Confidential Information of the Disclosing Party in accordance with clause 12.3;
 - (c) except as set forth in clause 14.4.3, subject to the entry into a SoW for the services and entry into suitable confidentiality agreement, Fujifilm shall assist Atara in transferring the manufacturing responsibility for the Product(s) to Atara, an Affiliate of Atara or a Third Party designee, as determined by Atara. Such assistance shall include transfer of the Processes, procedures and reasonable data required to manufacture such Product(s) in accordance with the Product Specifications, Process Specifications and Regulatory Authority guidelines but shall not include Fujifilm's Confidential Information or Background IP. Atara shall bear the reasonable costs and expenses of such transfer;
 - (d) in the event of a termination of the Agreement as set out in clause 14.2.1(d), if requested by Fujifilm (i) Atara, the Competitor and Fujifilm will take customary and reasonable measures to protect the confidential and proprietary information of all parties and to address any applicable regulatory or legal matters, such as use of "clean" teams, limiting direct access to another party's confidential or proprietary information, prohibiting "clean"

team members from participating in competitive activity during the remaining term of this Agreement, and up to [***] thereafter (as may be required and permitted by law); and (ii) enter into a confidentiality agreement in customary form with respect to the same; and

the provisions of the following clauses together with any provision which expressly or by implication is intended to come into or remain in force on or after termination shall continue in full force and effect clauses 1, 6, 8.6, 8.7, 9, 10, 11 (except for 11.2 and 11.8), 12, 14, 16, 17.2.2, 18, 19.5, 22 and 23.

14.7 <u>Potential Additional Consequences of an Atara Led Termination[***]</u>

14.7.1 If, during the Term, Atara contracts for Fujifilm Services under a Scope of Work(s) and the Charges for those Fujifilm Services [***], the parties will negotiate in good faith and mutually agree in the relevant Scope of Work(s) the consequences of an Atara Led Termination of such Scope of Work(s).

15. FORCE MAJEURE

- 15.1 Subject to clause 15.2, neither party will be held liable to the other party nor be deemed to have breached this Agreement for failure or delay performing any obligation under this Agreement (except the payment of money due hereunder for Fujifilm Services rendered) to the extent that such failure or delay is caused by or results from acts or events beyond the reasonable control of such party, including acts of God, embargoes, war, acts of war (whether war be declared or not), terrorism, insurrections, riots, civil commotions, strikes, lockouts, or other labor disturbances (other than strikes, lockouts, or labor disturbances involving a party's own employees), government actions, fire, earthquakes, floods, epidemics, pandemics, or quarantines ("Force Majeure") and for so long as such failure or delay continues to be caused by or result from such Force Majeure event. The parties agree the effects of the COVID-19 pandemic that is ongoing as of the Effective Date may be invoked as a Force Majeure for the purposes of this Agreement even though the pandemic is ongoing and those effects may be reasonably foreseeable as of the Effective Date.
- 15.2 If a party is delayed or prevented from performing its obligations due to a Force Majeure Event such party shall:
 - 15.2.1 give notice of such delay or prevention due to the Force Majeure Event to the non-affected party as soon as reasonably practical stating the commencement date and extent of such delay or prevention, the cause thereof and its estimated duration;
 - 15.2.2 use reasonable endeavors to mitigate the effects of such Force Majeure Event, provided that such party shall not be required to procure materials or services at unreasonable prices or under unreasonable terms; and
 - 15.2.3 resume performance of its obligations as soon as reasonably practicable.
- 15.3 If a party's delay or prevention caused by the Force Majeure Event in question or the COVID-19 pandemic continues for more that[***], any party to the affected Scope of Work may give notice in writing to the other to terminate that Scope of Work. The notice to terminate must specify the termination date, which must not be less than [***] after the date on which the notice is given, and once such notice has been validly given, that Scope of Work will terminate on that termination date.

16. **DISPUTE RESOLUTION**

16.1 **Quality Disputes:** If there is a dispute in relation to or in connection with the QA Documents, such dispute shall be dealt with in accordance with the procedures set out in the Quality Agreement .

16.2 Business Escalation:

- 16.2.1 In respect of any dispute concerning this Agreement (other than a dispute in connection with the QA Documents and/or the Quality Agreement) the parties shall seek to resolve the matter as follows:
 - (a) by referral in writing summarizing the nature of the dispute by a party in the first instance to the decision of each party's Program Manager;
 - (b) if the dispute is not resolved within [***] of its referral to the Program Managers it shall be referred to the decision of Fujifilm's Chief Business Officer and Atara's Head of Manufacturing and Supply for other business issues; and
 - (c) if the dispute is not resolved within [***] of its referral to Fujifilm's Chief Business Officer and Atara's Head of Manufacturing and Supply for other business issues, it shall be referred to the decision of each party's President or Chief Executive Officer (as applicable/appropriate).
- 16.3 Arbitration: If the dispute is not resolved after the escalation provided pursuant to clause 16.2.1, the matter will be submitted for mediation administered by the American Arbitration Association under its Commercial Mediation Procedures before resorting to legal remedies, unless otherwise agreed to by the parties in writing. The parties shall equally share any fees or expenses of the mediator. If the matter is not resolved through mediation, then the parties shall be free to avail themselves of any and all legal remedies; provided that the parties irrevocably agree that any legal action or proceeding arising out of or relating to this Agreement brought by any party or its successors or assigns against any other party shall be brought and determined in any federal or state court located in the Court of Chancery of the State of Delaware, provided that if jurisdiction is not then available in the Court of Chancery of the State of Delaware, then any such legal action or proceeding may be brought in any federal court located in the State of Delaware or any other Delaware state court. Each of the parties hereby irrevocably submits to the exclusive jurisdiction of the aforesaid courts for itself and with respect to its property, generally and unconditionally, with regard to any such action or proceeding arising out of or relating to this Agreement and the transactions contemplated hereby. Each of the parties agrees not to commence any action, suit or proceeding relating thereto except in the Court of Chancery of the State of Delaware, provided that if jurisdiction is not then available in the Court of Chancery of the State of Delaware, then any such legal action or proceeding may be brought in any federal court located in the State of Delaware or any other Delaware state court as described herein. Each of the parties further agrees that notice as provided herein shall constitute sufficient service of process and the parties further waive any argument that such service is insufficient. Each of the parties hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any action or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby, (a) any claim that it is not personally subject to the jurisdiction of the courts in Delaware as described herein for any reason, (b)

that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) that (i) the suit, action or proceeding in any such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper or (iii) this Agreement, or the subject matter hereof, may not be enforced in or by such courts.

16.4 <u>General:</u> Notwithstanding the provisions of this clause 16, any party may commence or take proceedings or seek remedies before the courts or any other competent authority for interim, interlocutory or injunctive remedies in relation to this Agreement.

17. **AUDIT**

17.1 <u>Quality Audit:</u>

- 17.1.1 Atara may carry out quality audits at the times, and in accordance with the terms, set out in the Quality Agreement provided that access by Atara and/or its representatives to records, information and systems shall be on a supervised basis, subject to Atara complying with the security and confidentiality requirements of Fujifilm to protect information which relates to anything other than a Program. Except for "for cause" audits, quality audits exceeding once per Calendar Year shall be at the reasonable expense of Atara.
- 17.1.2 Audit access shall not be extended to Fujifilm's confidential records [***].
- 17.1.3 If Fujifilm is in material breach of clause 4.1.1 of this Agreement or if Atarareasonably believes that Fujifilm is in material breach of clause 4.1.1 of this Agreement, Atara may upon giving reasonable written notice to Fujifilm carry out an audit on the same basis as in clauses 17.1.1 and 17.1.2.
- 17.1.4 Additional audits (other than those carried out pursuant to clause 17.1.3) may be carried out on the same basis as in clauses 17.1.1 and 17.1.2 subject to (i) payment of Fujifilm's costs and expenses and the agreement of a commercial rate; and (ii) Atara ensuring such audit will not delay or disrupt Fujifilm's operations at the Facility.

17.2 <u>Financial audits, evidence and assistance</u>:

- 17.2.1 Together with each invoice issued by Fujifilm to Atara hereunder, Fujifilm shall provide reasonably detailed documentation to validate the amounts included on each invoice which are subject to the true-up mechanism contemplated in Schedule 1 Atara may request reasonable additional validating information provided that (i) Atara shall not request such reasonable additional validating information provided that (i) Atara shall not request such reasonable additional validating information for a given amount more than once and (ii) it is acknowledged that Fujifilm shall not be obligated to provide copies of vendor invoices if prevented from doing so by Applicable Law or by Fujifilm's confidentiality obligations to such vendors and, following implementation of a new ERP system by Fujifilm at the Facility following the Effective Date, Fujifilm may elect not to provide copies of vendor invoices if those invoices do not accurately represent the amounts invoiced to Atara because of Fujifilm's use of SAP weighted average "actual cost".
- 17.2.2 Fujifilm will provide reasonable support to Atara in the event that Atara is audited by a third party or to support Atara's maintenance of its financial regulatory obligations (for example Sarbanes Oxley compliance).

18. NOTICES

- 18.1 Subject to clause 18.2 the parties may communicate with each other in any way that is normal in the course of their business.
- 18.2 Any notice given or required to be given under this Agreement shall only be effective if it is in writing, sent to a party at its address or email address and for the attention of the individual, as set out in Schedule 3 (or such other address, email address or individual as that party may notify the other in accordance with this clause 18) and is given in accordance with clauses 18.2 to 18.4 below.
- 18.3 Notice may be given by hand or sent by email, recorded delivery, registered post or airmail and will be deemed to have been duly served:
 - 18.3.1 if delivered by hand, at the time and date of delivery;
 - 18.3.2 if sent by email, at the time and date of sending (with confirmation copy to be sent by the method of clause 18.3.3 or 18.3.4);
 - 18.3.3 if sent by recorded delivery or registered post, 48 (forty-eight) hours from the date of posting (such date as evidenced by postal receipt); and
 - 18.3.4 if sent by registered airmail, five days from the date of posting,

provided that, where in the case of delivery by hand or transmission by email, such delivery or transmission occurs either after 4.00pm on a Business Day, or on a day other than a Business Day, service will be deemed to occur at 9.00 am on the next Business Day.

18.4 In proving service of a notice it will be sufficient to prove that delivery was made or that the envelope containing the notice or document was properly addressed and posted (either by prepaid first class recorded delivery post or by prepaid airmail, as the case may be) or that no failed delivery message was received, as the case may be.

19. EXPORT/IMPORT CONTROLS AND SANCTIONS COMPLIANCE

- 19.1 Each party shall at all times during the term of this Agreement comply with applicable Sanctions or Export/Import Laws and ensure that it has in place appropriate controls and safeguards to prevent any action being taken by it that would amount to or result in a violation of or non-compliance with any Sanctions or Export/Import Laws.
- 19.2 Each party shall provide all information that the other may reasonably require from time to time in order for the other to assess and/or manage its compliance with Sanctions and Export/Import Laws (including, as applicable, provision of end-user statements or applicable Authorizations and notifying the other of any restrictions or export compliance obligations prior to providing Fujifilm access to controlled information/technology).
- 19.3 Atara will not directly or indirectly use, sell, dispose of, (re)export, transship or otherwise transfer any Product, software, technology or Confidential Information: (i) unlawfully to any country in respect of which a Sanctions Authority maintains Sanctions or a Sanctioned Person; (ii) in a manner that would expose Fujifilm to the risk of negative consequences under Sanctions; or (iii) in violation of Export/Import Laws.

- 19.4 If any Authorization is required so that the performance of Program does not contravene any Sanctions or Export/Import LawsAtara will at its own cost and expense obtain that Authorization and Fujifilm shall provide any commercially reasonable assistance (including reasonable information) that Atara may require for the purposes of obtaining that Authorization.Atara's rights and Fujifilm's obligations under this Agreement or any SoW in relation to Program shall immediately be suspended if any required Authorization is not obtained. Any such suspension shall be treated as an Atara Delay.
- 19.5 Each party shall Indemnify the other party against any and all Liabilities in connection with Third Party claims which a party incurs as a result of the other party's non-compliance with the terms of this clause 19.
- 19.6 In this clause 19 the following terms have the following meanings:

"Authorization"	all consents,	licenses,	authorizations,	approvals,	permissions,	registrations,	certificates	and
	clearances and any precondition in any relevant jurisdiction;							

"Export/Import Laws"	(a) any laws of the United States of America, the United Kingdom, th&uropean Union or of any of its Member States that relate to the control of (re)export, transfer or import of Products, software or technology and technical data; or (b) any other (re)export, transfer or import controls or restrictions imposed or adopted by any government, state or regulatory authority in a country in which obligations under this Agreement are to be performed;
"Sanctions"	any economic, financial, trade or other sanction, embargo, import or export ban, prohibition on transfer of funds or assets or on performing services or equivalent measure imposed by any Sanctions Authority or by the laws of any state or any union of states from time to time;
"Sanctions Authority"	means (a) the European Union, (b) any Member State of the European Union or the United Kingdom, (c) the United States of America, and in each case their official institutions, agencies and regulatory bodies imposing or enforcing sanctions legislation in any country or territory from which or into which Atara operates; and
"Sanctioned Person"	any person who appears on or is owned, operated or controlled by any person who appears on any list issued or maintained by any Sanctions Authority or is referred to in any list or public announcement issued by any Sanctions Authority, in each case as amended, supplemented or substituted from time to time.

20. MODERN SLAVERY AND CORRUPTION

20.1 In performing its respective duties under this Agreement each party shall (a) use commercially reasonable efforts to hold itself and its suppliers to the highest performance, ethical and compliance standards, (b) comply with all laws in all jurisdictions where such party operates,

but at a minimum including all applicable anti-slavery laws in the United Kingdomor the U.S.A, and (c) use commercially reasonable efforts to encourage fair and equal treatment for all persons, the provision of safe and healthy working conditions and respect for the environment, and adopt appropriate management systems to conduct business in an ethical manner. Further, in performing its duties under this Agreement, each party acknowledges the value and importance of performance and ethical behavior in its performance under this Agreement.

- 20.2 Each party warrants that on the Effective Date and each SoW Effective Date, it, its directors, officers or employees have not offered, promised, given, authorized, solicited or accepted any undue pecuniary or other advantage of any kind (or implied that they will or might do any such thing at any time in the future) in any way connected with this Agreement or a SoW and that it has taken reasonable measures to prevent subcontractors, agents or other third parties, subject to its control or determining influence, from doing so.
- 20.3 The parties agree that, at all times in connection with and throughout the term of this Agreement, they will comply with and that they will take reasonable measures to ensure that their subcontractors, agents or other third parties will comply with all applicable anticorruption legislation including the U.K. Bribery Act 2010 and the U.S. Foreign Corrupt Practices Act 1977.
- 20.4 Each party shall not do, or omit to do, any act that would cause one of the other parties to be in breach of any anti-corruption legislation including the U.K. Bribery Act 2010 and the U.S. Foreign Corrupt Practices Act 1977.

21. ASSIGNMENT AND SUB-CONTRACTING

- 21.1 Either party may, without prior written consent of the other party, assign or transfer its rights and responsibilities under this Agreement to:
 - 21.1.1 an Affiliate provided that such Affiliate has reasonable financial creditworthiness; or
 - 21.1.2 a purchaser of all or substantially all of the equity of the assigning party provided that such third party has reasonable financial creditworthiness and, in the case of assignment by Atara, is not a Competitor; or
 - 21.1.3 a purchaser of all or substantially all of assets to which this Agreement relates provided that such third party has reasonable financial creditworthiness and, in the case of assignment by Atara, is not a Competitor; or
 - 21.1.4 solely where Atara is such assigning party, an exclusive licensee of any Product provided that (i) such third party has reasonable financial creditworthiness and, in the case of assignment by Atara, is not a Competitor and (ii) that Atara no longer requires services from Fujifilm under this Agreement,

but not otherwise without prior written consent of the other party (such consent not to be unreasonably withheld or delayed) and provided that (a) the assignee agrees in writing to assume all obligations undertaken by its assignor in this Agreement and (b) in relation to assignment in part no such assignment shall relieve the assigning party of responsibility for the performance of any of its obligations under this Agreement. For the avoidance of doubt, any Atara assignee with a market capitalization of greater than [***] as of the effective date of any such assignment shall be deemed to have "reasonable financial creditworthiness."

- 21.2 If a party assigns or transfers all or any of its rights and responsibilities under clause21.1 it shall promptly notify the other parties in writing.
- 21.3 Fujifilm may sub-contract all or any of its obligations under this Agreement provided that in relation to any subcontract of performance of Fujifilm Services, Fujifilm will obtain Atara's prior written consent (which may be by signature of the relevant SoW(s) which specify that an obligation will be sub-contracted). Subcontracted Work to Affiliates of Fujifilm shall be independent, arm's length, and at commercially reasonable costs. Notwithstanding the foregoing, Fujifilm may utilize on-site temporary employees and contract workers, to perform Fujifilm Services without obtaining the prior consent of Atara.
- 21.4 The appointment or engagement by Fujifilm of any subcontractor (including Atara), on-site temporary employees or contract workers shall not relieve Fujifilm from any liability or obligation under this Agreement or the Quality Agreement and Fujifilm shall be responsible for all acts and omissions of the subcontractor to the same extent as if they were Fujifilm's own acts or omissions.

22. GENERAL

- 22.1 Entire agreement: This Agreement and the interpretive letter agreement entered into by the parties simultaneously to this Agreement which is subject to the terms of this Agreement, contain all the terms which the parties have agreed with respect to their subject matter and supersede all previous agreements and understandings between the parties (whether oral or in writing) relating to such subject matter. Each party acknowledges and agrees that it has not been induced to enter into this Agreement by a statement or promise which it does not contain. Each party confirms that save as otherwise expressly set out in this Agreement, the other party gives no warranties either in this Agreement or elsewhere in connection with the provision of each Program. Nothing in this clause 22.1 shall exclude or limit a party' s liability for fraud, including fraudulent misrepresentation.
- 22.2 <u>Third party rights</u>: Save as expressly set out in this Agreement, the parties do not intend that any person who is not a party to this Agreement shall have any right to enjoy the benefit or enforce any of the terms of this Agreement.
- 22.3 <u>Variations</u>: With the exception of Changes, which shall be subject to clause 13, novariation of this Agreement shall be valid unless in writing and signed by a duly authorized representative of each of the parties. A party is entitled to assume that a representative of another party is authorized to act on that party's behalf if that individual is apparently or seemingly acting in the normal course of the business relationship. An exchange of emails shall not be capable of constituting an agreement to vary this Agreement.
- 22.4 <u>Waiver</u>: No failure or delay by a party to exercise any right or remedy provided under this Agreement or by law shall constitute a waiver of that or any other right or remedy, nor shall it preclude or restrict the further exercise of that or any other right or remedy. The single or partial exercise by any party of any right, power or remedy under this Agreement shall not in any circumstances preclude any other or further exercise of it, or the exercise of any right, power or remedy. A waiver by any party of a breach of any provision of this Agreement shall not be considered as a waiver of a subsequent breach of the same or any other provision of this Agreement.
- 22.5 <u>Severability</u>: If any provision of this Agreement or a SoW is found by any court or administrative body of competent jurisdiction to be invalid, illegal or unenforceable in any

jurisdiction then it shall be deemed modified to the minimum extent necessary to make it valid, legal and enforceable. If such modification is not possible that provision shall be deemed to be omitted from this Agreementor the SoW(as applicable) in so far as this Agreement or that SoWrelates to that jurisdiction and the validity and enforceability of that provision in other jurisdictions and the other provisions of this Agreement or SoW shall not be affected or impaired

- 22.6 <u>Counterparts</u>:
 - 22.6.1 This Agreement may be executed in any number of counterparts. Any party may enter into this Agreement by executing a counterpart and all the counterparts taken together will constitute one and the same agreement. This Agreement shall not be effective until each party has signed one counterpart.
 - 22.6.2 Transmission of an executed counterpart of this Agreement (but for the avoidance of doubt not just a signature page) by email (in PDF, JPEG or other agreed format) shall take effect as delivery of an executed counterpart of this Agreement. If this method of delivery is adopted, without prejudice to the validity of the Agreement so made, each party shall provide the others with the original of such counterpart as soon as reasonably possible thereafter.
- 22.7 <u>Publicity</u>: The parties anticipate that there may be opportunities for joint or independent press releases or other announcements relating to the activities contemplated hereby. Notwithstanding the foregoing, no party shall use the name of the other party(ies) or the names of the employees of the other party(ies) nor disclose the terms of this Agreement or any SoW in any press releases, advertising or sales promotional material or in any publication without prior written permission of such party(ies). Such consent shall not be unreasonably withheld. Notwithstanding the foregoing, no such consent shall be necessary to the extent disclosure reiterates information from a prior approved press release or public statement so long as the information remains true, correct, the most current information, and is not misleading or otherwise inadequate as to subsequent developments with respect to the subject matters set forth therein.

23. GOVERNING LAW

23.1 The formation, existence, construction, performance, validity and all aspects whatsoever of this Agreement or any term of it and any issues, disputes or claims arising out of or in connection with it (whether contractual or non-contractual in nature) shall be governed by, and construed in accordance with, the State Law of Delaware, and subject to clause 16 the parties irrevocably submit to the jurisdiction of the courts of Delaware.

[The remainder of this page is intentionally left blank.]

FUJIFILM DIOSYNTH BIOTECHNOLOGIES CALIFORNIA, INC

ATARA BIOTHERAPEUTICS, INC.

By:	/s/ Andy Fenny	By:	/s/ Pascal Touchon
Name:	Andy Fenny	Name:	Pascal Touchon
Title:	Chief Business Officer	Title:	CEO
Date:	January 26, 2022	Date:	1/26/2022

1.1 [***]

Schedule 2 - Forecasting and Orders

[***]

Fujifilm:	Copied to:
[***]	[***]
Atara:	Copied to:
[***]	[***]

3

Schedule 4 - Atara Deliverables

4

Atara Deliverables will be described in each Scope of Work but may include:

SCHEDULE 6: CO-OPERATIVE MANAGEMENT

1. Joint Steering Committee ("JSC")

- 1.1. The parties will establish a JSC to give direction and advice relating to the activities conducted under the terms of this Agreement.
- 1.2. The JSC will be a cross functional committee composed of at least three representatives appointed by each of Atara and Fujifilm, with one representative from each of Atara and Fujifilm having oversight of quality activities, and with one representative from each of Atara and Fujifilm having oversight of the transfer and implementation of manufacturing Processes at the Facility.
- 1.3. All such representatives will be of at least site leadership management level or director (or equivalent level of Atara or Fujifilm).
- 1.4. The JSC will be responsible for:
 - 1.4.1. reviewing the parties' performance through agreed tracking of Key Performance Indicators that may include Quality metrics and Schedule Adherence
 - 1.4.2. tracking adherence to the terms and conditions of this Agreement;
 - 1.4.3. addressing unresolved issues escalated that materially impact critical milestones, budget, supply, inventory targets, regulatory compliance, and/or product quality; overseeing and monitoring the manufacturing activities pursuant to this Agreement;
 - 1.4.4. settling disputes or disagreements (other than quality related disputes which shall be resolved in accordance with the Quality Agreement) that are unresolved by Program team;
 - 1.4.5. investigating and advising on manufacturing and supply issues;
 - 1.4.6. reviewing and recommending for approval by the parties any changes to this Agreement proposed by the Program team, and
 - 1.4.7. performing such other functions as appropriate to fulfill the purposes of this Agreement as determined by the parties.
- 1.5. The JSC will meet at least[***] (virtually or in person) during the term of this Agreement or as otherwise agreed by the JSC or as necessary to make determinations as required of it. Each party will be responsible for all of its own expenses of travelling to and participating in any of these meetings if held in person.

2. <u>Committee Administration</u>

- 2.1. All decisions of the JSC and any other joint subcommittee or team formed under this Agreement, except as expressly set forth herein, will be made by the unanimous agreement of all of its members or their designated representatives, and will be reflected in written meeting minutes which summarily address topics discussed, delegation of work, schedules and decisions of such committee or team.
- 2.2. Such written minutes will be subject to approval by the authorized representatives of the parties; provided, however, that neither the JSC nor any joint committee or team herein may amend or waive any provision of this Agreement, including the financial terms, it being understood that this Agreement may be amended, and any provision of this Agreement may

be waived, only by a written amendment signed by authorized representatives of each party that specifically states that the parties are amending or waiving this Agreement.

3. Disputes

3.1. In the event of disputes at the JSC that it is unable, despite the good faith efforts of all members, to resolve with[#**] then the dispute shall be referred to the dispute management process sunder clause 16 (in the case of a business escalation, commencing at clause 16.2.1(b)).

CERTIFICATION OF THE CHIEF EXECUTIVE OFFICER

PURSUANT TO

SECURITIES EXCHANGE ACT RULES 13A-14(A) AND 15D-14(A)

I, Pascal Touchon, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Atara Biotherapeutics, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 5, 2022

/s/ Pascal Touchon

Pascal Touchon Chief Executive Officer (Principal Executive Officer)

CERTIFICATION OF THE PRINCIPAL FINANCIAL AND ACCOUNTING OFFICER

PURSUANT TO

SECURITIES EXCHANGE ACT RULES 13A-14(A) AND 15D-14(A)

I, Utpal Koppikar certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Atara Biotherapeutics, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 5, 2022

/s/ Utpal Koppikar

Utpal Koppikar Chief Financial Officer (Principal Financial and Accounting Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Atara Biotherapeutics, Inc. (the "Company") on Form 10-Q for the quarter ended March 31, 2022, as filed with the Securities and Exchange Commission (the "Report"), Pascal Touchon, Chief Executive Officer of the Company, and Utpal Koppikar, Chief Financial Officer of the Company, respectively, do each hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 5, 2022

/s/ Pascal Touchon Pascal Touchon Chief Executive Officer (Principal Executive Officer)

/s/ Utpal Koppikar

Utpal Koppikar Chief Financial Officer (Principal Financial and Accounting Officer)

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.